UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE **SECURITIES EXCHANGE ACT OF 1934**

For the quarterly period ended June 30, 2019

OR

 \square TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE **SECURITIES EXCHANGE ACT OF 1934**

For the transition period from to

Commission File Number 000-30929

TG THERAPEUTICS, INC.

(Exact name of registrant as specified in its charter)

Delaware 36-3898269

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

2 Gansevoort Street, 9th Floor New York, New York 10014

(Address including zip code of principal executive offices)

(212) 554-4484

(Registrant's telephone number, including area code)

Securities filed pursuant to Section 12(b) of the Act:

Title of Class	Trading Symbol(s)	Exchange Name
Common Stock, par value \$0.001	TGTX	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1932
during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing
requirements for the past 90 days.

Yes ⊠ No □

Indicate by check mark whether the registrent has submitted electronically and posted on its corporate Web site, if any every Interactive Data File required to

be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months registrant was required to submit and post such files). ✓ Yes □ No	,
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a small emerging growth company. See definition of "large accelerated filer," "accelerated filer," "smaller reporting company," and Rule 12b-2 of the Exchange Act.	
Large accelerated filer ⊠	Accelerated filer \square
Non-accelerated filer \Box (Do not check if smaller reporting company)	Smaller reporting company \square
	Emerging growth company \square
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition per revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box	riod for complying with any new or
Indicate by checkmark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).	
Yes □ No ⊠	
There were 94,755,151 shares of the registrant's common stock, \$0.001 par value, outstanding as of August 2, 2019.	

TG THERAPEUTICS, INC. FORM 10-Q FOR THE QUARTER ENDED JUNE 30, 2019

TABLE OF CONTENTS

SPECIAL C	CAUTIONARY NOTICE REGARDING FORWARD-LOOKING STATEMENTS	3
PART I	FINANCIAL INFORMATION	4
T. 1	P' '100 (4
Item 1	Financial Statements:	4
	Condensed Consolidated Balance Sheets	4
	Condensed Consolidated Building Sheets	T
	Condensed Consolidated Statements of Operations (unaudited)	5
	*	
	Condensed Consolidated Statements of Stockholders' Equity (unaudited)	6
	Condensed Consolidated Statements of Cash Flows (unaudited)	7
	Notes to Condensed Consolidated Financial Statements (unaudited)	8
	Notes to Condensed Consolidated Financial Statements (unaudited)	٥
Item 2	Management's Discussion and Analysis of Financial Condition and Results of Operations	22
	<u></u>	
Item 3	Quantitative and Qualitative Disclosures About Market Risk	29
Item 4	Controls and Procedures	29
DADETI	OTHER INFORMATION	20
PART II	OTHER INFORMATION	30
Item 1	<u>Legal Proceedings</u>	30
TCHT I	<u> 168ai i roccounigo</u>	30
Item 1A	Risk Factors	30
Item 6	<u>Exhibits</u>	65
	•	
	2	

SPECIAL CAUTIONARY NOTICE REGARDING FORWARD-LOOKING STATEMENTS

Certain matters discussed in this report, including matters discussed under the captions "Business" and "Management's Discussion and Analysis of Financial Condition and Results of Operations," may constitute forward-looking statements for purposes of the Securities Act of 1933, as amended, or the Securities Act, and the Securities Exchange Act of 1934, as amended, or the Exchange Act, and involve known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from the future results, performance or achievements expressed or implied by such forward-looking statements. In some cases, you can identify forward-looking statements by words such as "anticipate," "believe," "contemplate," "continue," "could," "estimate," "expect," "intend," "may," "plan," "potential," "predict," "project," "seek," "should," "target," "will," "would" or the negative of these words or other comparable terminology, although not all forward-looking statements contain these identifying words. All written or oral forward-looking statements attributable to us are expressly qualified in their entirety by these cautionary statements. In addition, with respect to all of our forward-looking statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995. Such forward-looking statements include, but are not limited to, statements about our:

- expectations for increases or decreases in expenses;
- expectations for the clinical and pre-clinical development, manufacturing, regulatory approval, and commercialization of our pharmaceutical product candidates or any other products we may acquire or in-license;
- use of clinical research centers and other contractors;
- expectations as to the timing of commencing or completing pre-clinical and clinical trials and the expected outcomes of those trials;
- expectations for incurring capital expenditures to expand our research and development and manufacturing capabilities;
- expectations for generating revenue or becoming profitable on a sustained basis;
- expectations or ability to enter into marketing and other partnership agreements;
- expectations or ability to enter into product acquisition and in-licensing transactions;
- expectations or ability to build our own commercial infrastructure to manufacture, market and sell our drug candidates;
- products being accepted by doctors, patients or payors;
- ability to compete against other companies and research institutions;
- ability to secure adequate protection for our intellectual property;
- ability to attract and retain key personnel;
- availability of reimbursement for our products;
- estimates of the sufficiency of our existing cash and cash equivalents and investments to finance our operating requirements, including expectations regarding the value and liquidity of our investments;
- stock price and its volatility; and
- expectations for future capital requirements.

The forward-looking statements contained in this report reflect our views and assumptions only as of the date this report is signed. Except as required by law, we assume no responsibility for updating any forward-looking statements.

We qualify all of our forward-looking statements by these cautionary statements. In addition, with respect to all of our forward-looking statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995.

Our actual results may differ materially from the results anticipated in these forward-looking statements due to a variety of factors, including, without limitation, those discussed under the captions "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and elsewhere in this report, as well as other factors which may be identified from time to time in our other filings with the Securities and Exchange Commission, or the SEC, or in the documents where such forward-looking statements appear. Given these uncertainties, you should not place undue reliance on these forward-looking statements. Our forward-looking statements do not reflect the potential impact of any future acquisitions, mergers, dispositions, joint ventures or investments we may make or enter into.

You should read this report and the documents that we have filed as exhibits to this report completely and with the understanding that our actual future results, performance or achievements may be materially different from what we expect. Any forward-looking statements contained in this report reflect our current views with respect to future events or to our future financial performance and involve known and unknown risks, uncertainties and other important factors that may cause our actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by these forward-looking statements. The forward-looking statements contained in this report reflect our views and assumptions only as of the date this report is signed. Except as required by law, we assume no obligation to update or revise these forward-looking statements for any reason, even if new information becomes available in the future.

This report also contains estimates, projections and other information concerning our industry, our business and the markets for certain diseases, including data regarding the estimated size of those markets, and the incidence and prevalence of certain medical conditions. Information that is based on estimates, forecasts, projections, market research or similar methodologies is inherently subject to uncertainties and actual events or circumstances may differ materially from events and circumstances reflected in this information. Unless otherwise expressly stated, we obtained this industry, business, market and other data from reports, research surveys, studies and similar data prepared by market research firms and other third parties, industry, medical and general publications, government data and similar sources.

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL INFORMATION

TG Therapeutics, Inc.

Condensed Consolidated Balance Sheets (in thousands, except share and per share amounts)

	June		December 31, 2018		
	J)	Jnaudited)		(Note 1)	
Assets					
Current assets:					
Cash and cash equivalents	\$	57,228	\$	41,958	
Short-term investment securities		27,748		26,943	
Prepaid research and development		8,478		9,691	
Other current assets		987		439	
Total current assets		94,441		79,031	
Restricted cash		1,246		1,241	
Leasehold interest, net		2,203		2,294	
Equipment, net		237		251	
Right of use asset		7,659		_	
Goodwill		799		799	
Total assets	\$	106,585	\$	83,616	
Liabilities and stockholders' equity Current liabilities:					
Accounts payable and accrued expenses	\$	31,498	\$	36,377	
Other current liabilities	Ф	21,478	Φ	219	
Lease liability – current portion		1,283		217	
Accrued compensation		1,708		2,258	
Total current liabilities	_	55,967		38,854	
Deferred rent		33,907		1,462	
Deferred revenue, net of current portion		838		914	
Long-term debt		28,513		714	
Lease liability – non current		8,062			
Long-term liabilities		0,002		18,350	
	_	93,380	_	59,580	
Total liabilities		93,380	_	39,380	
Commitments and contingencies					
Stockholders' equity: Preferred stock, \$0.001 par value per share (10,000,000 shares authorized, none issued and					
outstanding as of June 30, 2019 and December 31, 2018)					
Common stock, \$0.001 par value per share (150,000,000 shares authorized, 93,405,248 and				_	
83,911,855 shares issued, 93,363,939 and 83,870,546 shares outstanding at June 30, 2019 and					
December 31, 2018, respectively)		93		84	
Additional paid-in capital		613,059		552,531	
Treasury stock, at cost, 41,309 shares at June 30, 2019 and December 31, 2018		(234)		(234)	
Accumulated deficit		(599,713)		(528,345)	
Total stockholders' equity		13,205		24,036	
	\$	106,585	\$	83,616	
Total liabilities and stockholders' equity	Ψ	100,505	Ψ	65,010	

The accompanying notes are an integral part of the condensed consolidated financial statements.

Condensed Consolidated Statements of Operations (in thousands, except share and per share amounts) (Unaudited)

		Three months ended June 30,				Six months ended June 30,			
		2019		2018		2019		2018	
License revenue	\$	38	\$	38	\$	76	\$	76	
Costs and expenses:									
Research and development:									
Non-cash stock expense associated with in-licensing agreements		100		3,000		100		4,000	
Noncash compensation		1,352		888		2,841		3,747	
Other research and development		31,414		34,812		62,310		65,971	
Total research and development	_	32,866	_	38,700		65,251		73,718	
General and administrative:									
Noncash compensation		405		3,375		797		7,854	
Other general and administrative		2,311		2,308		4,260		4,426	
Total general and administrative		2,716		5,683		5,057		12,280	
Total costs and expenses		35,582		44,383		70,308		85,998	
Operating loss		(35,544)		(44,345)		(70,232)		(85,922)	
Other (income) expense:									
Interest expense		1,077		223		1,851		436	
Other income		(408)		(426)		(715)		(687)	
Total other (income) expense, net		669		(203)	_	1,136		(251)	
Net loss	\$	(36,213)	\$	(44,142)	\$	(71,368)	\$	(85,671)	
Basic and diluted net loss per common share	\$	(0.42)	\$	(0.59)	\$	(0.85)	\$	(1.18)	
Weighted average shares used in computing basic and diluted net loss per common share	_	86,800,017	_	74,256,348	_	84,002,700		72,456,657	

The accompanying notes are an integral part of the condensed consolidated financial statements.

Condensed Consolidated Statements of Stockholders' Equity (in thousands, except share amounts)
(Unaudited)

Treasury Stock

Common Stock

	Shares	Amount	Additional paid-in capital	Shares	Amount	Accumulated Deficit	Total
Balance at January 1, 2018	73,181,750	\$ 73	\$ 422,017	41,309	\$ (234)	\$ (354,863)	\$ 66,993
Issuance of restricted stock	87,500	*	*				
Forfeiture of restricted stock	(123,911)	*	*				
Issuance of common stock in At-the- Market offerings (net of offering costs of \$0.9 million)	3,989,344	4	52,426				52,430
Compensation in respect of restricted stock granted to employees, directors and consultants			7,337				7,337
Net loss						(41,529)	(41,529)
Balance at March 31, 2018	77,134,683	\$ 77	\$ 481,781	41,309	\$ (234)	\$ (396,392)	\$ 85,232
Issuance of restricted stock	1,333,011	1	(1)				
Forfeiture of restricted stock	(6,750)	*	*				
Issuance of common stock in At-the- Market offerings (net of offering costs of \$1.9 million) Compensation in respect of restricted	3,744,605	4	52,018				52,022
stock granted to employees, directors and consultants			4,264				4,264
Shares issued in connection with in- licensing agreements	333,868	*	4,000				4,000
Net loss	333,808		4,000			(44,142)	(44,142)
Balance at June 30, 2018	82,539,417	\$ 82	\$ 542,062	41,309	\$ (234)		\$ 101,376
Balance at June 30, 2010							
	Commo	on Stock		Treasu	ry Stock		
			Additional				
	Shares	Amount	Additional paid-in capital	Shares	Amount	Accumulated Deficit	Total
Balance at January 1, 2019	83,911,855	\$ 84	paid-in capital \$ 552,531	Shares 41,309	Amount \$ (234)	Deficit	Total \$ 24,036
Issuance of restricted stock	83,911,855 23,000	\$ 84	paid-in capital \$ 552,531 *	41,309	\$ (234)	Deficit	\$ 24,036
•	83,911,855 23,000 	\$ 84	paid-in capital \$ 552,531	41,309	\$ (234)	Deficit	\$ 24,036
Issuance of restricted stock	83,911,855 23,000	\$ 84	paid-in capital \$ 552,531 *	41,309	\$ (234)	Deficit	\$ 24,036
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million)	83,911,855 23,000 	\$ 84 *	paid-in capital \$ 552,531 * 993	41,309	\$ (234)	Deficit	\$ 24,036
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted	83,911,855 23,000 (67,628)	\$ 84 *	paid-in capital \$ 552,531 * 993	41,309	\$ (234)	Deficit	\$ 24,036 993
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors	83,911,855 23,000 (67,628)	\$ 84 *	paid-in capital \$ 552,531 * 993 *	41,309	\$ (234)	Deficit	\$ 24,036 993 27,499
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted	83,911,855 23,000 (67,628)	\$ 84 *	paid-in capital \$ 552,531 * 993	41,309	\$ (234)	Deficit	\$ 24,036 993
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss	83,911,855 23,000 (67,628)	\$ 84 *	paid-in capital \$ 552,531 * 993 * 27,494 1,882	41,309	\$ (234)	Deficit \$ (528,345) (35,156)	\$ 24,036 993 27,499
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019	83,911,855 23,000 (67,628) 4,715,000	\$ 84 * * 5	paid-in capital \$ 552,531 * 993 * 27,494 1,882 582,900	41,309 	\$ (234) 	Deficit \$ (528,345) (35,156)	\$ 24,036 993 27,499 1,882 (35,156)
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019 Issuance of restricted stock Forfeiture of restricted stock	83,911,855 23,000 (67,628) 4,715,000	\$ 84 * * 5	paid-in capital \$ 552,531 * 993 * 27,494 1,882	41,309 	\$ (234) (234)	Deficit \$ (528,345) (35,156) (563,501)	\$ 24,036 993 27,499 1,882 (35,156)
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019 Issuance of restricted stock Forfeiture of restricted stock Issuance of common stock in public offering Issuance of common stock in At-the-	83,911,855 23,000 (67,628) 4,715,000 88,582,227 1,245,080	\$ 84 * * 5 89	paid-in capital \$ 552,531 * 993 * 27,494 1,882 582,900 (1)	41,309	\$ (234) (234)	Deficit \$ (528,345) (35,156) (563,501)	\$ 24,036 993 27,499 1,882 (35,156)
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019 Issuance of restricted stock Forfeiture of restricted stock Issuance of common stock in public offering Issuance of common stock in At-the-Market offerings (net of offering costs of \$0.5 million)	83,911,855 23,000 (67,628) 4,715,000 88,582,227 1,245,080	\$ 84 * * 5 89	paid-in capital \$ 552,531 * 993 * 27,494 1,882 582,900 (1) *	41,309	\$ (234) (234)	Deficit \$ (528,345) (35,156) (563,501)	\$ 24,036
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019 Issuance of restricted stock Forfeiture of restricted stock Issuance of common stock in public offering Issuance of common stock in At-the-Market offerings (net of offering costs	83,911,855 23,000 (67,628) 4,715,000 88,582,227 1,245,080 (38,418)	\$ 84 * * 5	paid-in capital \$ 552,531 * 993 * 27,494 1,882 582,900 (1) *	41,309	\$ (234) (234)	Deficit \$ (528,345)	\$ 24,036
Issuance of restricted stock Warrants issued with debt financing Forfeiture of restricted stock Issuance of common stock in public offering (net of offering costs of \$0.2 million) Compensation in respect of restricted stock granted to employees, directors and consultants Net loss Balance at March 31, 2019 Issuance of restricted stock Forfeiture of restricted stock Issuance of common stock in public offering Issuance of common stock in At-the-Market offerings (net of offering costs of \$0.5 million) Compensation in respect of restricted stock granted to employees, directors	83,911,855 23,000 (67,628) 4,715,000 88,582,227 1,245,080 (38,418)	\$ 84 * * 5	paid-in capital \$ 552,531 * 993 * 27,494 1,882 582,900 (1) * 11 28,392	41,309	\$ (234) (234)	Deficit \$ (528,345) (35,156) (563,501) (36,212)	\$ 24,036 993 27,499 1,882 (35,156) 19,254 11 28,395 1,757 (36,212)

The accompanying notes are an integral part of the condensed consolidated financial statements.

* Amount less than one thousand dollars.

Condensed Consolidated Statements of Cash Flows (unaudited) (in thousands) (Unaudited)

	Six months ended June 30,			
		2019		2018
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net loss	\$	(71,368)	\$	(85,671)
Adjustments to reconcile net loss to net cash used in operating activities:				
Noncash stock compensation expense		3,639		11,601
Noncash licensing expense		100		4,000
Depreciation		46		41
Amortization of premium on investment securities		(149)		(9)
Amortization of debt issuance costs		308		_
Non-cash change in lease liability and right-of-use asset		902		
Change in fair value of notes payable and accrued interest		78		82
Changes in assets and liabilities:		((2		(1.012)
Decrease (increase) in prepaid research and development and other current assets		663		(1,213)
Decrease in leasehold interest		91		53
(Increase) decrease in accrued interest receivable		(10)		22
(Decrease) increase in accounts payable and accrued expenses		(5,588)		8,898
Decrease in lease liability		(678)		_
Increase in interest payable		443		_
Increase in other current liabilities		2,388		46
Increase in deferred rent				46
Decrease in deferred revenue		(76)	_	(76)
Net cash used in operating activities		(69,211)		(62,226)
CASH FLOWS FROM INVESTING ACTIVITIES:				
Purchases of equipment		(32)		(58)
Investment in held-to-maturity securities		(14,146)		(6,965)
Proceeds from maturity of short-term securities		13,500		14,400
Net cash (used in) provided by investing activities		(678)		7,377
CASH FLOWS FROM FINANCING ACTIVITIES:				
Proceeds from sale of common stock, net		55,965		104,453
Proceeds from debt financings		29,679		_
Financing costs paid		(480)		_
Net cash provided by financing activities		85,164		104,453
Net cash provided by inhancing activities		03,101		101,133
NET INCREASE IN CASH, CASH EQUIVALENTS AND RESTRICTED CASH		15,275		49,604
CASH, CASH EQUIVALENTS AND RESTRICTED CASH AT BEGINNING OF PERIOD		43,199		57,305
GAGIL GAGILEOLINALENTS AND RESTRICTED GAGILAT END OF BERIOD	\$	58,474	\$	106,909
CASH, CASH EQUIVALENTS AND RESTRICTED CASH AT END OF PERIOD	Ψ <u></u>	30,474	Ψ	100,707
Reconciliation to amounts on consolidated balance sheets:				
Cash and cash equivalents	\$	57,228	\$	105,672
Restricted cash		1,246		1,237
Total cash, cash equivalents and restricted cash	\$	58,474	\$	106,909
NONCASH TRANSACTIONS				
Accrued offering costs	\$	60	\$	_
Deferred financing costs	\$	988	\$	_
Warrants issued with debt financing	\$	993	\$	_
Shares to be issued in connection with in-licensing	\$	100	\$	
	Ψ	100	Ψ	

The accompanying notes are an integral part of the condensed consolidated financial statements.

Notes to Condensed Consolidated Financial Statements (unaudited)

Unless the context requires otherwise, references in this report to "TG," the "Company," "we," "us" and "our" refer to TG Therapeutics, Inc. and our subsidiaries.

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Description of Business

We are a biopharmaceutical company dedicated to developing and delivering medicines for patients with B-cell mediated diseases, including Chronic Lymphocytic Leukemia (CLL), non-Hodgkin Lymphoma (NHL) and Multiple Sclerosis (MS). We have developed a robust B-cell directed research and development (R&D) platform for identification of key B-cell pathways of interest and rapid clinical testing. Currently, we have five B-cell targeted drug candidates in clinical development, with the lead two therapies, ublituximab (TG-1101) and umbralisib (TGR-1202), in pivotal trials for CLL and NHL, with ublituximab also in pivotal trials for MS. Ublituximab is a novel anti-CD20 monoclonal antibody (mAb) that has been glycoengineered for enhanced potency over first generation antibodies. Umbralisib is an oral, once daily inhibitor of PI3K delta. Umbralisib also uniquely inhibits CK1-epsilon, which may allow it to overcome certain tolerability issues associated with first generation PI3K delta inhibitors. When used together in combination therapy, ublituximab and umbralisib are referred to as "U2". Additionally, we have recently brought into Phase 1 clinical development TG-1501, an anti-PD-L1 monoclonal antibody, TG-1701, a covalently-bound Bruton's Tyrosine Kinase (BTK) inhibitor, and TG-1801, an anti-CD47/CD19 bispecific antibody.

We also actively evaluate complementary products, technologies and companies for in-licensing, partnership, acquisition and/or investment opportunities. To date, we have not received approval for the sale of any of our drug candidates in any market and, therefore, have not generated any product sales from our drug candidates.

The accompanying unaudited condensed consolidated financial statements were prepared in accordance with U.S. generally accepted accounting principles, or GAAP, for interim financial information and with the instructions to Quarterly Report on Form 10-Q and Article 10 of Regulation S-X of the Exchange Act. Accordingly, they may not include all of the information and footnotes required by GAAP for complete financial statements. All adjustments that are, in the opinion of management, of a normal recurring nature and are necessary for a fair presentation of the condensed consolidated financial statements have been included. Nevertheless, these condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements contained in our Annual Report on Form 10-K for the year ended December 31, 2018. The accompanying condensed December 31, 2018 balance sheet has been derived from these statements. The results of operations for the six months ended June 30, 2019 are not necessarily indicative of the results that may be expected for the entire fiscal year or any other interim period.

In December 2018, the Company created an Australian corporation, TG Therapeutics AUS Pty Ltd, as a wholly-owned subsidiary. This corporation's functional currency, the Australian dollar, is also its reporting currency, and its financial statements are translated to U.S. dollars, the Company's reporting currency, prior to consolidation. The accompanying consolidated financial statements include the accounts of the Company and its subsidiaries, and all intercompany accounts and transactions have been eliminated in consolidation.

Liquidity and Capital Resources

We have incurred operating losses since our inception, expect to continue to incur operating losses for the foreseeable future, and may never become profitable. As of June 30, 2019, we have an accumulated deficit of approximately \$599.7 million.

Our major sources of cash have been proceeds from the private placement and public offering of equity securities, as well as debt financings. We have not yet commercialized any of our drug candidates and cannot be sure if we will ever be able to do so. Even if we commercialize one or more of our drug candidates, we may not become profitable. Our ability to achieve profitability depends on many factors, including our ability to obtain regulatory approval for our drug candidates; successfully complete any post-approval regulatory obligations; and successfully commercialize our drug candidates alone or in partnership. We may continue to incur substantial operating losses even if we begin to generate revenues from our drug candidates.

As of June 30, 2019, we had \$85.0 million in cash and cash equivalents, and investment securities. The Company believes its cash, cash equivalents, and investment securities on hand as of June 30, 2019 along with the additional capital raised in the third quarter of 2019 (see Note 5) will be sufficient to fund the Company's planned operations into the fourth quarter of 2020. The actual amount of cash that we will need to operate is subject to many factors, including, but not limited to, the timing, design and conduct of clinical trials for our drug candidates. We are dependent upon significant future financing to provide the cash necessary to execute our current operations, including the commercialization of any of our drug candidates.

Our common stock is listed on the Nasdaq Capital Market and trades under the symbol "TGTX."

Recently Issued Accounting Standards

In July 2018, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") No. 2018-11, "Leases - Targeted Improvements" ("ASU 2018-11") as an update to ASU 2016-02, Leases ("ASU 2016-02" or "Topic 842") issued on February 25, 2016. ASU 2016-02 is effective for public business entities for fiscal years beginning January 1, 2019. ASU 2016-02 required companies to adopt the new leases standard at the beginning of the earliest period presented in the financial statements, which is January 1, 2017, using a modified retrospective transition method where lessees must recognize lease assets and liabilities for all leases even though those leases may have expired before the effective date of January 1, 2017. Lessees must also provide the new and enhanced disclosures for each period presented, including the comparative periods.

ASU 2018-11 provides an entity with an additional (and optional) transition method to adopt the new leases standard. Under this new transition method, an entity initially applies the new lease standard at the adoption date and recognizes a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption. Consequently, an entity's reporting for the comparative periods presented in the financial statements in which it adopts the new lease standard will continue to be in accordance with ASC 840, "Leases" ("ASC 840"). An entity that elects this additional (and optional) transition method must provide the required ASC 840 disclosures for all periods that continue to be in accordance with ASC 840. The amendments do not change the existing disclosure requirements in ASC 840.

ASU 2018-11 is effective for public business entities for fiscal years beginning after December 15, 2018, and interim periods within those fiscal years, with earlier adoption permitted. The Company adopted ASU 2018-11 on January 1, 2019 using a modified retrospective method and will not restate comparative periods. We elected the package of practical expedients permitted under the transition guidance, which allows us to carryforward our historical lease classification and our assessment on whether a contract is or contains a lease. The adoption of this guidance resulted in the addition of material balances of right of use assets and lease liability to our consolidated balance sheets at January 1, 2019, primarily relating to our lease of office space (see Note 8). The impact to our consolidated statements of operations was not material as a result of this standard.

In June 2018, the FASB issued ASU No. 2018-07, "Improvements to Nonemployee Share-Based Payment Accounting" ("ASU 2018-07"). ASU 2018-07 expands the scope of FASB Topic 718, Compensation – Stock Compensation ("Topic 718") to include share-based payment transactions for acquiring goods and services from nonemployees. An entity should only remeasure equity-classified awards for which a measurement date has not been established through a cumulative-effect adjustment to retained earnings as of the beginning of the fiscal year of adoption. Upon transition, the entity is required to measure these nonemployee awards at fair value as of the adoption date. The entity must not remeasure assets that are completed. Disclosures required at transition include the nature of and reason for the change in accounting principle and, if applicable, quantitative information about the cumulative effect of the change on retained earnings or other components of equity.

ASU 2018-07 is effective for public business entities for fiscal years beginning after December 15, 2018, including interim periods within that fiscal year. The Company adopted ASU 2018-07 on January 1, 2019. The adoption of ASU 2018-07 did not have a material effect on our consolidated financial statements as of January 1, 2019. The adoption of ASU 2018-07 had no impact on nonemployee performance awards as they are measured based on the outcome that is probable.

Other pronouncements issued by the FASB or other authoritative accounting standards group with future effective dates are either not applicable or not significant to our consolidated financial statements.

Use of Estimates

The preparation of financial statements in conformity with U.S. generally accepted accounting principles ("GAAP") requires management to make estimates and judgments that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the applicable reporting period. On an ongoing basis, we evaluate our estimates and judgments, including those related to accrued expenses and stock-based compensation. Actual results could differ from those estimates. Such differences could be material to the financial statements.

Cash and Cash Equivalents

We treat liquid investments with original maturities of less than three months when purchased as cash and cash equivalents.

Restricted Cash

We record cash pledged or held in trust as restricted cash. As of June 30, 2019 and December 31, 2018, we have approximately \$ 1.2 million of restricted cash pledged to secure a line of credit as a security deposit for an Office Agreement (see Note 8).

Investment Securities

Investment securities at June 30, 2019 and December 31, 2018 consist of short-term government securities. We classify these securities as held-to-maturity. Held-to-maturity securities are those securities in which we have the ability and intent to hold the security until maturity. Held-to-maturity securities are recorded at amortized cost, adjusted for the amortization or accretion of premiums or discounts. Premiums and discounts are amortized or accreted over the life of the related held-to-maturity security as an adjustment to yield using the effective interest method.

A decline in the market value of any investment security below cost, that is deemed to be other than temporary, results in a reduction in the carrying amount to fair value. The impairment is charged to operations and a new cost basis for the security is established. Other-than-temporary impairment charges are included in interest and other income (expense), net. Unrealized gains, if determined to be temporary, are included in accumulated other comprehensive income in equity. Dividend and interest income are recognized when earned.

Credit Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist primarily of cash and cash equivalents and short-term investments. The Company maintains its cash and cash equivalents and short-term investments with high-credit quality financial institutions. At times, such amounts may exceed federally-insured limits.

Revenue Recognition

Effective January 1, 2018, the Company began recognizing revenue under ASC Topic 606, *Revenue from Contracts with Customers* ("ASC 606"), using the modified retrospective transition method. The impact of adopting the new revenue standard was not material to our consolidated financial statements and there was no adjustment to beginning retained earnings on January 1, 2018. The core principle of this new revenue standard is that a company should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. The following five steps are applied to achieve that core principle:

- Step 1: Identify the contract with the customer
- Step 2: Identify the performance obligations in the contract
- Step 3: Determine the transaction price
- Step 4: Allocate the transaction price to the performance obligations in the contract
- Step 5: Recognize revenue when the company satisfies a performance obligation

In order to identify the performance obligations in a contract with a customer, a company must assess the promised goods or services in the contract and identify each promised good or service that is distinct. A performance obligation meets ASC 606's definition of a "distinct" good or service (or bundle of goods or services) if both of the following criteria are met:

- The customer can benefit from the good or service either on its own or together with other resources that are readily available to the customer (i.e., the good or service is capable of being distinct).
- The entity's promise to transfer the good or service to the customer is separately identifiable from other promises in the contract (i.e., the promise to transfer the good or service is distinct within the context of the contract).

If a good or service is not distinct, the good or service is combined with other promised goods or services until a bundle of goods or services is identified that is distinct.

The transaction price is the amount of consideration to which an entity expects to be entitled in exchange for transferring promised goods or services to a customer, excluding amounts collected on behalf of third parties (for example, some sales taxes). The consideration promised in a contract with a customer may include fixed amounts, variable amounts, or both. Variable consideration is included in the transaction price only to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is subsequently resolved.

The transaction price is allocated to each performance obligation on a relative standalone selling price basis. The transaction price allocated to each performance obligation is recognized when that performance obligation is satisfied, at a point in time or over time as appropriate.

Research and Development Costs

Generally, research and development costs are expensed as incurred. Nonrefundable advance payments for goods or services that will be used or rendered for future research and development activities are deferred and amortized over the period that the goods are delivered or the related services are performed, subject to an assessment of recoverability. We make estimates of costs incurred in relation to external clinical research organizations, or CROs, and clinical site costs. We analyze the progress of clinical trials, including levels of patient enrollment, invoices received and contracted costs when evaluating the adequacy of the amount expensed and the related prepaid asset and accrued liability. Significant judgments and estimates must be made and used in determining the accrued balance and expense in any accounting period. We review and accrue CRO expenses and clinical trial study expenses based on work performed and rely upon estimates of those costs applicable to the stage of completion of a study. Accrued CRO costs are subject to revisions as such trials progress to completion. Revisions are charged to expense in the period in which the facts that give rise to the revision become known. With respect to clinical site costs, the financial terms of these agreements are subject to negotiation and vary from contract to contract. Payments under these contracts may be uneven, and depend on factors such as the achievement of certain events, the successful recruitment of patients, the completion of portions of the clinical trial or similar conditions. The objective of our policy is to match the recording of expenses in our financial statements to the actual services received and efforts expended. As such, expense accruals related to clinical site costs are recognized based on our estimate of the degree of completion of the event or events specified in the specific clinical study or trial contract.

Prepaid research and development in our consolidated balance sheets includes, among other things, certain costs to third party service providers related to development and manufacturing services as well as clinical development. These agreements often require payments in advance of services performed or goods received. Accordingly, as of June 30, 2019 and December 31, 2018, we recorded approximately \$8.5 million and \$9.7 million, respectively, in prepaid research and development related to such advance agreements.

Income Taxes

Income taxes are accounted for under the asset and liability method. Deferred tax assets and liabilities are recognized for the future tax consequences attributable to temporary differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases, operating losses and tax credit carryforwards. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in operations in the period that includes the enactment date. If the likelihood of realizing the deferred tax assets or liability is less than "more likely than not," a valuation allowance is then created.

We, and our subsidiaries, file income tax returns in the U.S. Federal jurisdiction and in various states. We have tax net operating loss carryforwards that are subject to examination for a number of years beyond the year in which they were generated for tax purposes. Since a portion of these net operating loss carryforwards may be utilized in the future, many of these net operating loss carryforwards will remain subject to examination. We recognize interest and penalties related to uncertain income tax positions in income tax expense.

Stock-Based Compensation

We recognize all stock-based payments to employees and non-employee directors (as compensation for service) as noncash compensation expense in the condensed consolidated financial statements based on the fair values of such payments. Stock-based compensation expense recognized each period is based on the value of the portion of stock-based payment awards that is ultimately expected to vest during the period. Forfeitures are estimated at the time of grant and revised, if necessary, in subsequent periods if actual forfeitures differ from those estimates.

In addition, because some of the options, restricted stock and warrants issued to employees, consultants and other third-parties vest upon achievement of certain milestones, the total expense is uncertain. Compensation expense for such awards that vest upon the achievement of milestones is recognized when the achievement of such milestones becomes probable.

Basic and Diluted Net Loss Per Common Share

Basic net loss per share of our common stock is calculated by dividing net loss applicable to the common stock by the weighted average number of our common stock outstanding for the period. Diluted net loss per share of common stock is the same as basic net loss per share of common stock since potentially dilutive securities from stock options, stock warrants and convertible preferred stock would have an antidilutive effect either because we incurred a net loss during the periods presented or because such potentially dilutive securities were out of the money and the Company realized net income during the periods presented. The amounts of potentially dilutive securities excluded from the calculation were 7,844,712 and 5,165,890 for the three and six months ended June 30, 2019 and 2018, respectively.

The following outstanding shares of potentially dilutive securities were excluded from the computation of net loss per share attributable to common stockholders for the periods presented because including them would have been antidilutive:

	Three and Six Months	s Ended June 30,
	2019	2018
Unvested restricted stock	5,137,196	4,589,940
Stock options	2,543,710	560,000
Warrants	147,058	
Shares issuable upon note conversion	16,748	15,950
Total	7,844,712	5,165,890

Long-Lived Assets and Goodwill

Long-lived assets are reviewed for potential impairment when circumstances indicate that the carrying value of long-lived tangible and intangible assets with finite lives may not be recoverable. Management's policy in determining whether an impairment indicator exists, a triggering event, comprises measurable operating performance criteria as well as qualitative measures. If an analysis is necessitated by the occurrence of a triggering event, we make certain assumptions in determining the impairment amount. If the carrying amount of an asset exceeds its estimated future cash flows, an impairment charge is recognized.

Goodwill is reviewed for impairment annually, or earlier when events arise that could indicate that an impairment exists. We test for goodwill impairment using a two-step process. The first step compares the fair value of the reporting unit with the unit's carrying value, including goodwill. When the carrying value of the reporting unit is greater than fair value, the unit's goodwill may be impaired, and the second step must be completed to measure the amount of the goodwill impairment charge, if any. In the second step, the implied fair value of the reporting unit's goodwill is compared with the carrying amount of the unit's goodwill. If the carrying amount is greater than the implied fair value, the carrying value of the goodwill must be written down to its implied fair value. We will continue to perform impairment tests annually, at December 31, and whenever events or changes in circumstances suggest that the carrying value of an asset may not be recoverable.

NOTE 2 - CASH AND CASH EQUIVALENTS

The following tables summarize our cash and cash equivalents at June 30, 2019 and December 31, 2018:

(in thousands)	Ju	ne 30, 2019	December 31, 2018		
Checking and bank deposits	\$	54,997	\$	39,268	
Money market funds		2,231		2,690	
Total	\$	57,228	\$	41,958	

NOTE 3 – INVESTMENT SECURITIES

Our investments as of June 30, 2019 and December 31, 2018 are classified as held-to-maturity. Held-to-maturity investments are recorded at amortized cost.

The following tables summarize our investment securities at June 30, 2019 and December 31, 2018:

(in thousands)	June 30, 2019							
	Gross Amortized cost, unrealized as adjusted holding gains		unre	Gross ealized ng losses		nated fair value		
Short-term investments:								
Obligations of domestic governmental agencies (maturing between July 2019 and June 2020) (held-to-maturity)	\$	27,748	\$	45	\$	_	\$	27,793
Total short-term investment securities	\$	27,748	\$	45	\$		\$	27,793

	December 31, 2018							
	Amortized cost, unreal		Gross Gross unrealized unrealize holding gains holding los		alized		nated fair value	
Short-term investments:								
Obligations of domestic governmental agencies (maturing between January 2019 and November 2019) (held-to-maturity)	\$	26,943	\$	2	\$	10	\$	26,935
Total short-term investment securities	\$	26,943	\$	2	\$	10	\$	26,935

NOTE 4 – FAIR VALUE MEASUREMENTS

We measure certain financial assets and liabilities at fair value on a recurring basis in the condensed consolidated financial statements. The fair value hierarchy ranks the quality and reliability of inputs, or assumptions, used in the determination of fair value and requires financial assets and liabilities carried at fair value to be classified and disclosed in one of the following three categories:

- Level 1 quoted prices in active markets for identical assets and liabilities;
- Level 2 inputs other than Level 1 quoted prices that are directly or indirectly observable; and
- Level 3 unobservable inputs that are not corroborated by market data.

As of June 30, 2019 and December 31, 2018, the fair values of cash and cash equivalents, restricted cash, and notes and interest payable, approximate their carrying value.

At the time of our merger (we were then known as Manhattan Pharmaceuticals, Inc.) with Ariston Pharmaceuticals, Inc. ("Ariston") in March 2010, Ariston issued \$15.5 million of five-year 5% notes payable (the "5% Notes") in satisfaction of several note payable issuances. The 5% Notes and accrued and unpaid interest thereon are convertible at the option of the holder into common stock at the conversion price of \$1,125 per share. Ariston agreed to make quarterly payments on the 5% Notes equal to 50% of the net product cash flow received from the exploitation or commercialization of Ariston's product candidates, AST-726 and AST-915. We have no obligations under the 5% Notes aside from (a) 50% of the net product cash flows from Ariston's product candidates, if any, payable to noteholders; and (b) the conversion feature, discussed above.

The cumulative liability to the Ariston subsidiary including accrued and unpaid interest of the 5% Notes was approximately \$18.8 million at June 30, 2019 and \$18.4 million at December 31, 2018. No payments have been made on the 5% Notes since the merger and through June 30, 2019.

In December 2011, we elected the fair value option for valuing the 5% Notes. The fair value option was elected in order to reflect in our financial statements the assumptions that market participants use in evaluating these financial instruments.

As of December 31, 2013, as a result of expiring intellectual property rights and other factors, it was determined that net product cash flows from AST-726 were unlikely. As we have no other obligations under the 5% Notes aside from the net product cash flows and the conversion feature, the conversion feature was used to estimate the 5% Notes' fair value as of June 30, 2019 and December 31, 2018. The assumptions, assessments and projections of future revenues are subject to uncertainties, difficult to predict, and require significant judgment. The use of different assumptions, applying different judgment to inherently subjective matters and changes in future market conditions could result in significantly different estimates of fair value and the differences could be material to our condensed consolidated financial statements.

The following tables provide the fair value measurements of applicable financial liabilities as of June 30, 2019 and December 31, 2018:

(in thousands)	Financial liabilities at fair value as of June 30, 2019								
	Level 1		Level 2		Level 3		Fotal		
5% Notes	\$ 	\$		\$	145	\$	145		
Total	\$ 	\$		\$	145	\$	145		
	Finan	cial liabiliti	ies at fair va	lue as of D	ecember 31	, 2018			
	Level 1		Level 2		vel 3	Total			
5% Notes	\$ <u></u>	\$	<u></u>	\$	67	\$	67		
Total	\$ 	\$		\$	67	\$	67		

The Level 3 amounts above represent the fair value of the 5% Notes and related accrued interest.

The Company's financial instruments include cash, cash equivalents consisting of money market funds, accounts payable and debt. Cash, cash equivalents, accounts payable and debt are stated at their respective historical carrying amounts, which approximate fair value due to their short-term nature.

The following table summarizes the changes in Level 3 instruments during the six months ended June 30, 2019:

(in thousands)

Fair value at December 31, 2018	\$ 67
Interest accrued on face value of 5% Notes	456
Change in fair value of Level 3 liabilities	 (378)
Fair value at June 30, 2019	\$ 145

The change in the fair value of the Level 3 liabilities is reported in other (income) expense in the accompanying condensed consolidated statements of operations.

NOTE 5 - STOCKHOLDERS' EQUITY

Preferred Stock

Our amended and restated certificate of incorporation authorizes the issuance of up to 10,000,000 shares of preferred stock, \$ 0.001 par value, with rights senior to those of our common stock, issuable in one or more series. Upon issuance, we can determine the rights, preferences, privileges and restrictions thereof. These rights, preferences and privileges could include dividend rights, conversion rights, voting rights, terms of redemption, liquidation preferences, sinking fund terms and the number of shares constituting any series or the designation of such series, any or all of which may be greater than the rights of common stock

Common Stock

Our amended and restated certificate of incorporation authorizes the issuance of up to 150,000,000 shares of \$ 0.001 par value common stock.

In May 2017, we filed a shelf registration statement on Form S-3 (the "2017 S-3"), which was declared effective in June 2017, replacing the 2015 S-3. Under the 2017 S-3, the Company may sell up to a total of \$300 million of its securities. In connection with the 2017 S-3, we entered into an At-the-Market Issuance Sales Agreement (the "2017 ATM") with Jefferies LLC, Cantor Fitzgerald & Co., FBR Capital Markets & Co., SunTrust Robinson Humphrey, Inc., Raymond James & Associates, Inc., Ladenburg Thalmann & Co. Inc. and H.C. Wainwright & Co., LLC (each a "2017 Agent" and collectively, the "2017 Agents"), relating to the sale of shares of our common stock. Under the 2017 ATM we pay the 2017 Agents a commission rate of up to 3.0% of the gross proceeds from the sale of any shares of common stock.

In March 2019, we completed an underwritten public offering of 4,100,000 shares of our common stock (plus a 30-day underwriter overallotment option to purchase up to an additional 615,000 shares of common stock, which was exercised), at a price of \$5.87 per share. Proceeds from this offering, including the overallotment, after underwriting discounts and offering expenses were approximately \$27.5 million.

During the six months ended June 30, 2019, we sold a total of 3,616,359 shares of common stock under the 2017 ATM for aggregate total gross proceeds of approximately \$28.9 million at an average selling price of \$7.99 per share, resulting in net proceeds of approximately \$28.4 million after deducting commissions and other transactions costs.

Subsequent to the second quarter, from July 1, 2019 through August 6, 2019, we sold an aggregate of 1,473,692 shares of common stock pursuant to the 2017 ATM for aggregate total gross proceeds of approximately \$11.9 million at an average selling price of \$8.05 per share, resulting in net proceeds of approximately \$11.6 million after deducting commissions and other transactions costs.

The 2017 S-3 is currently our only active shelf registration statement. After deducting shares already sold there is approximately \$68.1 million of common stock that remains available for sale under the 2017 S-3. We may offer the securities under the 2017 S-3 from time to time in response to market conditions or other circumstances if we believe such a plan of financing is in the best interests of our stockholders. We believe that the 2017 S-3 provides us with the flexibility to raise additional capital to finance our operations as needed.

Equity Incentive Plans

The TG Therapeutics, Inc. Amended and Restated 2012 Incentive Plan ("2012 Incentive Plan") was approved by stockholders in June 2018. As of June 30, 2019, 6,637,199 shares of restricted stock and 2,543,710 options were outstanding and up to an additional 1,427,369 shares may be issued under the 2012 Incentive Plan.

Effective as of January 1, 2017, we entered into an amendment (the "Amendment") to the employment agreement entered as of December 15, 2011 (together with the Amendment, the "Employment Agreement") with Michael S. Weiss, our Executive Chairman and Chief Executive Officer and President. Under the Amendment, Mr. Weiss will remain as Chief Executive Officer and President, removing the interim status. Simultaneously, we entered into a Strategic Advisory Agreement (the "Advisory Agreement") with Caribe BioAdvisors, LLC (the "Advisor") owned by Mr. Weiss to provide the services of Mr. Weiss as Chairman of the Board and as Executive Chairman. As part of the Amendment, Mr. Weiss also agreed to forfeit 3,381,866 restricted shares previously granted under the Employment Agreement that were predominantly subject to time-based vesting over the next three years. Simultaneously, (i) Mr. Weiss was issued 418,371 restricted shares under the Employment Agreement that vest in 2018 and 2019 and (ii) the Advisor was issued 2,960,000 restricted shares under the Advisory Agreement that vested on market capitalization thresholds ranging from \$375 million to \$750 million. In accordance with GAAP, there was no incremental stock compensation expense recognition as a result of the modification.

Stock Options

The following table summarizes stock option activity for the six months ended June 30, 2019:

	Number of shares	Weighted- average exercise price		Number of average exercise Contractual		Aggregate rinsic Value
Outstanding at December 31, 2018	1,916,900	\$	6.50	9.75	\$ _	
Granted	740,000		6.90			
Exercised	_		_			
Forfeited	(113,190)		6.33			
Expired	<u></u>		_			
Outstanding at June 30, 2019	2,543,710	\$	6.63	9.39	\$ 5,684,217	
Expected to vest at June 30, 2019	2,543,710	\$	6.63	9.39	\$ 5,684,217	
Exercisable at June 30, 2019		\$	<u> </u>		\$ 	

Total expense associated with the stock options was approximately \$0.8 million and zero during the three months ended June 30, 2019 and 2018, respectively, and \$1.5 million and zero during the six months ended June 30, 2019 and 2018, respectively. As of June 30, 2019, the stock options outstanding include options granted to both employees and non-employees which are both time-based and milestone-based. Stock-based compensation for milestone-based options will be recorded if and when a milestone occurs.

The fair value of the Company's option awards granted during the six months ended June 30, 2019 and 2018 were estimated on the grant date using the Black-Scholes option-pricing model using the assumptions below:

	Six months	ended
	June 30, 2019	June 30, 2018
Volatility	172.99-291.61	186.99-305.77
Expected term (in years)	5.0-6.25	5.0-6.25
Risk-free rate	1.96-2.49%	2.56-2.72%
Expected dividend yield	<u> </u>	%

Restricted Stock

Certain employees, directors and consultants have been awarded restricted stock. The restricted stock vesting consists of milestone and time-based vesting.

The following table summarizes restricted share activity for the six months ended June 30, 2019:

	Number of Shares	Avera	ighted ge Grant air Value
Outstanding at December 31, 2018	6,095,692	\$	8.07
Granted	1,268,080		12.95
Vested	(620,527)		8.91
Forfeited	(106,046)		7.82
Outstanding at June 30, 2019	6,637,199	\$	7.76

Total expense associated with restricted stock grants was approximately \$1.0 million and \$4.3 million during the three months ended June 30, 2019 and 2018, respectively, and \$2.1 million, and \$11.6 million during the six months ended June 30, 2019 and 2018, respectively. As of June 30, 2019, there was approximately \$4.5 million of total unrecognized compensation cost related to unvested time-based restricted stock, which is expected to be recognized over a weighted-average period of 1.2 years. This amount does not include, as of June 30, 2019, 4,182,391 shares of restricted stock outstanding which are milestone-based and vest upon certain corporate milestones. Until the measurement date is reached for milestone awards, the total amount of compensation expense remains uncertain. We record compensation expense based on the fair value of the award at the grant date.

Stock-Based Compensation

The following table summarizes stock-based compensation expense information about restricted stock and stock options for the three and six months ended June 30, 2019:

(in thousands)	 nths ended 0, 2019	onths ended e 30, 2019
Stock-based compensation expense associated with restricted stock	\$ 950	\$ 2,127
Stock-based compensation expense associated with option grants	 807	1,512
Total	\$ 1,757	\$ 3,639

Warrants

The following table summarizes warrant activity for the six months ended June 30, 2019:

		Wei	ghted-		
	Warrants	average Varrants exercise price		•	ggregate insic Value
Outstanding at December 31, 2018		\$		\$	
Issued	147,058		4.08		
Exercised	_				
Expired					
Outstanding at June 30, 2019	147,058	\$	4.08	\$	672,055

There was no stock compensation expense related to warrants during the six months ended June 30, 2019.

NOTE 6 – OTHER LIABILITIES

The following is a summary of Notes Payable included in other current liabilities on the Company's condensed consolidated balance sheet:

	 June 30, 2019				December 31, 2018						
(in thousands)	 rrent on, net		current on, net	,	Total		rrent on, net		current on, net		Total
Convertible 5% Notes Payable	\$ 145	\$		\$	145	\$	67	\$	_	\$	67
Total	\$ 145	\$		\$	145	\$	67	\$		\$	67

16

Convertible 5% Notes Payable

The 5% Notes and accrued and unpaid interest thereon are convertible at the option of the holder into common stock at the conversion price of \$1,125 per share. We have no obligation under the 5% Notes aside from (a) 50% of the net product cash flows from Ariston's product candidates, if any, payable to noteholders; and (b) the conversion feature, discussed above. Interest accrues monthly, is added to principal on an annual basis, every March 8, and is payable at maturity, which was March 8, 2015 (see Note 4 for further details).

The cumulative liability including accrued and unpaid interest of these notes was approximately \$18.8 million at June 30, 2019 and \$18.4 million at December 31, 2018. No payments have been made on the 5% Notes as of June 30, 2019.

In December 2011, we elected the fair value option for valuing the 5% Notes. The fair value option was elected in order to reflect in our financial statements the assumptions that market participants use in evaluating these financial instruments (see Note 4 for further details).

Current Liabilities

In 2018, we entered into an agreement with a contract manufacturer for the clinical and potential commercial supply of one of our product candidates. As part of this agreement, the contract manufacturer has agreed to defer payment of certain costs and expenses under the agreement in exchange for the payment of an administrative fee. To date we have incurred expenses related to this agreement of approximately \$20.7 million as of June 30, 2019, which includes both service fees and raw material costs. No payments have been made to the contract manufacturer as of June 30, 2019. Accordingly, as of June 30, 2019, \$20.7 million is included in current liabilities in the Company's unaudited condensed consolidated balance sheet, of which \$16.7 million and \$4.0 million are due in the first and second quarter of 2020, respectively. As of December 31, 2018, \$18.4 million is included in long-term liabilities in the Company's condensed consolidated balance sheet. We will incur an administrative fee of six percent (6%) per year starting from the date of invoice issuance. For the three and six months ended June 30, 2019, we have accrued \$ million in administrative fees in connection with these costs, which has been included in interest expense in the Company's unaudited condensed consolidated statements of operations.

NOTE 7 – LONG-TERM DEBT

On February 28, 2019 (the "Closing Date"), the Company ("Borrower") entered into a term loan facility of up to \$60.0 million ("Term Loan") with Hercules Capital, Inc., ("Hercules"), the proceeds of which will be used for its ongoing research and development programs and for general corporate purposes. The Term Loan is governed by a loan and security agreement, dated February 28, 2019 (the "Loan Agreement"), which provides for up to four separate advances. The first advance of \$30.0 million was drawn on the Closing Date. Two additional advances of \$10.0 million may be drawn at the Borrower's option but subject to the clinical trial milestones, and the fourth advance of \$10.0 million, available in minimum increments of \$5.0 million, is available through December 15, 2020 subject to the approval of Hercules' investment committee.

The Term Loan will mature on March 1, 2022 (the "Loan Maturity Date"). Each advance accrues interest at a per annum rate of interest equal to the greater of either (i) the "prime rate" as reported in The Wall Street Journal plus 4.75%, and (ii) 10.25%. The Term Loan provides for interest-only payments until October 1, 2020. The interest-only period may be extended to April 1, 2021 if the Borrower, on or before September 30, 2020, achieves either the third milestone or the Company has raised at least an amount equal to \$150.0 million in unrestricted net cash proceeds from one or more equity financings, subordinated indebtedness and/or upfront proceeds from business development transactions permitted under the Loan Agreement, in each case after February 7, 2019, and prior to September 30, 2020. Thereafter, amortization payments will be payable monthly in eighteen installments (or, if the period requiring interest-only payments has been extended to April 1, 2021, in twelve installments) of principal and interest (subject to recalculation upon a change in prime rates). At its option upon seven business days' prior written notice to Hercules, the Company may prepay all or any portion greater than or equal to \$5.0 million of the outstanding advances by paying the entire principal balance (or portion thereof), all accrued and unpaid interest, subject to a prepayment charge of 3.0%, if such advance is prepaid in any of the first twelve months following the Closing Date, 1.5%, if such advance is prepaid after twelve months following the Closing Date but on or prior to twenty-four months following the Closing Date, and 0% thereafter. In addition, an end of term charge equal to 3.25% of the aggregate principal amount of the loan extended by Hercules is due on the maturity date. Amounts outstanding during an event of default shall be payable on demand and shall accrue interest at an additional rate of 4.0% per annum of the past due amount outstanding.

The Term Loan is secured by a lien on substantially all of the assets of the Borrower, other than intellectual property and contains customary covenants and representations, including a liquidity covenant, financial reporting covenant and limitations on dividends, indebtedness, collateral, investments, distributions, transfers, mergers or acquisitions, taxes, corporate changes, deposit accounts, and subsidiaries. As of June 30, 2019, the Company has been in compliance with all covenants.

The events of default under the Loan Agreement include, without limitation, and subject to customary grace periods, (1) the Borrower's failure to make any payments of principal or interest under the Loan Agreement, promissory notes or other loan documents, (2) the Borrower's breach or default in the performance of any covenant under the Loan Agreement, (3) the occurrence of a material adverse effect, (4) the Borrower making a false or misleading representation or warranty in any material respect, (5) the Borrower's insolvency or bankruptcy, (6) certain attachments or judgments on the Borrower's assets, or (7) the occurrence of any material default under certain agreements or obligations of the Borrower involving indebtedness in excess of \$750,000. If an event of default occurs, Hercules is entitled to take enforcement action, including acceleration of amounts due under the Loan Agreement.

The Loan Agreement also contains warrant coverage of 2% of the total amount funded. A warrant (the "Warrant") was issued by Borrower to Hercules to purchase 147,058 shares of common stock with an exercise price of \$4.08. The Warrant shall be exercisable for seven years from the date of issuance. Hercules may exercise the Warrant either by (a) cash or check or (b) through a net issuance conversion. The shares will be registered and freely tradeable within six months of issuance. The Company accounted for the Warrant as an equity instrument since it was indexed to the Company's common shares and met the criteria for classification in shareholders' equity. The relative fair value of the Warrant on the date of issuance was approximately \$1.0 million and was treated as an offset to the Term Loan. This amount will be amortized to interest expense using the straight line method, which approximates the effective interest method, over the life of the Term Loan.

The Company estimated the fair value of the Warrant using the Black-Scholes model based on the following key assumptions:

Exercise Price	\$ 4.08
Common share price on date of issuance	\$ 6.80
Volatility	195.9%
Risk-free interest rate	2.63%
Expected dividend yield	<u> </u>
Contractual term (in years)	7.00 years

The Company incurred financing expenses of \$2.8 million related to the Hercules Loan Agreement which are recorded as an offset to long-term debt on the Company's condensed consolidated balance sheet. The debt issuance costs are being amortized over the term of the debt using the straight line method, which approximates the effective interest method, and are included in interest expense in the Company's unaudited condensed consolidated statements of operations. Amortization of debt issuance costs was \$0.2 million and \$0.3 million for the three and six months ended June 30, 2019, respectively. At June 30, 2019, the remaining unamortized balance of debt issuance costs were \$2.5 million.

Long-term debt as of as of June 30, 2019, is as follows (in thousands):

	June	e 30, 2019
Long-term debt	\$	30,000
End of term fee		975
		30,975
Less: unamortized debt issuance costs		(2,462)
		28,513
Less: Current portion		
Long-term debt- non-current	\$	28,513

NOTE 8 - LEASES

In October 2014, we entered into an agreement (the "Office Agreement") with Fortress Biotech, Inc. ("FBIO") to occupy approximately 45% of the 24,000 square feet of New York City office space leased by FBIO, which is now our corporate headquarters. The Office Agreement requires us to pay our respective share of the average annual rent and other costs of the 15-year lease. We approximate an average annual rental obligation of \$1.1 million under the Office Agreement. We began to occupy this new space in April 2016, with rental payments beginning in the third quarter of 2016. At January 1, 2019 we recognized a lease liability and corresponding Right of Use (ROU) asset based on the present value of the remaining lease payments for all of our leased office spaces, the majority of which is comprised of our New York City office space. The present values of our lease liability and corresponding ROU asset are \$9.3 million and \$7.7 million, respectively, as of June 30, 2019. Our leases have remaining lease terms of 1 year to 12 years. One lease has a renewal option to extend the lease for an additional term of 2 years.

The initial commitment period of the 45% rate is for a period of three (3) years. We and FBIO are currently determining actual office space utilization annually and if our utilization differs from the amount we have been billed, we will either receive credits or be assessed incremental utilization charges. Also in connection with this lease, in October 2014 we pledged \$0.6 million to secure a line of credit as a security deposit for the Office Agreement, which has been recorded as restricted cash in the accompanying consolidated balance sheets. Additional collateral of \$0.6 million was pledged in April 2018 to increase the letter of credit for the office space.

The following components of lease expense are included in the Company's condensed consolidated statements of operations for the three and six months ended June 30, 2019:

(in thousands)	T)	hree months ended, June 30, 2019	Six months ended, June 30, 2019
Operating lease cost	\$	575	\$ 993
Net lease cost	\$	575	\$ 993

As of June 30, 2019, the weighted-average remaining operating lease term was 9.6 years and the weighted-average discount rate for operating leases was 10.25%. Cash paid for amounts included in the measurement of operating lease liabilities during the six months ended June 30, 2019 was \$0.7 million.

The balance sheet classification of lease liabilities was as follows:

(in thousands)	_ Jun	June 30, 2019		
Liabilities				
Lease liability – current portion	\$	1,283		
Lease liability – non-current	<u> </u>	8,062		
Total lease liability	\$	9,345		

As of June 30, 2019, the maturities of lease liabilities were as follows:

(in thousands)	Operating leases
Remainder of 2019	\$ 685
2020	1,348
2021	1,353
2022	1,374
2023	1,367
After 2023	10,282
Total lease payments	16,409
Less: Interest	(7,064)
Present value of lease liabilities(*)	\$ 9,345

(*) As our leases do not provide an implicit rate, we use our incremental borrowing rate based on the information available at commencement date and considering the term of the lease to determine the present value of lease payments. We used the incremental borrowing rate of 10.25% on February 28, 2019, for operating leases that commenced prior to that date.

NOTE 9 – LICENSE AGREEMENTS

TG-1101 (Ublituximab)

In November 2012, we entered into an exclusive (within the territory) sublicense agreement with Ildong Pharmaceutical Co. Ltd. ("Ildong") relating to the development and commercialization of ublituximab in South Korea and Southeast Asia. Under the terms of the sublicense agreement, Ildong has been granted a royalty bearing, exclusive right, including the right to grant sublicenses, to develop and commercialize ublituximab in South Korea, Taiwan, Singapore, Indonesia, Malaysia, Thailand, Philippines, Vietnam, and Myanmar.

An upfront payment of \$2.0 million, which was received in December 2012, net of \$0.3 million of income tax withholdings, is being recognized as license revenue on a straight-line basis over the life of the agreement, which is through the expiration of the last licensed patent right or 15 years after the first commercial sale of a product in such country, unless the agreement is earlier terminated, and represents the estimated period over which we will have certain ongoing responsibilities under the sublicense agreement. We recorded license revenue of approximately \$,000 for each of the three months ended June 30, 2019 and 2018, and \$,000 for each of the six months ended June 30, 2019 and 2018 and, at June 30, 2019 and December 31, 2018, have deferred revenue of approximately \$1.0 million and \$1.1 million, respectively, associated with this \$2 million payment (approximately \$,000 of which has been classified in current liabilities at June 30, 2019 and December 31, 2018).

We may receive up to an additional \$5.0 million in payments upon the achievement of pre-specified milestones. In addition, upon commercialization, Ildong will make royalty payments to us on net sales of ublituximab in the sublicense territory.

TG-1701: BTK

In January 2018, we entered into a global exclusive license agreement with Jiangsu Hengrui Medicine Co. ("Hengrui"), to acquire worldwide intellectual property rights, excluding Asia but including Japan, and for the research, development, manufacturing, and commercialization of products containing or comprising of any of Hengrui's Bruton's Tyrosine Kinase inhibitors containing the compounds of either TG-1701 (SHR1459 or EBI1459) or TG1702 (SHR1266 or EBI1266). Pursuant to the agreement, in April 2018, we paid Jiangsu an upfront fee of \$1.0 million in our common stock recorded to non-cash stock expense associated with in-licensing agreements in our condensed consolidated statement of operations. In addition, in June 2019, we paid Jiangsu the first milestone of \$0.1 million in our common stock recorded to non-cash stock expense associated with in-licensing agreements in our consolidated statement of operations. Jiangsu is eligible to receive milestone payments totaling approximately \$350 million upon and subject to the achievement of certain milestones. Various provisions allow for payments in conjunction with the agreement to be made in cash or our common stock, while others limit the form of payment. Royalty payments in the low double digits are due on net sales of licensed products and revenue from sublicenses.

TG-1801: anti-CD47/anti-CD19

In June 2018, we entered into a Joint Venture and License Option Agreement with Novimmune SA ("Novimmune") to collaborate on the development and commercialization of Novimmune's novel first-in-class anti-CD47/anti-CD19 bispecific antibody known as TG-1801 (previously NI-1701). The companies will jointly develop the product on a worldwide basis, focusing on indications in the area of hematologic B-cell malignancies. We serve as the primary responsible party for the development, manufacturing and commercialization of the product. Pursuant to the agreement, in June 2018 we paid Novimmune an upfront payment of \$3.0 million in our common stock recorded to non-cash stock expense associated with in-licensing agreements in our consolidated statement of operations. Further milestone payments will be paid based on early clinical development, and the Company will be responsible for the costs of clinical development of the product through the end of the Phase 2 clinical trials, after which the Company and Novimmune will be jointly responsible for all development and commercialization costs. The Company and Novimmune will each maintain an exclusive option, exercisable at specific times during development, for the Company to license the rights to TG-1801, in which case Novimmune is eligible to receive additional milestone payments totaling approximately \$185 million as well as tiered royalties on net sales in the high single to low double digits upon and subject to the achievement of certain milestones.

NOTE 10 - RELATED PARTY TRANSACTIONS

In October 2014, we entered into an agreement (the "Office Agreement") with Fortress Biotech, Inc. ("FBIO"), to occupy approximately 45% of the 24,000 square feet of New York City office space leased by FBIO, which is now our corporate headquarters. The Office Agreement requires us to pay our respective share of the average annual rent and other costs of the 15-year lease. We approximate an average annual rental obligation of \$1.1 million under the Office Agreement. We began to occupy this new space in April 2016, with rental payments beginning in the third quarter of 2016. At January 1, 2019, we recognized a lease liability of \$9.3 million, with a corresponding Right of Use (ROU) asset of \$7.7 million based on the present value of the remaining lease payments for all of our leased office spaces, the majority of which is comprised of our New York City office space. Mr. Weiss, our Executive Chairman and CEO, is also Executive Vice Chairman of FBIO.

Under the Office Agreement, we agreed to pay FBIO our portion of the build out costs, which have been allocated to us at the 45% rate mentioned above. The allocated build-out costs have been recorded in Leasehold Interest, net on the Company's consolidated balance sheets and will be amortized over the 15-year term of the Office Agreement. The initial commitment period of the 45% rate is for a period of three (3) years. We and FBIO are currently determining actual office space utilization annually and if our utilization differs from the amount we have been billed, we will either receive credits or be assessed incremental utilization charges. Also, in connection with this lease, in October 2014 we pledged \$0.6 million to secure a line of credit as a security deposit for the Office Agreement, which has been recorded as restricted cash in the accompanying consolidated balance sheets. Additional collateral of \$0.6 million was pledged in April 2018 to increase the letter of credit for the office space.

In July 2015, we entered into a Shared Services Agreement (the "Shared Services Agreement") with FBIO to share the cost of certain services such as facilities use, personnel costs and other overhead and administrative costs. This Shared Services Agreement requires us to pay our respective share of services utilized. In connection with the Shared Services Agreement, we incurred expenses of approximately \$0.4 million and \$1.1 million for shared services for the six months ended June 30, 2019 and 2018, respectively, and expenses of approximately \$0.2 million and \$0.8 million for the three months ended June 30, 2019 and 2018, primarily related to shared personnel.

In March 2015, we entered into a Global Collaboration Agreement ("Collaboration Agreement") with Checkpoint for the development and commercialization of anti-PD-L1 and anti-GITR antibody research programs in the field of hematological malignancies. The Collaboration Agreement was amended in June 2019 and upon execution of the amendment we incurred an upfront fee of \$1.0 million. We incurred expenses of approximately \$1.4 million and \$0.5 million for the six months ended June 30, 2019 and 2018, respectively, and expenses of approximately \$1.1 million and \$0.2 million for the three months ended June 30, 2019 and 2018, the majority of which relates to the upfront fee of \$1.0 million resulting from the amendment. The relevant expenses are recorded in other research and development in the accompanying condensed consolidated statement of operations.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis contain forward-looking statements about our plans and expectations of what may happen in the future. Forward-looking statements are based on a number of assumptions and estimates that are inherently subject to significant risks and uncertainties, and our results could differ materially from the results anticipated by our forward-looking statements as a result of many known or unknown factors, including, but not limited to, those factors discussed in "Risk Factors." See also the "Special Cautionary Notice Regarding Forward-Looking Statements" set forth at the beginning of this report.

You should read the following discussion and analysis in conjunction with the unaudited condensed consolidated financial statements, and the related footnotes thereto, appearing elsewhere in this report, and in conjunction with management's discussion and analysis and the audited consolidated financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2018.

OVERVIEW

We are a biopharmaceutical company dedicated to developing and delivering medicines for patients with B-cell mediated diseases, including Chronic Lymphocytic Leukemia (CLL), non-Hodgkin Lymphoma (NHL) and Multiple Sclerosis (MS). We have developed a robust B-cell directed research and development (R&D) platform for identification of key B-cell pathways of interest and rapid clinical testing. Currently, we have five B-cell targeted drug candidates in clinical development, with the lead two therapies, ublituximab (TG-1101) and umbralisib (TGR-1202), in pivotal trials for CLL and NHL, with ublituximab also in pivotal trials for MS. Ublituximab is a novel anti-CD20 monoclonal antibody (mAb) that has been glycoengineered for enhanced potency over first generation antibodies. Umbralisib is an oral, once daily inhibitor of PI3K delta. Umbralisib also uniquely inhibits CK1-epsilon, which may allow it to overcome certain tolerability issues associated with first generation PI3K delta inhibitors. When used together in combination therapy, ublituximab and umbralisib are referred to as "U2". Additionally, we have recently brought into Phase 1 clinical development TG-1501, an anti-PD-L1 monoclonal antibody, TG-1701, a covalently-bound Bruton's Tyrosine Kinase (BTK) inhibitor, and TG-1801, an anti-CD47/CD19 bispecific antibody.

We also actively evaluate complementary products, technologies and companies for in-licensing, partnership, acquisition and/or investment opportunities. To date, we have not received approval for the sale of any of our drug candidates in any market and, therefore, have not generated any product sales from our drug candidates.

OUR PRODUCTS UNDER DEVELOPMENT

We have leveraged our B-cell platform to develop a robust drug pipeline of both targeted orally available, potent and selective small molecule kinase inhibitors and intravenously delivered "off-the-shelf" immunotherapies that leverage the patient's own immune system to fight cancer. We currently own worldwide development and commercial rights, subject to certain limited geographical restrictions, to all of our pre-clinical and clinical programs. The following table summarizes our most advanced drug candidates:

Clinical Drug Candidate	Initial Target Disease	Stage of Development
(molecular target)		(pivotal study)
Ublituximab/TG-1101 (anti-CD20/mAb)	Chronic Lymphocytic Leukemia	Phase 3 trial (UNITY-CLL)
	Relapsing Multiple Sclerosis	Phase 3 trials (ULTIMATE I and II)
Umbralisib/TGR-1202 (PI3K delta inhibitor)	Chronic Lymphocytic Leukemia	Phase 3 trial (UNITY-CLL)
	Marginal Zone Lymphoma	Phase 2b trial (UNITY-NHL)
	Follicular Lymphoma	Phase 2b trial (UNITY-NHL)
TG-1501 (anti-PDL1)	B-cell Cancers	Phase 1 trial
TG-1701 (BTK inhibitor)	B-cell Cancers	Phase 1 trial
TG-1801 (anti-CD47/CD19)	B-cell Cancers	Phase 1 trial

Phase 3 and Registration-Directed Clinical Trial Highlights

The company has initiated and enrolled several Phase 3 and registration-directed Phase 2b clinical trials (i.e., clinical trials that may support a marketing application for approval). The following are the current Phase 3 trials and registration-directed Phase 2b clinical trials:

- UNITY-NHL Phase 2b Trial Marginal Zone Lymphoma (MZL) and Follicular Lymphoma (FL)/ Small Lymphocytic Lymphoma (SLL) Cohorts (Umbralisib Monotherapy): UNITY-NHL is a broad Phase 2b registration-directed clinical trial designed to evaluate the efficacy and safety of umbralisib monotherapy and U2 combinations in patients with previously treated NHL. The MZL and the FL/SLL single agent umbralisib cohorts of this trial are fully enrolled. The primary objective of these cohorts is to assess the efficacy of single agent umbralisib as measured by Overall Response Rate (ORR).
 - UNITY-NHL MZL Single Agent Umbralisib Cohort: The MZL cohort enrolled adult patients who had at least one prior line of therapy that included an anti-CD20 monoclonal antibody. In February 2019, we announced that the MZL cohort met the primary endpoint of ORR as determined by Independent Review Committee (IRC) for all treated patients (n=69). The results met the Company's target guidance of 40-50% ORR. Interim safety and efficacy data from the MZL cohort were presented in an oral presentation at the American Association for Cancer Research (AACR) annual meeting on April 1, 2019. The data presented included safety and tolerability data on all 69 treated patients (safety population) and efficacy data on 42 patients who were enrolled at least 9 cycles (28 day cycles) prior to the data cut-off date (interim efficacy population). The safety population had a median duration of exposure of 6.9 months and no unexpected toxicities were observed. Analysis of the interim efficacy population showed an ORR by IRC of 52%, including 19% CR rate, an 88% clinical benefit rate by IRC (defined as patients obtaining Complete Response + Partial Response + Stable Disease) and a median duration of exposure of 10.1 months. These results were also presented in oral presentations during the 2019 American Society of Clinical Oncology (ASCO) annual meeting and the 2019 International Conference on Malignant Lymphoma (ICML) both held in June of 2019. Previously, in January 2019, the U.S. Food and Drug Administration (FDA) granted Breakthrough Therapy Designation (BTD) to umbralisib for the treatment of adult patients with MZL who have received at least one prior anti-CD20 regimen. The BTD was based on interim data from the MZL cohort of the UNITY-NHL trial. In April 2019, the FDA granted orphan drug designation to umbralisib for the treatment of patients with any of the three types of marginal zone lymphoma (MZL): nodal, extranodal, and splenic MZL. In June of 2019, we had a BTD meeting with the FDA to discuss the MZL submission strategy. Based on this meeting, we anticipate initiating a New Drug Application (NDA) submission for patients with previously treated MZL by year-end 2019.
 - UNITY-NHL FL/SLL Single Agent Umbralisib Cohort: The FL/SLL cohort enrolled adult patients who had two or more prior lines of therapy that included an anti-CD20 monoclonal antibody and an alkylating agent. This cohort is fully enrolled with approximately 125 patients.
 - UNITY-NHL Additional Cohorts: There are additional exploratory cohorts of the UNITY-NHL trial focused on Diffuse Large B-Cell Lymphoma (DLBCL) and Mantle Cell Lymphoma (MCL). In total, there are currently four cohorts in the UNITY-NHL trial including, MZL, FL/SLL, DLBCL, and MCL. Each cohort is enrolled and evaluated separately from the others.
- UNITY-CLL Phase 3 Trial Evaluating Umbralisib plus Ublituximab (U2): UNITY-CLL is a global Phase 3 randomized controlled clinical trial comparing the U2 combination to an active control arm of obinutuzumab plus chlorambucil in patients with both treatment naïve and relapsed or refractory CLL. Two additional arms evaluating single agent ublituximab and single agent umbralisib were also enrolled for purposes of evaluating contribution in the U2 combination regimen. The primary endpoint for this study is Progression Free Survival (PFS) which we intend to use to support a submission for approval of the U2 combination in CLL. The study completed enrollment in October 2017 with over 600 patients across the four treatment arms, with approximately 420 patients in the U2 and the active control arm combined. This trial is conducted under Special Protocol Assessment (SPA) with FDA. In March 2019, the UNITY-CLL DSMB met to conduct a pre-planned futility analysis of PFS. The DSMB determined that the trial was not futile and recommended the UNITY-CLL trial continue as planned. The pre-specified futility analysis of the UNITY-CLL trial did not allow for early stopping due to positive efficacy but only for lack of efficacy. During this meeting the DSMB also reviewed safety information from all 600+ CLL patients on trial, including over 300 treatment naïve and previously treated patients on umbralisib alone or in combination with ublituximab. Based on its review, no safety concerns were identified and the DSMB recommended the UNITY-CLL trial continue without modification.
- ULTIMATE I & II Trials Evaluating Single Agent Ublituximab in RMS: ULTIMATE I and ULTIMATE II are two independent Phase 3 trials. Each trial is a global, randomized, multi-center, double-blinded, double-dummy, active-controlled study comparing ublituximab to teriflunomide in subjects with relapsing forms of Multiple Sclerosis (RMS). The primary endpoint for each study is Annualized Relapse Rate (ARR) following 96 weeks of treatment which we intend to use to support a submission for approval of ublituximab in RMS. These trials are conducted under a SPA with the FDA. Full enrollment was completed in October of 2018, with approximately 1,100 subjects enrolled in both studies combined.

GENERAL CORPORATE

Our license revenues currently consist of license fees arising from our agreement with Ildong. We recognize upfront license fee revenues ratably over the estimated period in which we will have certain significant ongoing responsibilities under the sublicense agreement, with unamortized amounts recorded as deferred revenue.

We have not earned any revenues from the commercial sale of any of our drug candidates.

Our research and development expenses consist primarily of expenses related to in-licensing of new product candidates, fees paid to consultants and outside service providers for clinical and laboratory development, facilities-related and other expenses relating to the design, development, manufacture, testing and enhancement of our drug candidates and technologies including clinical trial related expenses. We expense our research and development costs as they are incurred.

Our general and administrative expenses consist primarily of salaries and related expenses for executive, finance and other administrative personnel, recruitment expenses, professional fees and other corporate expenses, including investor relations, legal activities and facilities-related expenses.

Our results of operations include non-cash compensation expenses as a result of the grants of restricted stock. Compensation expense for awards of restricted stock granted to employees and directors represents the fair value of the award recorded over the respective vesting periods of the individual awards. The expense is included in the respective categories of expense in the condensed consolidated statements of operations. We expect to continue to incur significant non-cash compensation expenses.

We recognize all stock-based payments to employees and non-employee directors (as compensation for service) as noncash compensation expense in the condensed consolidated financial statements based on the fair values of such payments. Stock-based compensation expense recognized each period is based on the value of the portion of stock-based payment awards that is ultimately expected to vest during the period. Forfeitures are estimated at the time of grant and revised, if necessary, in subsequent periods if actual forfeitures differ from those estimates.

In addition, because some of the options, restricted stock and warrants issued to employees, consultants and other third-parties vest upon achievement of certain milestones, the total expense is uncertain. Compensation expense for such awards that vest upon the achievement of milestones is recognized when the achievement of such milestones becomes probable.

Our clinical trials will be lengthy and expensive. Even if these trials show that our drug candidates are effective in treating certain diseases, there is no guarantee that we will be able to record commercial sales of any of our drug candidates in the near future, or at all. In addition, we expect losses to continue as we fund in-licensing and development of new drug candidates. As we further our development efforts, we may enter into additional third-party collaborative agreements and incur additional expenses, such as licensing fees and milestone payments. In addition, we may need to establish a commercial infrastructure required to manufacture, market and sell our medicines following approval, if any, by the FDA or a foreign health authority, which would result in incurring additional expenses. As a result, our quarterly results may fluctuate and a quarter-by-quarter comparison of our operating results may not be a meaningful indication of our future performance.

RESULTS OF OPERATIONS

Three months ended June 30, 2019 and 2018

License Revenue. License revenue was approximately \$38,000 for each of the three months ended June 30, 2019 and 2018. License revenue is related to the amortization of an upfront payment of \$2.0 million received in 2012 associated with our license agreement with Ildong. The upfront payment from Ildong will be recognized as license revenue on a straight-line basis through December 2025, which represents the estimated period over which the Company will have certain ongoing responsibilities under the sublicense agreement.

Noncash Stock Expense Associated with In-Licensing Agreement (Research and Development). Noncash stock expense associated with in-licensing agreement (research and development) amounted to \$0.1 million for the three months ended June 30, 2019, as compared to \$3.0 million during the comparable period in 2018. The expense during the three months ended June 30, 2018 was recorded in conjunction with the stock issued to Novimmune as an upfront payment for the license to the CD47/CD-19 program.

Noncash Compensation Expense (Research and Development). Noncash compensation expense (research and development) related to equity incentive grants totaled \$1.4 million for the three months ended June 30, 2019, as compared to \$0.9 million during the comparable period in 2018. The increase in noncash compensation expense was primarily due to noncash stock expense related to stock options during the period ended June 30, 2019.

Other Research and Development Expenses. Other research and development expenses decreased by \$3.4 million to \$31.4 million for the three months ended June 30, 2019, as compared to \$34.8 million for the three months ended June 30, 2018. The decrease was mainly due to the primary clinical programs for ublituximab and umbralisib having completed enrollment during 2018. We expect our other research and development costs to decrease modestly for the remainder of 2019 as our primary clinical programs near completion.

Noncash Compensation Expense (General and Administrative). Noncash compensation expense (general and administrative) related to equity incentive grants decreased by \$3.0 million to \$0.4 million for the three months ended June 30, 2019, as compared to \$3.4 million for the three months ended June 30, 2018. The decrease in noncash compensation expense was primarily related to greater compensation expense during the three months ended June 30, 2018 related to restricted stock granted to executive personnel.

Other General and Administrative Expenses. Other general and administrative expenses was \$2.3 million for the three months ended June 30, 2019 and 2018. We expect our other general and administrative expenses to remain at a comparable level for the remainder of 2019.

Interest Expense. Interest expense increased by \$0.9 million to \$1.1 million for the three months ended June 30, 2019, as compared to expense of \$0.2 million for the three months ended June 30, 2018. The increase is mainly due to the interest expense related to the Hercules financing agreement and the associated amortization of debt issuance costs for the three months ended June 30, 2019.

Other Income. Other income remained consistent between the two periods totaling \$0.4 million for the three months ended June 30, 2019 and 2018. We expect our other income to remain at a comparable level for the remainder of 2019.

Six months ended June 30, 2019 and 2018

License Revenue. License revenue was approximately \$0.1 million for each of the six months ended June 30, 2019 and 2018. License revenue for the six months ended June 30, 2019 and 2018 was related to the amortization of an upfront payment of \$2.0 million received in 2012 associated with our license agreement with Ildong.

Noncash Stock Expense Associated with In-Licensing Agreement (Research and Development). Noncash stock expense associated with in-licensing agreement (research and development) amounted to \$0.1 million for the six months ended June 30, 2019, as compared to \$4.0 million during the comparable period in 2018. The expense during the six months ended June 30, 2018 was recorded in conjunction with the stock issued to Novimmune and Jiangsu Hengrui as upfront payments for the licenses to the CD47/CD19 and BTK programs, respectively.

Noncash Compensation Expense (Research and Development). Noncash compensation expense (research and development) related to equity incentive grants totaled \$2.8 million for the six months ended June 30, 2019, as compared to \$3.7 million during the comparable period in 2018. The decrease in noncash compensation expense was primarily related to greater measurement date fair value of certain consultant restricted stock that was applicable during the period ended June 30, 2018.

Other Research and Development Expenses. Other research and development expenses decreased by \$3.7 million to \$62.3 million for the six months ended June 30, 2019, as compared to \$66.0 million for the six months ended June 30, 2018. The decrease was mainly due to the primary clinical programs for ublituximab and umbralisib having completed enrollment during 2018. We expect our other research and development costs to decrease modestly for the remainder of 2019 as our primary clinical programs near completion.

Noncash Compensation Expense (General and Administrative). Noncash compensation expense (general and administrative) related to equity incentive grants decreased by \$7.1 million to \$0.8 million for the six months ended June 30, 2019, as compared to \$7.9 million for the six months ended June 30, 2018. The decrease in noncash compensation expense was primarily related to greater compensation expense during the six months ended June 30, 2018 related to restricted stock granted to executive personnel.

Other General and Administrative Expenses. Other general and administrative expenses decreased by \$0.1 million to \$4.3 million for the six months ended June 30, 2019, as compared to \$4.4 million for the six months ended June 30, 2018. We expect our other general and administrative expenses to remain at a comparable level for the remainder of 2019.

Interest Expense. Interest expense increased by \$1.5 million to \$1.9 million for the six months ended June 30, 2019, as compared to expense of \$0.4 million for the six months ended June 30, 2018. The increase is mainly due to the interest expense related to the Hercules financing agreement and the associated amortization of debt issuance costs for the six months ended June 30, 2019.

Other Income. Other income remained consistent between the two periods totaling \$0.7 million for the six months ended June 30, 2019 and 2018. We expect our other income to remain at a comparable level for the remainder of 2019.

LIQUIDITY AND CAPITAL RESOURCES

Our primary sources of cash have been from the sale of equity securities, debt financings, the upfront payment from our Sublicense Agreement with Ildong, and warrant and option exercises. We have not yet commercialized any of our drug candidates and cannot be sure if we will ever be able to do so. Even if we commercialize one or more of our drug candidates, we may not become profitable. Our ability to achieve profitability depends on a number of factors, including our ability to obtain regulatory approval for our drug candidates, successfully complete any post-approval regulatory obligations and successfully commercialize our drug candidates alone or in partnership. We may continue to incur substantial operating losses even if we begin to generate revenues from our drug candidates.

As of June 30, 2019, we had approximately \$85.0 million in cash and cash equivalents, and investment securities. We anticipate that our cash, cash equivalents, and investment securities on hand as of June 30, 2019, combined with the additional capital raised in the third quarter of 2019, will be sufficient to fund the Company's planned operations into the fourth quarter of 2020. The actual amount of cash that we will need to operate is subject to many factors, including, but not limited to, the timing, design and conduct of clinical trials for our drug candidates. We are dependent upon significant financing to provide the cash necessary to execute our current operations, including the commercialization of any of our drug candidates.

Cash used in operating activities for the six months ended June 30, 2019 was \$69.2 million as compared to \$62.2 million for the six months ended June 30, 2018. The increase in cash used in operating activities was due primarily to increased expenditures associated with our ongoing clinical development programs and paydown of accounts payable and accrued expenses.

For the six months ended June 30, 2019, net cash used in investing activities was \$0.7 million as compared to cash provided by investing activities of \$7.4 million for the six months ended June 30, 2018. The decrease in net cash provided by investing activities was primarily due to the greater investment in treasury securities during the six months ended June 30, 2019.

For the six months ended June 30, 2019, net cash provided by financing activities of \$85.2 million related to the net proceeds from debt financings and net proceeds from the issuance of common stock as part of our 2017 ATM program and a public offering in March 2019.

OFF-BALANCE SHEET ARRANGEMENTS

We have not entered into any transactions with unconsolidated entities whereby we have financial guarantees, subordinated retained interests, derivative instruments or other contingent arrangements that expose us to material continuing risks, contingent liabilities, or any other obligations under a variable interest in an unconsolidated entity that provides us with financing, liquidity, market risk or credit risk support.

CONTRACTUAL OBLIGATIONS AND COMMITMENTS

As of June 30, 2019, we have known contractual obligations, commitments and contingencies of \$67.1 million related to our long-term debt, contract manufacturer and operating lease obligations.

Payment due by period (in thousands)

		Le	ss than 1					Mo	re than 5
	 Total	year		1-3 years		3-5 years		years	
Contractual obligations									_
Long-term debt	\$ 30,000	\$		\$	30,000	\$		\$	
Contract manufacturer	20,738		20,738						
Operating leases	 16,410		1,363		2,707		2,682		9,658
Total	\$ 67,148	\$	22,101	\$	32,707	\$	2,682	\$	9,658

The terms of certain of our licenses, royalties, development and collaboration agreements, as well as other research and development activities, require us to pay potential future milestone payments based on product development success. The above table excludes such obligations as the amount and timing of such obligations are unknown or uncertain.

CRITICAL ACCOUNTING POLICIES

The discussion and analysis of our financial condition and results of operations is based upon our consolidated financial statements, which have been prepared in accordance with U.S. generally accepted accounting principles. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amount of assets and liabilities and related disclosure of contingent assets and liabilities at the date of our financial statements and the reported amounts of revenues and expenses during the applicable period. Actual results may differ from these estimates under different assumptions or conditions.

We define critical accounting policies as those that are reflective of significant judgments and uncertainties and which may potentially result in materially different results under different assumptions and conditions. In applying these critical accounting policies, our management uses its judgment to determine the appropriate assumptions to be used in making certain estimates. These estimates are subject to an inherent degree of uncertainty. Our critical accounting policies include the following:

Revenue Recognition. Effective January 1, 2018, the Company began recognizing revenue under Accounting Standards Codification ("ASC") Topic 606, Revenue from Contracts with Customers ("ASC 606"), using the modified retrospective transition method. The impact of adopting the new revenue standard was not material to our consolidated financial statements and there was no adjustment to beginning retained earnings on January 1, 2018. The core principle of this new revenue standard is that a company should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. The following five steps are applied to achieve that core principle:

- Step 1: Identify the contract with the customer
- Step 2: Identify the performance obligations in the contract
- Step 3: Determine the transaction price
- Step 4: Allocate the transaction price to the performance obligations in the contract
- Step 5: Recognize revenue when the company satisfies a performance obligation

In order to identify the performance obligations in a contract with a customer, a company must assess the promised goods or services in the contract and identify each promised good or service that is distinct. A performance obligation meets ASC 606's definition of a "distinct" good or service (or bundle of goods or services) if both of the following criteria are met:

- The customer can benefit from the good or service either on its own or together with other resources that are readily available to the customer (i.e., the good or service is capable of being distinct).
- The entity's promise to transfer the good or service to the customer is separately identifiable from other promises in the contract (i.e., the promise to transfer the good or service is distinct within the context of the contract).

If a good or service is not distinct, the good or service is combined with other promised goods or services until a bundle of goods or services is identified that is distinct.

The transaction price is the amount of consideration to which an entity expects to be entitled in exchange for transferring promised goods or services to a customer, excluding amounts collected on behalf of third parties (for example, some sales taxes). The consideration promised in a contract with a customer may include fixed amounts, variable amounts, or both. Variable consideration is included in the transaction price only to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is subsequently resolved.

The transaction price is allocated to each performance obligation on a relative standalone selling price basis. The transaction price allocated to each performance obligation is recognized when that performance obligation is satisfied, at a point in time or over time as appropriate.

Stock Compensation. We have granted stock options and restricted stock to employees, directors and consultants, as well as warrants to other third parties. For employee, director and consultant grants the value of each option award is estimated on the date of grant using the Black-Scholes option-pricing model. The Black-Scholes model takes into account volatility in the price of our stock, the risk-free interest rate, the estimated life of the option, the closing market price of our stock and the exercise price. We base our estimates of our stock price volatility on the historical volatility of our common stock and our assessment of future volatility; however, these estimates are neither predictive nor indicative of the future performance of our stock. For purposes of the calculation, we assumed that no dividends would be paid during the life of the options and warrants. The estimates utilized in the Black-Scholes calculation involve inherent uncertainties and the application of management judgment. In addition, we are required to estimate the expected forfeiture rate and only recognize expense for those equity awards expected to vest. As a result, if other assumptions had been used, our recorded stock-based compensation expense could have been materially different from that reported. In addition, because some of the options, restricted stock and warrants issued to employees, consultants and other third-parties vest upon the achievement of certain milestones, the total expense is uncertain. Compensation expense for such awards that vest upon the achievement of milestones is recognized when the achievement of such milestone becomes probable.

Accruals for Clinical Research Organization and Clinical Site Costs. We make estimates of costs incurred in relation to external clinical research organizations, or CROs, and clinical site costs. We analyze the progress of clinical trials, including levels of patient enrollment, invoices received and contracted costs when evaluating the adequacy of the amount expensed and the related prepaid asset and accrued liability. Significant judgments and estimates must be made and used in determining the accrued balance and expense in any accounting period. We review and accrue CRO expenses and clinical trial study expenses based on work performed and rely upon estimates of those costs applicable to the stage of completion of a study. Accrued CRO costs are subject to revisions as such trials progress to completion. Revisions are charged to expense in the period in which the facts that give rise to the revision become known. With respect to clinical site costs, the financial terms of these agreements are subject to negotiation and vary from contract. Payments under these contracts may be uneven, and depend on factors such as the achievement of certain events, the successful recruitment of patients, the completion of portions of the clinical trial or similar conditions. The objective of our policy is to match the recording of expenses in our financial statements to the actual services received and efforts expended. As such, expense accruals related to clinical site costs are recognized based on our estimate of the degree of completion of the event or events specified in the specific clinical study or trial contract.

Accounting For Income Taxes. In preparing our consolidated financial statements, we are required to estimate our income taxes in each of the jurisdictions in which we operate. This process involves management estimation of our actual current tax exposure and assessment of temporary differences resulting from differing treatment of items for tax and accounting purposes. These differences result in deferred tax assets and liabilities. We must then assess the likelihood that our deferred tax assets will be recovered from future taxable income and, to the extent we believe that recovery is not likely, we must establish a valuation allowance. To the extent we establish a valuation allowance or increase this allowance in a period, we must include an expense within the tax provision in the consolidated statements of operations. Significant management judgment is required in determining our provision for income taxes, our deferred tax assets and liabilities and any valuation allowance recorded against our net deferred tax assets. We have fully offset our deferred tax assets with a valuation allowance. Our lack of earnings history and the uncertainty surrounding our ability to generate taxable income prior to the reversal or expiration of such deferred tax assets were the primary factors considered by management in maintaining the valuation allowance.

RECENTLY ISSUED ACCOUNTING STANDARDS

In July 2018, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") No. 2018-11, "Leases - Targeted Improvements" ("ASU 2018-11") as an update to ASU 2016-02, Leases ("ASU 2016-02" or "Topic 842") issued on February 25, 2016. ASU 2016-02 is effective for public business entities for fiscal years beginning January 1, 2019. ASU 2016-02 required companies to adopt the new leases standard at the beginning of the earliest period presented in the financial statements, which is January 1, 2017, using a modified retrospective transition method where lessees must recognize lease assets and liabilities for all leases even though those leases may have expired before the effective date of January 1, 2017. Lessees must also provide the new and enhanced disclosures for each period presented, including the comparative periods.

ASU 2018-11 provides an entity with an additional (and optional) transition method to adopt the new leases standard. Under this new transition method, an entity initially applies the new lease standard at the adoption date and recognizes a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption. Consequently, an entity's reporting for the comparative periods presented in the financial statements in which it adopts the new lease standard will continue to be in accordance with ASC 840, "Leases" ("ASC 840"). An entity that elects this additional (and optional) transition method must provide the required ASC 840 disclosures for all periods that continue to be in accordance with ASC 840. The amendments do not change the existing disclosure requirements in ASC 840.

ASU 2018-11 is effective for public business entities for fiscal years beginning after December 15, 2018, and interim periods within those fiscal years, with earlier adoption permitted. The Company adopted ASU 2018-11 on January 1, 2019 using a modified retrospective method and will not restate comparative periods. We elected the package of practical expedients permitted under the transition guidance, which allows us to carryforward our historical lease classification and our assessment on whether a contract is or contains a lease. The adoption of this guidance resulted in the addition of material balances of right of use assets and lease liability to our consolidated balance sheets at January 1, 2019, primarily relating to our lease of office space (see Note 8). The impact to our consolidated statements of operations was not material as a result of this standard.

In June 2018, the FASB issued ASU No. 2018-07, "Improvements to Nonemployee Share-Based Payment Accounting" ("ASU 2018-07"). ASU 2018-07 expands the scope of FASB Topic 718, Compensation – Stock Compensation ("Topic 718") to include share-based payment transactions for acquiring goods and services from nonemployees. An entity should only remeasure equity-classified awards for which a measurement date has not been established through a cumulative-effect adjustment to retained earnings as of the beginning of the fiscal year of adoption. Upon transition, the entity is required to measure these nonemployee awards at fair value as of the adoption date. The entity must not remeasure assets that are completed. Disclosures required at transition include the nature of and reason for the change in accounting principle and, if applicable, quantitative information about the cumulative effect of the change on retained earnings or other components of equity.

ASU 2018-07 is effective for public business entities for fiscal years beginning after December 15, 2018, including interim periods within that fiscal year. The Company adopted ASU 2018-07 on January 1, 2019. The adoption of ASU 2018-07 did not have a material effect on our consolidated financial statements as of January 1, 2019. The adoption of ASU 2018-07 had no impact on nonemployee performance awards as they are measured based on the outcome that is probable.

Other pronouncements issued by the FASB or other authoritative accounting standards group with future effective dates are either not applicable or not significant to our consolidated financial statements.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The primary objective of our investment activities is to preserve principal while maximizing our income from investments and minimizing our market risk. We invest in government and investment-grade corporate debt in accordance with our investment policy. Some of the securities in which we invest have market risk. This means that a change in prevailing interest rates, and/or credit risk, may cause the fair value of the investment to fluctuate. For example, if we hold a security that was issued with a fixed interest rate at the then-prevailing rate and the prevailing interest rate later rises, the fair value of our investment will probably decline. As of June 30, 2019, our portfolio of financial instruments consists of cash equivalents, including bank deposits, and investments. Due to the short-term basis as well as the nature of our investments, we believe there is no material exposure to interest rate risk, and/or credit risk, arising from our investments. Due to the short-term basis as well as the nature of our investments, we believe there is no material exposure to interest rate risk, and/or credit risk, arising from our investments.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

As of June 30, 2019, management carried out, under the supervision and with the participation of our Chief Executive Officer and Chief Financial Officer, an evaluation of the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act). Our disclosure controls and procedures are designed to provide reasonable assurance that information we are required to disclose in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in applicable rules and forms. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, as of June 30, 2019, our disclosure controls and procedures were effective.

Internal Control Over Financial Reporting

There were no changes in our internal control over financial reporting during the quarter ended June 30, 2019 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We, and our subsidiaries, are not a party to, and our property is not the subject of, any material pending legal proceedings.

ITEM 1A. RISK FACTORS.

You should carefully consider the following risks and uncertainties. If any of the following occurs, our business, financial condition or operating results could be materially harmed. An investment in our securities is speculative in nature, involves a high degree of risk, and should not be made by an investor who cannot bear the economic risk of its investment for an indefinite period of time and who cannot afford the loss of its entire investment. You should carefully consider the following risk factors and the other information contained elsewhere in this Annual Report before making an investment in our securities.

Risks Related to Our Financial Position and Need for Additional Capital

We are a biopharmaceutical company with a limited operating history and have not generated any revenue from drug sales. We have incurred significant operating losses since our inception and anticipate that we will incur continued losses for the foreseeable future.

We are a biopharmaceutical company with a limited operating history on which investors can base an investment decision. Biopharmaceutical drug development is a highly speculative undertaking and involves a substantial degree of risk. We commenced operations in January 2012. Our operations to date have been limited primarily to organizing and staffing our company, business planning, raising capital, developing our technology, identifying potential drug candidates and undertaking pre-clinical studies and commencing clinical trials and conducting Phase 3 and registration directed clinical trials for our most advanced drug candidates, ublituximab and umbralisib. We have never generated any revenue from drug sales. We have not obtained regulatory approvals for any of our drug candidates.

We have not yet demonstrated our ability to successfully complete large-scale, pivotal clinical trials, obtain regulatory approvals, manufacture a commercial scale drug, or arrange for a third party to do so on our behalf, or conduct sales and marketing activities necessary for successful commercialization. Typically, it takes many years to develop one new drug from the time it is discovered to when it is available for treating patients. Consequently, any predictions you make about our future success or viability may not be as accurate as they could be if we had a longer operating history. We will need to transition from a company with a research and development focus to a company capable of supporting commercial activities. We may not be successful in such a transition.

Since inception, we have focused substantially all of our efforts and financial resources on clinical trials and manufacturing of our drug candidates. To date, we have financed our operations primarily through public offerings of our common stock and a debt financing. Through August 6, 2019, we have received an aggregate of approximately \$520 million from such transactions, including approximately \$490 million in aggregate gross proceeds from the sale of common stock in one or more offerings and through the use of our ATM from January 2012 through August 2019 but excluding \$30.0 million available under the term loan facility with Hercules that we secured in February 2019. As of June 30, 2019, there was \$30.0 million available to borrow under that facility, subject to certain conditions of funding.

Since inception, we have incurred significant operating losses. As of June 30, 2019, we had an accumulated deficit of \$599.7 million. Substantially all of our operating losses have resulted from costs incurred in connection with our research and development programs and from general and administrative costs associated with our operations. We expect to continue to incur significant expenses and operating losses for the foreseeable future. Our prior losses, combined with expected future losses, have had and will continue to have an adverse effect on our stockholders' deficit and working capital. We expect our research and development expenses to significantly increase in connection with continuing our existing clinical trials and beginning additional clinical trials. In addition, if we obtain marketing approval for our drug candidates, we will incur significant sales, marketing and outsourced-manufacturing expenses. We have incurred and will continue to incur additional costs associated with operating as a public company. As a result, we expect to continue to incur significant and increasing operating losses for the foreseeable future. Because of the numerous risks and uncertainties associated with developing pharmaceuticals, we are unable to predict the extent of any future losses or when we will become profitable, if at all. Even if we do become profitable, we may not be able to sustain or increase our profitability on a quarterly or annual basis. Our ability to become profitable depends upon our ability to generate revenue.

To date, we have not generated any significant revenue from our drug candidates and we do not expect to generate any revenue from the sale of drugs in the near future. Our ability to become profitable depends upon our ability to generate significant continuing revenues. To obtain significant continuing revenues, we must succeed, either alone or with others, in developing, obtaining regulatory approval for and manufacturing and marketing our product candidates. Accordingly, we do not expect to generate significant and sustained revenue unless and until we obtain marketing approval of, and begin to sell umbralisib, ublituximab and/or one of our other product candidates. Our ability to generate revenue depends on a number of factors, including, but not limited to, our ability to:

- Successfully complete clinical trials that meet their clinical endpoints;
- Initiate and successfully complete all safety, pharmacokinetic, biodistribution, non-clinical studies required to obtain US and Foreign marketing approval for our drug candidates;
- Obtain approval from the US FDA and Foreign equivalent to market and sell our drug candidates;
- Establish commercial manufacturing capabilities alone and/or with third parties that are satisfactory to the regulatory authorities, cost effective, and that are capable of providing commercial supply of our drug candidates;
- Establish a commercial infrastructure to commercialize our drug candidates, if approved, by developing a sales force and/or entering into collaborations with third parties; and
- Achieve market acceptance of our drug candidates in the medical community and with third-party payors.

If we are unable to generate significant continuing revenues, we will not become profitable and we may be unable to continue our operations without continued funding.

We will need to raise substantial additional funding. If we are unable to raise capital when needed, we would be forced to delay, reduce or eliminate some of our drug development programs or commercialization efforts.

The development of pharmaceuticals is capital-intensive. We are currently advancing our most advanced drug candidates, umbralisib, ublituximab, TG-1501, TG-1701 and TG-1801 through clinical development. While we may experience short-term decreases in clinical trial expenses as our larger Phase 3 clinical trials complete and before our Phase 1 and 2 programs can advance into Phase 2 and 3, we do expect over time our overall expenses will increase in connection with our ongoing activities, particularly as we continue the research and development of, initiate or continue clinical trials of, and seek marketing approval for, our drug candidates. In addition, depending on the status of regulatory approval or, if we obtain marketing approval for any of our drug candidates, we expect to incur significant commercialization expenses related to drug sales, marketing, manufacturing and distribution. Moreover, in anticipation of filing for regulatory approvals for umbralisib and ublituximab we will need to expend substantial resources on manufacturing and NDA/ biologics license application (BLA) preparation over the next 12-24+ months, which could exceed any cost savings associated with lower clinical trial expenses during the same period.

While this timing is our current estimate, the amount and timing of our future funding requirements will depend on many factors, including, but not limited to, the following:

- the progress of our clinical trials, including expenses to support the trials and milestone payments that may become payable under our license agreements:
- the costs and timing of regulatory approvals;
- the costs and timing of clinical and commercial manufacturing supply arrangements for each product candidate;
- the costs of establishing sales or distribution capabilities;
- the success of the commercialization of our products;
- our ability to establish and maintain strategic collaborations, including licensing and other arrangements;
- the costs involved in enforcing or defending patent claims or other intellectual property rights; and
- the extent to which we in-license or invest in other indications or product candidates.

Required additional sources of financing to continue our operations in the future might not be available on favorable terms, if at all. If we do not succeed in raising additional funds on acceptable terms, we might be unable to complete planned preclinical and clinical trials or obtain approval of any of our product candidates from the FDA or any foreign regulatory authorities. In addition, we could be forced to discontinue product development, reduce or forego sales, marketing and medical educational efforts that are required for a successful launch of umbralisib and/or ublituximab or any of our product candidates and otherwise forego attractive business opportunities. Any additional sources of financing will likely involve the issuance of our equity securities, which would have a dilutive effect to stockholders. Currently, none of our product candidates have been approved by the FDA or any foreign regulatory authority for sale. Therefore, for the foreseeable future, we will have to fund all of our operations and capital expenditures from cash on hand and amounts raised in future offerings or financings. Accordingly, our prospects must be considered in light of the uncertainties, risks, expenses and difficulties frequently encountered by companies in the early stages of operations and the competitive environment in which we operate.

Raising additional capital may cause dilution to our stockholders, restrict our operations or require us to relinquish rights to our technologies or drug candidates and occupy valuable management time and resources.

Until such time, if ever, as we can generate substantial drug revenues, we expect to finance our cash needs through a combination of public and private equity offerings, debt financings, collaborations, strategic alliances and licensing arrangements. We do not have any committed external source of funds, other than funds already borrowed under the loan and security agreement that we entered into with Hercules in February 2019 (See Note 7 to our condensed consolidated financial statements for more information). To the extent that we raise additional capital through the sale of common stock or securities convertible or exchangeable into common stock, the ownership interest of our stockholders will be diluted, and the terms of these securities may include liquidation or other preferences that materially adversely affect the rights of our common stockholders. Debt financing, if available, would increase our fixed payment obligations and may involve agreements that include restrictive covenants limiting or restricting our ability to take specific actions, such as incurring additional debt, making capital expenditures, declaring dividends, acquiring, selling or licensing intellectual property rights and other operating restrictions that could adversely impact our ability to conduct our business. We could also be required to seek funds through collaborations, strategic alliances or licensing arrangements with third parties at a time that is not desirable to us and we may be required to relinquish valuable rights to some intellectual property, future revenue streams, research programs or drug candidates or to grant licenses on terms that may not be favorable to us, any of which may have a material adverse effect on our business, operating results and prospects. Debt financing, if available, may involve agreements that include covenants limiting or restricting our ability to take specific actions, such as incurring additional debt, making capital expenditures or declaring dividends. For example, see our risk factors und

Additionally, fundraising efforts may divert our management from their day-to-day activities, which may adversely affect our ability to develop and commercialize our drug candidates. Dislocations in the financial markets have generally made equity and debt financing more difficult to obtain and may have a material adverse effect on our ability to meet our fundraising needs. We cannot guarantee that future financing will be available in sufficient amounts or on terms acceptable to us, if at all. Moreover, the terms of any financing may adversely affect the holdings or the rights of our stockholders and the issuance of additional securities, whether equity or debt, by us, or the possibility of such issuance, may cause the market price of our shares to decline. The sale of additional equity or convertible securities would dilute all of our stockholders.

All product candidate development timelines and projections in this report are based on the assumption of further financing.

The timelines and projections in this report are predicated upon the assumption that we will raise additional financing in the future to continue the development of our product candidates. In the event we do not successfully raise subsequent financing, our product development activities will necessarily be curtailed commensurate with the magnitude of the shortfall. If our product development activities are slowed or stopped, we would be unable to meet the timelines and projections outlined in this filing. Failure to progress our product candidates as anticipated will have a negative effect on our business, future prospects, and ability to obtain further financing on acceptable terms, if at all, and the value of the enterprise.

Due to limited resources we may fail to capitalize on programs or product candidates that may present a greater commercial opportunity or for which there is a greater likelihood of success.

Because we have limited resources, we may forego or delay pursuit of opportunities with certain programs or product candidates or for indications that later prove to have greater commercial potential. Our estimates regarding the potential market for a product candidate could be inaccurate, and our spending on current and future research and development programs may not yield any commercially viable products. If we do not accurately evaluate the commercial potential for a particular product candidate, we may relinquish valuable rights to that product candidate through strategic collaboration, licensing or other arrangements in cases in which it would have been more advantageous for us to retain sole development and commercialization rights to such product candidate. Alternatively, we may allocate internal resources to a product candidate in a therapeutic area in which it would have been more advantageous to enter into a partnering arrangement.

If any of these events occur, we may be forced to abandon or delay our development efforts with respect to a particular product candidate or fail to develop a potentially successful product candidate, which could have a material adverse effect on our business, financial condition, results of operations and prospects.

Risks Related to our Indebtedness

Our level of indebtedness and debt service obligations could adversely affect our financial condition, and may make it more difficult for us to fund our operations.

In February 2019, we entered into a Loan and Security Agreement (the "Loan Agreement"), with Hercules Capital, Inc., a Maryland corporation ("Hercules") (See Note 7 for more information). Under the Loan Agreement, Hercules will provide access to term loans with an aggregate principal amount of up to \$60.0 million (the "Term Loan"). Concurrently with the closing of the Loan Agreement, we borrowed an initial tranche of \$30.0 million. In addition, we have incurred short term liabilities of approximately \$20.7 million with a contract manufacturing organization (CMO) for the scale-up, tech-transfer, and long-term supply of one of our drug candidates. This is an expensive and lengthy process and we expect to incur additional obligations associated with these ongoing manufacturing activities over the course of the next 24 months, and potentially longer. To date, this CMO has provided payment terms which we believe are reasonable, however no assurance can be given that such terms will continue to be available to us in the future. No assurances can be made that the obligations associated with the Loan Agreement and the CMO will not have a material adverse impact on our financial condition.

All obligations under the Loan Agreement are secured by substantially all of our existing property and assets, excluding intellectual property. This indebtedness may create additional financing risk for us, particularly if our business or prevailing financial market conditions are not conducive to paying off or refinancing its outstanding debt obligations at maturity. This indebtedness could also have important negative consequences, including:

- We will need to repay the indebtedness by making payments of interest and principal, which will reduce the amount of money available to finance our operations, our research and development efforts and other general corporate activities; and
- our failure to comply with the restrictive covenants in the Loan Agreement could result in an event of default that, if not cured or waived, would
 accelerate our obligation to repay this indebtedness, and Hercules could seek to enforce its security interest in the assets securing such
 indebtedness.

To the extent additional debt is added to our current debt levels, the risks described above could increase.

We may not have cash available in an amount sufficient to enable it to make interest or principal payments on its indebtedness when due.

Failure to satisfy our current and future debt obligations under the Loan Agreement, or breaching any of its covenants under the Loan Agreement, subject to specified cure periods with respect to certain breaches, could result in an event of default and, as a result, Hercules could accelerate all of the amounts due. In the event of an acceleration of amounts due under the Loan Agreement as a result of an event of default, we may not have enough available cash or be able to raise additional funds through equity or debt financings to repay such indebtedness at the time of such acceleration. In that case, we may be required to delay, limit, reduce or terminate our product candidate development or commercialization efforts or grant to others rights to develop and market product candidates that we would otherwise prefer to develop and market our self. Hercules could also exercise its rights as collateral agent to take possession and dispose of the collateral securing the term loans for its benefit, which collateral includes substantially all of our property other than intellectual property. Our business, financial condition and results of operations could be materially adversely affected as a result of any of these events. We are subject to certain restrictive covenants which, if breached, could have a material adverse effect on our business and prospects.

The Loan Agreement imposes operating and other restrictions on the Company. Such restrictions will affect, and in many respects limit or prohibit, our ability and the ability of any future subsidiary to, among other things:

- dispose of certain assets;
- change its lines of business;
- engage in mergers, acquisitions or consolidations;
- incur additional indebtedness;
- create liens on assets;
- pay dividends and make contributions or repurchase our capital stock; and
- engage in certain transactions with affiliates.

Risks Related to Drug Development and Regulatory Approval

If we are unable to obtain regulatory approval for our most advanced drug candidates or other drug candidates and ultimately commercialize our most advanced drug candidates or other drug candidates, or experience significant delays in doing so, our business will be materially harmed.

We are a development-stage biopharmaceutical company, and do not currently have any commercial products that generate revenues or any other sources of revenue. We may never be able to successfully develop marketable products. Our pharmaceutical development methods are unproven and may not lead to commercially viable products for a variety of reasons. We have substantially invested all of our efforts and financial resources in the identification and pre-clinical and clinical development of our drug candidates, including ublituximab, umbralisib, TG-1501, TG-1701 and TG-1801. Our ability to generate drug revenues, which we do not expect will occur for a number of years, if ever, will depend heavily on the successful completion of our current and future Phase 3 and registration-directed clinical trials and eventual commercialization of our drug candidates, which may never occur. We currently generate no revenues from sales of any drugs, and we may never be able to develop or commercialize a marketable drug. Each of our drug candidates will require additional non-clinical or clinical development, management of clinical, non-clinical and manufacturing activities, regulatory approval in multiple jurisdictions, obtaining manufacturing supply, building of a commercial organization, substantial investment and significant marketing efforts before we generate any revenues from drug sales. The success of our most advanced drug candidates and other drug candidates will depend on several factors, including the following:

- Successful completion of our UNITY-NHL trial, UNITY-CLL trial and our ULTIMATE I and II trials;
- Receipt of regulatory approvals from applicable regulatory authorities for our drug candidates;
- Establishing commercial manufacturing capabilities or making arrangements with third-party manufacturers for clinical supply and commercial manufacturing;
- Obtaining and maintaining patent and trade secret protection or regulatory exclusivity for our drug candidates;
- Launching commercial sales of our drug candidates, if and when approved, whether alone or in collaboration with others;
- Acceptance of the drug candidates, if and when approved, by patients, the medical community and third-party payors;
- Effectively differentiating and competing with other therapies;
- Obtaining and maintaining healthcare coverage and adequate reimbursement;
- Enforcing and defending intellectual property rights and claims; and
- Maintaining an acceptable safety profile of the drug candidates following approval.

If we do not achieve one or more of these factors in a timely manner or at all, we could experience significant delays or an inability to successfully commercialize our drug candidates, which would materially harm our business.

If we are unable to develop or receive regulatory approval for or successfully commercialize any of our product candidates, we will not be able to generate product revenues and we may not be able to continue our operations. Even if we are able to develop or receive regulatory approval for or successfully commercialize any of our product candidates, we may not be able to gain market acceptance for our product candidates and future products and may never become profitable.

Because the results of preclinical studies and early clinical trials are not necessarily predictive of future results, any product candidate we advance into clinical trials may not have favorable results in later clinical trials, if any, or receive regulatory approval.

Pharmaceutical development has inherent risks. The outcome of pre-clinical development testing and early clinical trials may not be predictive of the outcome of later clinical trials, and interim results of a clinical trial do not necessarily predict final results. Moreover, pre-clinical and clinical data are often susceptible to varying interpretations and analyses, and many companies that have believed their drug candidates performed satisfactorily in pre-clinical studies and clinical trials have nonetheless failed to obtain marketing approval of their drug candidates. Once a drug candidate has displayed sufficient pre-clinical data to warrant clinical investigation, we will be required to demonstrate through adequate and well-controlled clinical trials that our product candidates are effective with a favorable benefit-risk profile for use in populations for their target indications before we can seek regulatory approvals for their commercial sale. Success in early clinical trials does not mean that later clinical trials will be successful because product candidates in later-stage clinical trials may fail to demonstrate sufficient safety or efficacy despite having progressed through initial clinical testing. Companies frequently experience significant setbacks in advanced clinical trials, even after earlier clinical trials have shown promising results. In addition, there is typically an extremely high rate of failure of pharmaceutical candidates proceeding through clinical trials.

For instance, early clinical results seen with ublituximab (TG-1101) and/or umbralisib (TGR-1202) in a small number of patients may not be reproduced in expanded or larger clinical trials such as in our UNITY-CLL, UNITY-NHL and ULTIMATE I and II programs. Additionally, individually reported outcomes of patients treated in clinical trials may not be representative of the entire population of treated patients in such studies. Further, larger scale Phase 3 studies, which are often conducted internationally, are inherently subject to increased operational risks compared to earlier stage studies. including the risk that the results could vary on a region to region, or country to country basis which could materially adversely affect the study's outcome or the opinion of the validity of the study results by applicable regulatory agencies. In addition, early clinical trial results from interim analysis or from the review of a Data Safety Monitoring Board ("DSMB") or similar safety committee may not be reflective of the results of the entire study, when completed. Clinical trial results can change over time as additional patients are accrued to a study or as additional follow-up is conducted, which may result in a material negative impact on the preliminary results. For instance, we recently announced that the MZL cohort of the UNITY-NHL study met the primary endpoint of overall response rate (ORR), falling within our target range which was between 40%-50%, and while we believe that the ORR results have the potential to increase over time within this range, or beyond, as patients continue to be followed, no assurance can be given that that will be the case, and the results may ultimately fall at the lower end of the range. Further, time to event based endpoints such as duration of response (DOR) and PFS have the potential to change, sometimes drastically, with longer follow-up. No assurance can be provided that the ORR, DOR and PFS data from the MZL cohort of the UNITY-NHL study will be supportive of an FDA approval or will support broad market uptake based on the profile of competitor drugs which may be available. Additionally, while the MZL cohort of the UNITY-NHL study met its primary endpoint, each cohort of this study is operated and analyzed independently, and no assurance can be given that other cohorts from the UNITY-NHL study, including the FL/SLL cohort, MCL cohort and the DLBCL cohort, will meet their primary endpoint, have a positive outcome, or will be supportive of an FDA filing.

All of our Phase 3 and registration directed clinical trials such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II utilize international clinical research sites, including sites in eastern European countries. The Company works with what we believe are reputable Clinical Research Organization's ("CRO") and clinical research sites in conducting our studies internationally. Nevertheless, the risk of fraud, incompetence, unexpected patient variability and other issues affecting the quality and the outcome of our Phase 3 and registration directed studies could arise from US or international sites. If that were to occur, the study could be negatively impacted, potentially even preventing it from being useful for regulatory approval. If such event were to occur, it would have a substantial negative impact on the Company.

Additionally, many of the results reported in our early clinical trials rely on local investigator assessed safety and efficacy outcomes which may differ from results assessed in a blinded, independent, centrally reviewed manner, often required of adequate and well controlled registration directed clinical trials which may be undertaken at a later date. All of our current Phase 3 and registration directed studies such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II trials utilize blinded, independent, central review to assess the primary endpoint of such studies. If the results from interim analyses are not consistent with final results or results from our registration directed trials are different from the results found in the earlier studies of ublituximab and umbralisib, we may need to terminate or revise our clinical development plan, which could extend the time for conducting our development program and could have a material adverse effect on our business. For example, we recently presented to the FDA, interim results from the Marginal Zone Lymphoma ("MZL") cohort of our UNITY-NHL trial that supported the granting of Breakthrough Therapy Designation (BTD). These interim results were also presented at the 2019 American Association of Cancer Research (AACR) annual meeting, 2019 American Society of Clinical Oncology (ASCO) annual meeting, and the 2019 International Conference on Malignant Lymphoma (ICML). No assurance can be given that the final results from that cohort will reflect the activity seen in these interim results, or that the final results will be sufficient to file for accelerated approval for umbralisib for the treatment of MZL, and if filed that umbralisib will receive accelerated approval. Similarly, while early Phase 1 data for umbralisib and ublituximab alone and together looked promising there is no assurance that the UNITY-CLL trial will be positive. Moreover, while we believe one of the key differentiators for umbralisib is its tolerability and side effect profile compared to other drugs in the same class, no assurance can be given that a differentiated safety and tolerability profile will be realized in our Phase 3 or registration directed trials such as UNITY-CLL or UNITY-NHL. Specifically, we have not yet assessed the final safety data from the MZL cohort of the UNITY-NHL study as patients continue on therapy, therefore there can be no assurance given that the final safety data, once fully analyzed, will be consistent with prior safety data presented on umbralisib, will be differentiated from other similar agents in the same class, or that it will be favorable enough to support an FDA filing. In addition, no assurance can be given that new toxicities, or an increase in the severity or frequency of previously seen toxicities, will not be observed, which could have a material negative impact on the approvability or marketability of umbralisib or any of our product candidates. Finally, while the Phase 2 data for ublituximab in MS looked promising, no assurance can be given that the profile will carry into Phase 3 and that the ULTIMATE I and II clinical trials will be positive.

In addition to umbralisib and ublituximab, we have a number of compounds in early clinical development, such as TG-1501, TG-1701 and TG-1801. Many drugs fail in the early stages of clinical development for safety and tolerability issues, despite promising pre-clinical results. Accordingly, no assurance can be made that a safe and efficacious dose can be found for these compounds or that they will ever enter into advanced clinical trials alone or in combination with umbralisib and/or ublituximab.

Clinical drug development involves a lengthy and expensive process, with an uncertain outcome. We may incur additional costs or experience delays in completing, or ultimately be unable to complete, the development and commercialization of our drug candidates.

Our drug candidates umbralisib and ublituximab are in several Phase 3 and registration directed clinical trials such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II. As with all clinical trials, the risk of failure for our drug candidates is high. It is impossible to predict when or if any of our drug candidates will prove effective and safe in humans or will receive regulatory approval. Before obtaining marketing approval from regulatory authorities for the sale of any drug candidate, we must complete pre-clinical studies and then conduct extensive clinical trials to demonstrate the safety and efficacy of our drug candidates in humans. Clinical testing is expensive, difficult to design and implement, can take many years to complete and is uncertain as to outcome. A failure of one or more clinical trials can occur at any stage of testing. Accordingly, our current Phase 3 and registration directed trials, such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II and future clinical trials may not be successful.

Successful completion of our clinical trials is a prerequisite to submitting an NDA, or a BLA to the U.S. FDA and a Marketing Authorization Application ("MAA"), in the European Union for each drug candidate and, consequently, the ultimate approval and commercial marketing of our drug candidates. We do not know whether any of our clinical trials for our drug candidates will be completed on schedule, if at all.

Whether or not and how quickly we complete clinical trials is dependent in part upon the rate at which we are able to engage clinical research/trial sites and, thereafter, the rate of enrollment of patients, and the rate we collect, clean, lock and analyze the clinical trial database. Patient enrollment is a function of many factors, including the size of the patient population, the proximity of patients to clinical sites, the eligibility criteria for the study, the existence of competitive clinical trials, and whether existing or new drugs are approved for the indication we are studying. We are aware that other companies are currently conducting or planning clinical trials that seek to enroll patients with the same diseases that we are studying. We may experience numerous unforeseen events during, or as a result of, any current or future clinical trials that we could conduct that could delay or prevent our ability to receive marketing approval or commercialize our drug candidates, including:

- the FDA or other regulatory authorities may require us to submit additional data or impose other requirements before permitting us to initiate a clinical trial:
- health authorities or institutional review boards ("IRBs"), or ethics committees may not authorize us or our investigators to commence a clinical trial or conduct a clinical trial at a prospective trial site or country;
- we may experience delays in reaching, or fail to reach, agreement on acceptable terms with prospective trial sites and prospective contract
 research organizations, or CROs, the terms of which can be subject to extensive negotiation and may vary significantly among different CROs
 and trial sites;
- clinical trials of our drug candidates may produce negative or inconclusive results, and we may decide, or health authorities may require us, to conduct additional pre-clinical studies or clinical trials or we may decide to abandon drug development programs;
- the number of patients required for clinical trials of our drug candidates may be larger than we anticipate, and enrollment in these clinical trials may be slower than we anticipate or participants may drop out of these clinical trials or fail to return for post-treatment follow-up at a higher rate than we anticipate;
- our third-party contractors, including our clinical trial sites, may fail to comply with regulatory requirements or meet their contractual obligations to us in a timely manner, or at all, or may deviate from the clinical trial protocol or drop out of the trial, which may require that we add new clinical trial sites or investigators;
- we may elect to, or health authorities or IRBs or ethics committees may require that we or our investigators suspend or terminate clinical research for various reasons, including noncompliance with regulatory requirements or a finding that the participants are being exposed to unacceptable health risks;
- the cost of clinical trials of our drug candidates may be greater than we anticipate;
- the supply or quality of our drug candidates or other materials necessary to conduct clinical trials of our drug candidates may be insufficient or inadequate; and
- our drug candidates may have undesirable side effects or other unexpected characteristics, causing us or our investigators, health authorities, IRBs or ethics committees to suspend or terminate the trials, or reports may arise from pre-clinical or clinical testing of other cancer therapies that raise safety or efficacy concerns about our drug candidates.

We could encounter delays if a clinical trial is suspended or terminated by us, by the IRBs of the institutions in which such trials are being conducted, by the DSMB for such trial or by the FDA or other regulatory authorities. Such health authorities may impose a suspension or termination due to a number of factors, including failure to conduct the clinical trial in accordance with regulatory requirements or our clinical protocols, inspection of the clinical trial operations or trial site by the FDA or other regulatory authorities resulting in the imposition of a clinical hold, unforeseen safety issues or adverse side effects, failure to demonstrate a benefit from using a drug, changes in governmental regulations or administrative actions or lack of adequate funding to continue the clinical trial. The DSMBs for the UNITY-CLL and UNITY-NHL studies meet regularly to review ongoing safety from these studies, in an unblinded manner if applicable, and may recommend modification to the study design or closure of the study entirely based on their interpretation of the benefit/risk of the study. While we develop charters that guide the nature of the DSMB's meetings, their analysis and interpretation of study data occurs independently from us and is wholly within their control. Even if the DSMB finds no safety concerns and recommends no modifications to our ongoing studies, this does not mean the safety profile of these studies may support a marketing approval. Many of the factors that cause, or lead to, a delay in the commencement or completion of clinical trials may also ultimately lead to the denial of regulatory approval of our drug candidates. Further, the FDA may disagree with our clinical trial design and our interpretation of data from clinical trials, or may change the requirements for approval even after it has reviewed and commented on the design for our clinical trials. This could happen even for a protocol that has received an SPA. In September 2015, we announced a Phase 3 clinical trial for the combination of ublituximab plus umbralisib for patients with CLL, which is being conducted pursuant to a SPA with the FDA and in August 2017 we announced an SPA for our registration program for ublituximab in RMS. Many companies which have been granted SPAs have ultimately failed to obtain final approval to market their drugs. Since we are seeking approvals under SPAs for some of our product registration strategies, based on protocol designs negotiated with the FDA, we may be subject to enhanced scrutiny. Further, while changes or amendments to protocols are common during conduct of a clinical trial, protocol changes or amendments to a study that is being conducted under SPA will have to be reviewed and approved by the FDA to verify that the SPA agreement is still valid. The FDA's willingness to agree to changes or amendments to a protocol or statistical analysis plan under SPA agreement is wholly within their discretion. Such re-reviews also provide an opportunity for the FDA to scrutinize any aspect of the study design and conduct, even if previously agreed to under the existing SPA. Failure to reach agreement with the FDA for routine protocol changes or modifications for any study we conduct under SPA could have a material negative impact to our ability to execute these studies. Even if the primary endpoint in a Phase 3 clinical trial is achieved, a SPA does not guarantee approval. While UNITY-CLL is being operated in a blinded manner, it is an open label study; meaning that trial participants, investigators, site staff, some employees of our contract research organizations, and field level employees of TGTX (i.e. clinical research associates, monitors, etc.), among others, have knowledge of treatment arm assignments on a patient-level, which has the potential to introduce bias into study conduct. Further, while the ULTIMATE 1 and 2 clinical trials are double-blind, double-dummy studies, unblinding of treatment arm assignment may occur from time to time, for example, on the occurrence of unexpected safety events which may necessitate understanding of study treatment. While we believe we have put in place adequate firewalls to prevent inappropriate unblinding of study data consistent with standard industry practice for these types of studies, no assurance can be given that issues related to study conduct will not be raised. The FDA may raise issues of safety, study conduct, bias, deviation from the protocol, statistical power, patient completion rates, changes in scientific or medical parameters or internal inconsistencies in the study design or data prior to making its final decision. The FDA may also seek the guidance of an outside advisory committee prior to making its final decision.

Negative or inconclusive results from the clinical trials we conduct or unanticipated adverse medical events could cause us to have to repeat or terminate the clinical trials. If we are required to repeat or conduct additional clinical trials or other testing of our drug candidates beyond those that we currently contemplate, if we are unable to successfully complete clinical trials of our drug candidates or other testing, if the results of these trials or tests are not positive or are only modestly positive or if there are safety concerns, we may:

- be delayed in obtaining marketing approval for our drug candidates;
- not obtain marketing approval at all;
- obtain approval for indications or patient populations that are not as broad as intended or desired;
- be subject to post-marketing testing requirements; or
- have the drug removed from the market after obtaining marketing approval.

Our drug development costs will also increase if we experience delays in testing or regulatory approvals. We may also incur additional costs if enrollment is increased. All our current Phase 3 and registration-directed clinical trials, such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II enrolled a larger number of patients than our initial projections, adding significant costs to those studies over and above what had been projected. We do not know whether any of our clinical trials will begin as planned, will need to be restructured or will be completed on schedule, or at all. Certain clinical trials are designed to continue until a pre-determined number of events have occurred in the patients enrolled. Trials such as this are subject to delays stemming from patient withdrawal and from lower than expected event rates. UNITY-CLL is an event-driven study, which means the study can only end when a certain prespecified number of events have occurred. In the case of UNITY-CLL, an event is defined as disease progression or death. Given that these events cannot be predicted with certainty, predicting accurately when this study will reach a sufficient number of events to be complete is impossible. We have stated we believe the number of events can be reached by YE19 or in 2020 but there can be no assurance that that will occur and timelines for the completion of this study should not be relied on given the inherent uncertainty. Delays beyond early 2020 could have a material and adverse impact on the Company. Significant clinical trial delays also could shorten any periods during which we may have the exclusive right to commercialize our drug candidates or allow our competitors to bring products to market before we do and impair our ability to successfully commercialize our drug candidates. Any delays in our pre-clinical or future clinical development programs may harm our business, financial condition and prospects significantly.

In addition, principal investigators for our clinical trials may serve as scientific advisors or consultants to us from time to time and receive cash or equity compensation in connection with such services. If these relationships and any related compensation result in perceived or actual conflicts of interest, the integrity of the data generated at the applicable clinical trial site, or FDA's willingness to accept such data, may be jeopardized.

The sufficiency of our clinical trial results for accelerated approval are subject to the FDA's discretion.

We have and will continue to explore strategies for ublituximab and/or umbralisib that involve use of the FDA's accelerated approval pathway. Obtaining accelerated approval for an agent requires demonstration of meaningful benefit over all available therapies for a serious condition. While we believe we have an understanding of what is considered available therapy today, ultimately the determination of what constitutes available therapy is wholly up to the FDA and is subject to change. No assurance can be given that other agents will not receive full approval prior to our potential receipt of accelerated approval. If that were to occur, no assurance can be given that we would be successful in proving meaningful benefit over those later approved drugs. If we were unable to prove meaningful benefit over any such agents, we would be effectively blocked from receiving accelerated approval. We are currently awaiting final results from our UNITY-NHL trial, in particular the MZL cohort, which we are hoping will be useful for accelerated approval if positive. Even if the results are positive, no assurance can be given that umbralisib will obtain accelerated approval for a variety of reasons, including if a new treatment receives full approval prior to our potential receipt of accelerated approval. Previously, we were hopeful to utilize the results from our GENUINE study for accelerated approval but the intervening full approval of a drug called venetoclax for relapsed/refractory CLL has made that potential application more challenging. While no final decision has been made as to the filing of the GENUINE study for accelerated approval, the Company has no plans to pursue that filing at this time. No assurance can be given that a filing based on the GENUINE results will ever be made.

Finally, if any of our drugs were ever to receive accelerated approval, we would be required to conduct a post-market confirmatory study, which we may not complete, or if completed, may prove unsuccessful. In such instance, the FDA can remove the product from the market.

Our drug candidates may cause undesirable side effects that could delay or prevent their regulatory approval, limit the commercial profile of an approved label, or result in significant negative consequences following marketing approval, if any.

Unacceptable or undesirable adverse events caused by any of our product candidates that we take into clinical trials could cause either us, a DSMB, or regulatory authorities to interrupt, delay, modify or halt clinical trials and could result in a more restrictive label or the delay or denial of regulatory approval by the FDA or other regulatory authorities. This, in turn, could prevent us from commercializing the affected product candidate and generating revenues from its sale.

As is the case with all drugs, it is likely that there will be side effects associated with the use of our drug candidates. Results of our trials could reveal a high and unacceptable severity and prevalence of side effects. In such an event, our trials could be suspended or terminated and the FDA or comparable foreign regulatory authorities could order us to cease further development of or deny approval of our drug candidates for any or all targeted indications. The drug-related side effects could also affect patient recruitment or the ability of enrolled patients to complete the trial or result in potential product liability claims. Any of these occurrences may harm our business, financial condition and prospects significantly.

Many compounds that initially showed promise in early stage testing have later been found to cause side effects that prevented further development of the compound. Further, early clinical trials by their nature utilize a small sample of the potential patient population. With a limited number of patients and limited duration of exposure, rare and severe side effects of our drug candidates may only be uncovered with a significantly larger number of patients exposed to the drug candidate in Phase 3 or registration directed trials and on the market.

We are currently running our Phase 3 and registration-directed trials, such as UNITY-CLL, UNITY-NHL and ULTIMATE I and II and have not completed testing of any of our product candidates for the treatment of the indications for which we intend to seek product approval in humans, and we currently do not know the extent that the adverse events, if any, will be observed in patients who receive any of our product candidates. To date, clinical trials using ublituximab and umbralisib have demonstrated a toxicity profile that was deemed acceptable by the investigators performing such studies. Such interpretation may not be shared by future investigators or by the health authorities and in the case of ublituximab and umbralisib, even if deemed acceptable for oncology and/or autoimmune indications, it may not be acceptable for diseases outside the oncology and autoimmune settings, and likewise for any other product candidates we may develop. Additionally, the severity, duration and incidence of adverse events may increase in larger study populations such as found in our on-going Phase 3 and registration-directed trials. Particularly, with respect to umbralisib, although over 1,000 patients to date have been dosed amongst all ongoing umbralisib studies, the full adverse effect profile of umbralisib is not known. It is also unknown as additional patients are exposed for longer durations to umbralisib, whether greater frequency and/or severity of adverse events are likely to occur. Common toxicities of other drugs in the same class as umbralisib include high levels of liver toxicity, infections and colitis, the latter of which notably has presented with later onset, with incidence increasing with duration of exposure. No assurance can be given that an acceptable safety and tolerability profile for umbralisib will continue to be demonstrated in the future with longer durations of exposure, at the fixed 800mg dose being evaluated in our registration-directed trials and in multiple drug combinations. If any of our product candidates cause unacceptable adverse events in clinical trials, we may not be able to obtain marketing approval and generate revenues from its sale, or even if approved for sale may lack differentiation from competitive products, which could have a material adverse impact on our business and operations.

Additionally, in drug-combination clinical development, there is an inherent risk of drug-drug interactions between combination agents which may affect each component's individual pharmacologic properties and the overall efficacy and safety of the combination regimen. Both ublituximab and umbralisib are being evaluated in combination with each other, as well as with a variety of other active anti-cancer agents, which may cause unforeseen toxicity, or impact the severity, duration, and incidence of adverse events observed compared to those seen in the single agent studies of these agents. We also intend to explore multiple combination studies involving TG-1501, TG-1701, and TG-1801. Further, with multi-drug combinations, it is often difficult to interpret or properly assign attribution of an adverse event to any one particular agent, introducing the risk that toxicity caused by a component of a combination regimen could have a material adverse impact on the development of our product candidates. There can be no assurances given that the combination regimens being studied will display tolerability or efficacy suitable to warrant further testing or produce data that is sufficient to obtain marketing approval.

If any of our drug candidates receive marketing approval and we or others later identify undesirable or unacceptable side effects caused by such drug candidates (or any other similar drugs) after such approval, a number of potentially significant negative consequences could result, including:

- regulatory authorities may withdraw or limit their approval of such drug candidates;
- regulatory authorities may require a more significant clinical benefit for approval to offset the risk;
- regulatory authorities may require the addition of labeling statements, including warnings, contra-indications, or precautions, that could diminish the usage of the product or otherwise limit the commercial success of the affected product;
- we may be required to create a medication guide outlining the risks of such side effects for distribution to patients;
- we may be required to change the way such drug candidates are distributed or administered, or to conduct additional clinical trials;
- regulatory authorities may require a Risk Evaluation and Mitigation Strategy ("REMS"), plan to mitigate risks, which could include medication guides, physician communication plans, or elements to assure safe use, such as restricted distribution methods, patient registries and other risk minimization tools;
- we may be subject to regulatory investigations and government enforcement actions;
- we may decide to remove such drug candidates from the marketplace;
- we may not be able to enter into collaboration agreements on acceptable terms and execute on our business model;
- · we could be sued and held liable for injury caused to individuals exposed to or taking our drug candidates; and
- our reputation may suffer.

Any one or a combination of these events could prevent us from obtaining or maintaining regulatory approval and achieving or maintaining market acceptance of the affected product or could substantially increase the costs and expenses of commercializing the affected product, which in turn could significantly impact our ability to successfully commercialize our drug candidates and generate revenues.

Any product candidates we may advance through clinical development are subject to extensive regulation, which can be costly and time consuming, cause unanticipated delays or prevent the receipt of the required approvals.

The clinical development, manufacturing, labeling, storage, record-keeping, advertising, promotion, import, export, marketing and distribution of our product candidates or any future product candidates are subject to extensive regulation by the FDA in the United States and by comparable health authorities worldwide. In the United States, we are not permitted to market a product candidate until we receive approval of a BLA or NDA from the FDA. The process of obtaining BLA and NDA approval is expensive, often takes many years and can vary substantially based upon the type, complexity and novelty of the products involved. Approval policies or regulations may change and the FDA has substantial discretion in the pharmaceutical approval process, including the ability to delay, limit or deny approval of a product candidate for many reasons. In addition, the FDA may require post-approval clinical trials or studies which also may be costly. The FDA approval for a limited indication or approval with required warning language, such as a boxed warning, could significantly impact our ability to successfully market our product candidates. Finally, the FDA may require adoption of a REMS requiring prescriber training, post-market registries, or otherwise restricting the marketing and dissemination of these products. Despite the time and expense invested in the clinical development of product candidates, regulatory approval is never guaranteed. Assuming successful clinical development, we intend to seek product approvals in countries outside the United States. As a result, we would be subject to regulation by the European Medicines Agency ("EMA"), as well as the other regulatory agencies in these countries.

Approval procedures vary among countries and can involve additional product testing and additional administrative review periods. The time required to obtain approval in other countries might differ from that required to obtain FDA approval. Regulatory approval in one country does not ensure regulatory approval in another, but a failure or delay in obtaining regulatory approval in one country may negatively impact the regulatory process in others. As in the United States, the regulatory approval process in Europe and in other countries is a lengthy and challenging process. The FDA, and any other regulatory body around the world can delay, limit or deny approval of a product candidate for many reasons, including:

- the FDA or comparable foreign regulatory authorities may disagree with the design or implementation of our clinical trials;
- we may be unable to demonstrate to the satisfaction of the FDA or comparable foreign regulatory authorities that a product candidate is safe and effective for an indication;
- the FDA may not accept clinical data from trials conducted by individual investigators or in countries where the standard of care is potentially different from the United States;
- the results of clinical trials may not meet the level of statistical significance required by the FDA or comparable foreign regulatory authorities for approval:
- we may be unable to demonstrate that a product candidate's clinical and other benefits outweigh its safety risks;
- the FDA or comparable foreign regulatory authorities may disagree with our interpretation of data from preclinical studies or clinical trials;
- the data collected from clinical trials of our product candidates may not be sufficient to support the submission of a BLA, NDA or other submission or to obtain regulatory approval in the United States or elsewhere;
- the FDA or comparable foreign regulatory authorities may not approve the manufacturing processes or facilities of third-party manufacturers with which we or our collaborators currently contract for clinical supplies and plan to contract for commercial supplies; or
- the approval policies or regulations of the FDA or comparable foreign regulatory authorities may significantly change in a manner rendering our clinical data insufficient for approval.

In addition, raising questions about the safety of marketed pharmaceuticals may result in increased cautiousness by the FDA and other regulatory authorities in reviewing new pharmaceuticals based on safety, efficacy or other regulatory considerations and may result in significant delays in obtaining regulatory approvals. Regulatory approvals for our product candidates may not be obtained without lengthy delays, if at all. Any delay in obtaining, or inability to obtain, applicable regulatory approvals would prevent us from commercializing our product candidates.

A breakthrough therapy designation by the FDA for our drug candidates, including umbralisib for the treatment of adult patients with relapsed or refractory Marginal Zone Lymphoma (MZL) who have received at least one prior treatment including an ant i-CD20 monoclonal antibody, may not lead to a faster development or regulatory review or approval process, and it does not ensure that our drug candidates will receive marketing approval.

In January 2019, the FDA granted breakthrough therapy designation (also referred to as BTD) to umbralisib for the treatment of adult patients with relapsed or refractory MZL who have received at least one prior treatment including an anti-CD20 monoclonal antibody. We may also seek breakthrough therapy designation for some of our other drug candidates. A breakthrough therapy is defined as a drug that is intended, alone or in combination with one or more other drugs, to treat a serious or life-threatening disease or condition, and preliminary clinical evidence indicates that the drug may demonstrate substantial improvement over existing therapies on one or more clinically significant endpoints, such as substantial treatment effects observed early in clinical development. Our breakthrough therapy designation was based on interim data from the MZL cohort of the UNITY-NHL clinical trial. No assurance can be given that the full results from the MZL cohort of the UNITY-NHL clinical trial will be positive and support a filing for accelerated approval.

For drugs that have been designated as breakthrough therapies, interaction and communication between the FDA and the sponsor of the trial can help to identify the most efficient path for clinical development while minimizing the number of patients placed in ineffective control regimens. Drugs designated as breakthrough therapies by the FDA are also eligible for accelerated approval. Designation as a breakthrough therapy is wholly within the discretion of the FDA. Accordingly, even if we believe one of our drug candidates meets the criteria for designation as a breakthrough therapy, the FDA may disagree and instead determine not to grant such designation to the drug candidate. In any event, the receipt of a breakthrough therapy designation for a drug candidate may not result in a faster development process, review or approval compared to drugs considered for approval under conventional FDA procedures and does not assure ultimate approval by the FDA. In addition, even if one or more of our drug candidates qualify as breakthrough therapies, the FDA may later decide that the drugs no longer meet the conditions for qualification and rescind the designation.

A fast track designation by the FDA may not actually lead to a faster development or regulatory review or approval process.

We may seek fast track designation for some of our other drug candidates. If a drug is intended for the treatment of a serious or life-threatening condition and the drug demonstrates the potential to address unmet medical needs for this condition, the drug sponsor may apply for fast track designation. The FDA has broad discretion whether or not to grant this designation, so even if we believe a particular drug candidate is eligible for this designation, we cannot assure you that the FDA would decide to grant it. Even if we receive fast track designation for our other drug candidates, we may not experience a faster development process, review or approval compared to conventional FDA procedures. The FDA may withdraw fast track designation if it believes that the designation is no longer supported by data from our clinical development program.

While we have received orphan drug designation for umbralisib and ublituximab for specified indications, we may seek additional orphan drug designation for those and some of our other drug candidates. However, we may be unsuccessful in obtaining or may be unable to maintain the benefits associated with orphan drug designation, including the potential for market exclusivity.

Ublituximab received orphan-drug designation from the FDA for the treatment of Marginal Zone Lymphoma (Nodal and Extranodal) in September 2013, for the treatment of CLL in August of 2010, and orphan-drug designation by the EMA for the treatment of CLL in November of 2009. We also obtained orphan drug designation for umbralisib as monotherapy for the treatment of CLL in August 2016 and all three types of MZL (nodal, extranodal and splenic) in April 2019, and in January 2017, we announced that the FDA granted Orphan Drug designation covering the combination of ublituximab and umbralisib for the treatment of patients with CLL and DLBCL. As part of our business strategy, we may seek orphan drug designation for our other drug candidates, and we may be unsuccessful. Regulatory authorities in some jurisdictions, including the United States and the European Union, may designate drugs for relatively small patient populations as orphan drugs. Under the Orphan Drug Act, the FDA may designate a drug as an orphan drug if it is a drug intended to treat a rare disease or condition, which is generally defined as a patient population of fewer than 200,000 individuals annually in the United States, or a patient population greater than 200,000 in the United States where there is no reasonable expectation that the cost of developing the drug will be recovered from sales in the United States. In the United States, orphan drug designation entitles a party to financial incentives such as opportunities for grant funding towards clinical trial costs, tax advantages and user-fee waivers.

Even if we obtain orphan drug exclusivity for a drug, that exclusivity may not effectively protect the designated drug from competition because different drugs can be approved for the same condition. Even after an orphan drug is approved, the FDA can subsequently approve the same drug for the same condition if the FDA concludes that the later drug is clinically superior in that it is shown to be safer, more effective or makes a major contribution to patient care. In addition, a designated orphan drug may not receive orphan drug exclusivity if it is approved for a use that is broader than the indication for which it received orphan designation. Moreover, orphan drug exclusive marketing rights in the United States may be lost if the FDA later determines that the request for designation was materially defective or if the manufacturer is unable to assure sufficient quantity of the drug to meet the needs of patients with the rare disease or condition. Orphan drug designation neither shortens the development time or regulatory review time of a drug nor gives the drug any advantage in the regulatory review or approval process. While we intend to seek additional orphan drug designation for our other drug candidates, we may never receive such designations. Even if we receive orphan drug designation for any of our drug candidates, there is no guarantee that we will enjoy the benefits of those designations.

As all of our product candidates are still under development, manufacturing site additions, scale-up and process improvements implemented in the production of those product candidates may affect their ultimate activity or function.

Our product candidates are in the initial stages of development and are currently manufactured in relatively small batches for use in pre-clinical and clinical studies. Process improvements implemented to date have changed, and process improvements in the future may change, the activity/ and analytical profile of the product candidates, which may affect the safety and efficacy of the products. For instance, the manufacturing process for ublituximab has undergone several process improvements during the clinical trial process which have resulted in analytical differences between the materials. Such process improvements continued during the conduct of Phase 3 and material from more than one manufacturing process were utilized in the Phase 3 UNITY-CLL trial. While analytical differences exist between those materials, we do not believe the differences will alter the safety or efficacy profile of ublituximab. However, it is possible that additional and/or different adverse events may appear among patients exposed to drug product manufactured under one process compared to the other, or that adverse events may arise with greater frequency, intensity and duration among patients exposed to drug product manufactured under one process compared to the other. Additionally, the efficacy of ublituximab also can be negatively impacted by such process changes. Given the uncertainty of the impact on product specifications, quality and performance, process improvements made during Phase 3 development carry a higher level of risk then those made prior to Phase 3 development. If there are significant differences in product attributes between the two materials, we may need to adjust our statistical analysis plans of the Phase 3 study to confirm that there is no difference in safety or efficacy between product made by each process in order to and allow us to utilize data from all enrolled patients, as well as be able to integrate clinical safety and/or efficacy results across studies to support any potential marketing application. There can be no assurance given that such analyses will be successful in demonstrating no clinical differences between these drug products, which could substantially impact the approvability of the U2 combination based on the results of the UNITY-CLL study. In such circumstances, that would have a material adverse effect on the Company.

Further, no assurance can be given that the material manufactured from any future optimized processes, if any, for ublituximab or any of our product candidates will perform comparably to the product candidates as manufactured to date which could result in an unexpected safety or efficacy outcome as compared to the data published or presented to date. Similarly, following each round of process improvements, if any, for any of our drug candidates, future clinical trial results conducted with the new material will be subject to uncertainty related to the effects, if any, of those additional process improvements that were made

In addition, we have engaged a secondary manufacturer for ublituximab to meet our current clinical and future commercial needs and anticipate engaging additional manufacturing sources for umbralisib to meet expanded clinical trial and commercial needs. If a secondary manufacturer is not successful in replicating the product or experiences delays, or if regulatory authorities impose unforeseen requirements with respect to product comparability from multiple manufacturing sources, we may experience delays in clinical development. No assurance can be given that any additional manufacturers will be successful or that material manufactured by the additional manufacturers will perform comparably to ublituximab or umbralisib as manufactured to date and used in currently available pre-clinical data and or in early clinical trials presented publicly or reported in this or any previous filing, or that the relevant regulatory agencies will agree with our interpretation of comparability.

In addition, as we move closer to commercialization for ublituximab and umbralisib we will need to scale-up production to ensure adequate commercial supply. We are currently in the process of scaling up ublituximab. This is an expensive process and there can be no assurance given that such scale-up will be successful in providing pharmaceutical product that is of sufficient quantity, or of a quality that is consistent with our previously established specifications, or that meets the requirements set by regulatory agencies under which we may seek approval of our product candidates. If scale-up were not to succeed our ability to supply our anticipated market at a reasonable cost of goods would be negatively impacted. In such event, that would have a material adverse effect on the Company. Scale up could also require additional process improvement that might be required to accommodate new and larger equipment utilized in the scaled-up process. If that were to occur and we could not demonstrate to the FDA that the materials were analytically substantially similar, we might be required to run additional clinical testing to demonstrate that they are substantially similar. That would entail a significant delay and significant increase in total cost, all of which would have a material adverse effect on the Company.

Risks Related to Commercialization

The incidence and prevalence for target patient populations of our drug candidates have not been established with precision. If the market opportunities for our drug candidates are smaller than we estimate or if any approval that we obtain is based on a narrower definition of the patient population, our revenue and ability to achieve profitability will be adversely affected, possibly materially.

The precise incidence and/or prevalence of CLL, relapsed/refractory MZL, relapsed/refractory FL and MS are unknown. Our projections of both the number of people who are affected by disease within our target indications, as well as the subset of these people who have the potential to benefit from treatment with our product candidates, are based on our beliefs and estimates. Our beliefs are typically based on one on one and group interactions with target physicians and our estimates have been derived from a variety of sources, including the scientific literature, healthcare utilization databases and market research, and may prove to be incorrect. Further, new studies may change the estimated incidence or prevalence of these diseases.

The total addressable market opportunity for umbralisib and ublituximab for the treatment of patients with CLL, MZL, FL and MS will ultimately depend upon, among other things, the final label indication, approved for sale for these indications, acceptance by the medical community, patient access, drug pricing and reimbursement. The number of patients in major markets, including the number of addressable patients in those markets, may turn out to be lower than expected, patients may not be otherwise amenable to treatment with our drugs, new patients may become increasingly difficult to identify or gain access to, or patients and physicians may choose to utilize competitive products, all of which would adversely affect our results of operations and our business

We face substantial competition for treatments for our target indications, which may result in others commercializing drugs before or more successfully than we do resulting in the reduction or elimination of our commercial opportunity.

We operate in a highly competitive segment of the biotechnology and biopharmaceutical market. We face competition from numerous sources, including commercial pharmaceutical and biotechnology enterprises, academic institutions, government agencies, and private and public research institutions. Many of our competitors have significantly greater financial, product development, manufacturing and marketing resources. Large pharmaceutical companies have extensive experience in clinical testing and obtaining regulatory approval for drugs. Additionally, many universities and private and public research institutes are active in cancer research, some in direct competition with us. We may also compete with these organizations to recruit scientists and clinical development personnel. Smaller or early-stage companies may also prove to be significant competitors, particularly through collaborative arrangements with large and established companies.

Our commercial opportunity could be reduced or eliminated if our competitors develop and commercialize drugs that are safer, more effective, have fewer or less severe side effects, are more convenient or are less expensive than any drugs that we or our collaborators may develop. Our competitors also may obtain FDA or other regulatory approval for their drugs more rapidly than we may obtain approval for ours, which could result in our competitors establishing a strong market position before we or our collaborators are able to enter the market. The key competitive factors affecting the success of all of our drug candidates, if approved, are likely to be their efficacy, safety, convenience, price, the effectiveness of any related companion diagnostic tests, the level of generic competition and the availability of reimbursement from government and other third-party payors.

For the cancer indications for which we are developing our products there are a number of established therapies with which we will compete:

- For the treatment of CLL, if U2 is approved, we expect U2 to compete with recently approved drugs such as ibrutinib (AbbVie and Janssen), venetoclax (AbbVie and Roche), obinutuzumab (Roche), idelalisib (Gilead) and duvelisib (Verastem), and established treatments such as rituximab (Roche), and several generically available chemotherapies. Additionally, there are two second generation BTK inhibitors similar to ibrutinib in late-stage clinical testing for CLL that could enter the market in the next 12-36 months. Each of these agents can be used as monotherapy or in combination with one or more of the other agents.
- For the treatment of Marginal Zone Lymphoma, if approved, we expect umbralisib to compete with ibrutinib (AbbVie and Janssen) and established treatments such as rituximab and several generically available chemotherapies. Additionally, the combination of rituximab and lenalidomide (Celgene) has recently been approved by the FDA for the treatment of relapsed MZL.
- For the treatment of Follicular Lymphoma, if approved, we expect umbralisib to compete with recently approved drugs such as obinutuzumab (Roche), idelalisib (Gilead), copanlisib (Bayer), and duvelisib (Verastem), and established treatments such as rituximab (Roche), and several generically available chemotherapies. Each of these agents can be used as monotherapy or in combination with one or more of the other agents. The combination of rituximab and lenalidomide (Celgene) has also been studied in FL and has recently been approved by the FDA for the treatment of relapsed FL. There are also several PI3K delta inhibitors in earlier stages of development.
- In addition, a number of pharmaceutical companies are developing antibodies and bispecific antibodies targeting CD20, CD19, CD47 and other B-cell associated targets, chimeric antigen receptor T-cell ("CAR-T") immunotherapy, and other B-cell ablative therapy which, if approved, would potentially compete with U2 and umbralisib.

For Multiple Sclerosis for which we are developing ublituximab there are a number of established therapies with which we will compete:

• If ublituximab is approved, we expect ublituximab will primarily compete against other CD20 targeted agents, while the group of CD20 targeted agents will also compete broadly against a number of already approved MS therapies. Currently, there is one anti-CD20 monoclonal antibody approved, ocrelizumab (Roche), and another in Phase 3 development, of atumumab (Novartis), which is expected to enter the market in the next 12-24 months.

TG-1501, TG-1701 and TG-1801 if approved will also face competition from drugs on the market and under development that have the same mechanism of action as each of those drugs.

New developments, including the development of other pharmaceutical technologies and methods of treating disease, occur in the pharmaceutical and life sciences industries at a rapid pace. These developments may render our product candidates obsolete or noncompetitive. Compared to us, many of our potential competitors have substantially greater:

- research and development resources, including personnel and technology;
- regulatory experience;
- pharmaceutical development, clinical trial and pharmaceutical commercialization experience;
- experience and expertise in exploitation of intellectual property rights; and
- capital resources.

We will also face competition from these third parties in recruiting and retaining qualified personnel, establishing clinical trial sites, patient registration for clinical trials, and in identifying and in-licensing new product candidates.

Product liability lawsuits against us could cause us to incur substantial liabilities and could limit commercialization of any drug candidates that we may develop.

We will face an inherent risk of product liability exposure related to the testing of our drug candidates in human clinical trials and use of our drug candidates through compassionate use programs, and we will face an even greater risk if we commercially sell any drug candidates that we may develop. If we cannot successfully defend ourselves against claims that our drug candidates caused injuries, we could incur substantial liabilities. Regardless of merit or eventual outcome, liability claims may result in:

- decreased demand for any drug candidates that we may develop;
- injury to our reputation and significant negative media attention;
- withdrawal of clinical trial participants;
- significant costs to defend the related litigation;
- substantial monetary awards to trial participants or patients;
- loss of revenue; and
- the inability to commercialize any drug candidates that we may develop.

Although we maintain product liability insurance coverage, it may not be adequate to cover all liabilities that we may incur. We anticipate that we will need to increase our insurance coverage if we successfully commercialize any drug candidate. Insurance coverage is increasingly expensive. We may not be able to maintain insurance coverage at a reasonable cost or in an amount adequate to satisfy any liability that may arise.

If any product candidate that we successfully develop does not achieve broad market acceptance among physicians, patients, healthcare payors, and the medical community, the revenues that we generate from its sales will be limited.

Even if our product candidates receive regulatory approval, they may not gain market acceptance among physicians, patients, healthcare payors, and the medical community. Coverage and reimbursement of our product candidates by third-party payors, including government payors, generally would also be necessary for commercial success. The degree of market acceptance of any approved product would depend on a number of factors, including:

- the efficacy and safety as demonstrated in clinical trials;
- the timing of market introduction of such product candidate as well as competitive products;
- the clinical indications for which the product is approved;
- acceptance by physicians, major operators of cancer clinics and patients of the product as a safe and effective treatment;
- the potential and perceived advantages of the product candidate over alternative treatments;
- the safety of the product candidate in a broader patient group;
- the cost of treatment in relation to alternative treatments;
- the availability of adequate reimbursement and pricing by third parties and government authorities;
- changes in regulatory requirements by government authorities for the product candidate;
- relative convenience and ease of administration;
- the prevalence and severity of side effects and adverse events;
- the effectiveness of our sales and marketing efforts; and
- unfavorable publicity relating to the product.

If any product candidate is approved but does not achieve an adequate level of acceptance by physicians, hospitals, healthcare payors and patients, we may not generate sufficient revenue from these products and we may not become or remain profitable.

Even if we are able to commercialize any drug candidates, such drugs may become subject to unfavorable pricing regulations or third-party coverage and reimbursement policies, which would harm our business.

The regulations that govern regulatory approvals, pricing and reimbursement for new drugs vary widely from country to country. Some countries require approval of the sale price of a drug before it can be marketed. In many countries, the pricing review period begins after marketing approval is granted. In some foreign markets, prescription pharmaceutical pricing remains subject to continuing governmental control even after initial approval is granted. As a result, we might obtain marketing approval for a drug candidate in a particular country, but then be subject to price regulations that delay our commercial launch of the drug candidate, possibly for lengthy time periods, and negatively impact the revenues we are able to generate from the sale of the drug candidate in that country. Adverse pricing limitations may hinder our ability to recoup our investment in one or more drug candidates, even if our drug candidates obtain marketing approval.

Our ability to commercialize any drug candidates successfully also will depend in part on the extent to which coverage and reimbursement for these drug candidates and related treatments will be available from government authorities, private health insurers and other organizations. Government authorities and third-party payors, such as private health insurers and health maintenance organizations, decide which medications they will pay for and establish reimbursement levels. A primary trend in the U.S. healthcare industry and elsewhere is cost containment. Government authorities and third-party payors have attempted to control costs by limiting coverage and the amount of reimbursement for particular drugs. Increasingly, third-party payors are requiring that drug companies provide them with predetermined discounts from list prices and are challenging the prices charged for drugs. We cannot be sure that coverage will be available for any drug candidate that we commercialize and, if coverage is available, the level of reimbursement. Reimbursement may impact the demand for, or the price of, any drug candidate for which we obtain marketing approval. If reimbursement is not available or is available only to limited levels, we may not be able to successfully commercialize any drug candidate for which we obtain marketing approval.

There may be significant delays in obtaining reimbursement for newly approved drugs, and coverage may be more limited than the purposes for which the drug is approved by the FDA or similar regulatory authorities outside the United States. Moreover, eligibility for reimbursement does not imply that any drug will be paid for in all cases or at a rate that covers our costs, including research, development, manufacture, sale and distribution. Interim reimbursement levels for new drugs, if applicable, may also not be sufficient to cover our costs and may not be made permanent. Reimbursement rates may vary according to the use of the drug and the clinical setting in which it is used, may be based on reimbursement levels already set for lower-cost drugs and may be incorporated into existing payments for other services. Net prices for drugs may be reduced by mandatory discounts or rebates required by government healthcare programs or private payors and by any future relaxation of laws that presently restrict imports of drugs from countries where they may be sold at lower prices than in the United States. Third-party payors often rely upon Medicare coverage policy and payment limitations in setting their own reimbursement policies. Our inability to promptly obtain coverage and profitable payment rates from both government-funded and private payors for any approved drugs that we develop could have a material adverse effect on our operating results, our ability to raise capital needed to commercialize drugs and our overall financial condition.

We are subject to new legislation, regulatory proposals and managed care initiatives that may increase our costs of compliance and adversely affect our ability to market our products, obtain collaborators and raise capital.

In both the United States and certain foreign countries, there have been a number of legislative and regulatory changes to the healthcare system that could impact our ability to sell our products profitably. In particular, the Medicare Modernization Act of 2003 revised the payment methodology for many products reimbursed by Medicare, resulting in lower rates of reimbursement for many types of drugs, and added a prescription drug benefit to the Medicare program that involves commercial plans negotiating drug prices for their members. Since 2003, there have been a number of other legislative and regulatory changes to the coverage and reimbursement landscape for pharmaceuticals.

The Patient Protection and Affordable Care Act, as amended by the Health Care and Education Reconciliation Act of 2010, collectively, the "ACA," was enacted in 2010 and made significant changes to the United States' healthcare system. The ACA and any revisions or replacements of that Act, any substitute legislation, and other changes in the law or regulatory framework could have a material adverse effect on our business.

Among the provisions of the ACA, those of importance to our potential product candidates are:

- an annual, nondeductible fee on any entity that manufactures or imports specified branded prescription drugs and biologic agents, apportioned
 among these entities according to their market share in certain government healthcare programs;
- an increase in the statutory minimum rebates a manufacturer must pay under the Medicaid Drug Rebate Program to 23.1% and 13.0% of the average manufacturer price for branded and generic drugs, respectively;
- expansion of healthcare fraud and abuse laws, including the federal False Claims Act and the federal Anti-Kickback Statute, new government investigative powers and enhanced penalties for non-compliance;
- a new Medicare Part D coverage gap discount program, in which manufacturers must agree to offer 50% point-of-sale discounts off negotiated prices of applicable brand drugs to eligible beneficiaries during their coverage gap period, as a condition for a manufacturer's outpatient drugs to be covered under Medicare Part D;
- extension of a manufacturer's Medicaid rebate liability to covered drugs dispensed to individuals who are enrolled in Medicaid managed care organizations;
- expansion of eligibility criteria for Medicaid programs by, among other things, allowing states to offer Medicaid coverage to additional individuals and by adding new mandatory eligibility categories for certain individuals with income at or below 138% of the federal poverty level, thereby potentially increasing a manufacturer's Medicaid rebate liability;
- expansion of the entities eligible for discounts under the 340B Drug Pricing Program;
- the new requirements under the federal Open Payments program and its implementing regulations;
- a new requirement to annually report drug samples that manufacturers and distributors provide to physicians;
- a new regulatory pathway for the approval of biosimilar biological products, all of which will impact existing government healthcare programs and will result in the development of new programs; and
- a new Patient-Centered Outcomes Research Institute to oversee, identify priorities in, and conduct comparative clinical effectiveness research, along with funding for such research.

President Trump ran for office on a platform that supported the repeal of the ACA, and one of his first actions after his inauguration was to sign an Executive Order instructing federal agencies to waive or delay requirements of the ACA that impose economic or regulatory burdens on states, families, the health-care industry and others. Modifications to or repeal of all or certain provisions of the ACA have been attempted in Congress as a result of the outcome of the recent presidential and congressional elections, consistent with statements made by the incoming administration and members of Congress during the presidential and congressional campaigns and following the election.

In January 2017, Congress voted to adopt a budget resolution for fiscal year 2017, or the Budget Resolution, that authorizes the implementation of legislation that would repeal portions of the ACA. The Budget Resolution is not a law. However, it is widely viewed as the first step toward the passage of legislation that would repeal certain aspects of the ACA. In March 2017, following the passage of the budget resolution for fiscal year 2017, the United States House of Representatives passed legislation known as the American Health Care Act of 2017, which, if enacted, would amend or repeal significant portions of the ACA. Attempts in the Senate in 2017 to pass ACA repeal legislation, including the Better Care Reconciliation Act of 2017, so far have been unsuccessful. At the end of 2017, Congress passed the Tax Cuts and Jobs Act, which repealed the penalty for individuals who fail to maintain minimum essential health coverage as required by the ACA. Following this legislation, Texas and 19 other states filed a lawsuit alleging that the ACA is unconstitutional as the individual mandate was repealed, undermining the legal basis for the Supreme Court's prior decision. On December 14, 2018, Texas federal district court judge Reed O'Connor issued a ruling declaring that the ACA in its entirety is unconstitutional. While this decision has no immediate legal effect on the ACA and its provisions, this lawsuit is ongoing and the outcome through the appeals process may have a significant impact on our business.

The Bipartisan Budget Act of 2018, the "BBA," which set government spending levels for Fiscal Years 2018 and 2019, revised certain provisions of the ACA. Specifically, beginning in 2019, the BBA increased manufacturer point-of-sale discounts off negotiated prices of applicable brand drugs in the Medicare Part D coverage gap from 50% to 70%, ultimately increasing the liability for brand drug manufacturers. Further, this mandatory manufacturer discount applied to biosimilars beginning in 2019.

The 116th Congress has explored legislation intended to address the cost of prescription drugs. Notably, the major committees of jurisdiction in the Senate (Finance Committee, Health, Education, Labor and Pensions Committee, and Judiciary Committee), have marked up legislation intended to address various elements of the prescription drug supply chain. Proposals include a significant overhaul of the Medicare Part D benefit design, addressing patent "loopholes", and efforts to cap the increase in drug prices. The House Energy and Commerce Committee approved drug-related legislation intended to increase transparency of drug prices and also curb anti-competitive behavior in the pharmaceutical supply chain. In addition, the House Ways & Means Committee approved legislation intended to improve drug price transparency, including for drug manufacturers to justify certain price increases. While we cannot predict what proposals may ultimately become law, the elements under consideration could significantly change the landscape in which the pharmaceutical market operates.

The Trump Administration has also taken several regulatory steps to redirect ACA implementation. The Department of Health and Human Services ("HHS") finalized Medicare fee-for-service hospital payment reductions for Part B drugs acquired through the 340B Drug Pricing Program, which has been overturned by the courts. HHS also has signaled its intent to continue to pursue reimbursement policy changes for Medicare Part B drugs as a whole that likely would reduce hospital and physician reimbursement for these drugs.

HHS has made numerous other proposals aimed at lowering drug prices for Medicare beneficiaries and increasing price transparency. These proposals include giving Medicare Advantage and Part D plans flexibility in the availability of drugs in "protected classes," more transparency in the cost of drugs, including the beneficiary's financial liability, and less costly alternatives and permitting the use of step therapy as a means of prior authorization. HHS has also proposed requiring pharmaceutical manufacturers disclose the prices of certain drugs in direct-to-consumer television advertisements. The proposal related to protected classes has been withdrawn and the disclosure requirements have been rejected by the courts. In addition, a proposed rule that would have passed drug rebates to consumers at the point of sale also has been withdrawn. However, it appears the Trump Administration will continue to explore its authority to make regulatory changes to the pharmaceutical industry. For example, the Trump Administration released an Advance Notice of Proposed Rulemaking related to an international price index model. It is unclear what eventually will be proposed, but the President has alluded to the concept of "most favored nation" pricing with regards to U.S. drug purchasing. In addition, HHS, in conjunction with the FDA, announced that it will be exploring a reimportation pathway in certain instances and for certain drugs.

HHS also has taken steps to increase the availability of cheaper health insurance options, typically with fewer benefits and less generous. The Administration has also signaled its intention to address drug prices and to increase competition, including by increasing the availability of biosimilars and generic drugs. As these are regulatory actions, a new administration could undo or modify these efforts.

We expect that the ACA, as well as other healthcare reform measures that may be adopted in the future, may result in more rigorous coverage criteria and in additional downward pressure on the price that we receive for any approved drug. Any reduction in reimbursement from Medicare or other government healthcare programs may result in a similar reduction in payments from private payors. The implementation of cost containment measures or other healthcare reforms may prevent us from being able to generate revenue, attain profitability or commercialize our drugs.

Legislative and regulatory proposals have been made to expand post-approval requirements and restrict sales and promotional activities for drugs. We cannot be sure whether additional legislative changes will be enacted, or whether the FDA regulations, guidance or interpretations will be changed, or what the impact of such changes on the marketing approvals of our product candidates, if any, may be. In addition, increased scrutiny by the US Congress of the FDA's approval process may significantly delay or prevent marketing approval, as well as subject us to more stringent product labeling and post-marketing testing and other requirements.

There likely will continue to be, legislative and regulatory proposals at the federal and state levels directed at broadening the availability of healthcare and containing or lowering the cost of healthcare products and services. We cannot predict the initiatives that may be adopted in the future. The continuing efforts of the government, insurance companies, managed care organizations and other payors of healthcare services to contain or reduce costs of healthcare may adversely affect:

- our ability to generate revenues and achieve or maintain profitability;
- the demand for any products for which we may obtain regulatory approval;
- our ability to set a price that we believe is fair for our products;
- the level of taxes that we are required to pay; and
- the availability of capital.

In addition, governments may impose price controls, which may adversely affect our future profitability.

We cannot predict the likelihood, nature or extent of government regulation that may arise from future legislation or administrative or executive action, either in the United States or abroad.

We cannot predict the likelihood, nature or extent of how government regulation that may arise from future legislation or administrative or executive action taken by the U.S. presidential administration may impact our business and industry. In particular, the U.S. President has taken several executive actions, including the issuance of a number of Executive Orders, that could impose significant burdens on, or otherwise materially delay, the FDA's ability to engage in routine regulatory and oversight activities such as implementing statutes through rulemaking, issuance of guidance, and review and approval of marketing applications. Notably, on January 23, 2017, President Trump ordered a civilian hiring freeze for all executive departments and agencies, including the FDA, which prohibits the FDA from filling employee vacancies or creating new positions. Under the terms of the order, the freeze was to remain in effect until implementation of a plan to be recommended by the Director for the Office of Management and Budget ("OMB") in consultation with the Director of the Office of Personnel Management, to reduce the size of the federal workforce through attrition. An under-staffed FDA could result in delays in FDA's responsiveness or in its ability to review submissions or applications, issue regulations or guidance or implement or enforce regulatory requirements in a timely fashion or at all. This hiring freeze was lifted later in 2017. Moreover, on January 30, 2017, President Trump issued an Executive Order, applicable to all executive agencies, including the FDA, which requires that for each notice of proposed rulemaking or final regulation to be issued in fiscal year 2017, the agency shall identify at least two existing regulations to be repealed, unless prohibited by law. These requirements are referred to as the "two-for-one" provisions. This Executive Order includes a budget neutrality provision that requires the total incremental cost of all new regulations in the 2017 fiscal year, including repealed regulations, to be no greater than zero, except in limited circumstances. For fiscal years 2018 and beyond, the Executive Order requires agencies to identify regulations to offset any incremental cost of a new regulation and approximate the total costs or savings associated with each new regulation or repealed regulation. In interim guidance issued by the Office of Information and Regulatory Affairs within OMB on February 2, 2017, the administration indicates that the "two-for-one" provisions may apply not only to agency regulations, but also to significant agency guidance documents. It is difficult to predict how these requirements will be implemented, and the extent to which they will impact the FDA's ability to exercise its regulatory authority. If these executive actions impose constraints on the FDA's ability to engage in oversight and implementation activities in the normal course, our business may be negatively impacted.

In addition, on October 12, 2017, the President released an Executive Order intended to promote health care choices and competition and on June 24, 2019, the President released an Executive Order intended to improve price transparency and quality transparency. These may push HHS, FDA, and other relevant agencies to engage in rulemaking that may impact the pharmaceutical industry.

If, in the future, we are unable to establish sales and marketing capabilities or enter into agreements with third parties to sell and market our drug candidates, we may not be successful in commercializing our drug candidates if and when they are approved, and we may not be able to generate any revenue.

We do not currently have a sales or marketing infrastructure and have limited experience in the sale, marketing or distribution of drugs. To achieve commercial success for any approved drug candidate for which we retain sales and marketing responsibilities, we must build our sales, marketing, managerial, and other non-technical capabilities or make arrangements with third parties to perform these services. In the future, we may choose to build a focused sales and marketing infrastructure to sell, or participate in sales activities with our collaborators for, some of our drug candidates if and when they are approved.

In advance of FDA approval of our first product, we will need to make significant investments to build a commercial organization and infrastructure. We will need to hire a sales force and commercial support personnel, in order to build processes and systems to support a commercial launch prior to knowing whether our product will receive FDA approval. It is possible that the FDA approval is unexpectedly delayed or our product is not approved at all. In either case we will incur delays that may impede or significantly delay our ability to generate revenue and at the same time will incur significant expenses. If this were to occur, it would have a material adverse effect on the Company.

There are risks involved with both establishing our own sales and marketing capabilities and entering into arrangements with third parties to perform these services. For example, recruiting and training a sales force is expensive and time-consuming and could delay any drug launch. If the commercial launch of a drug candidate for which we recruit a sales force and establish marketing capabilities is delayed or does not occur for any reason, we would have prematurely or unnecessarily incurred these commercialization expenses. This may be costly, and our investment would be lost if we cannot retain or reposition our sales and marketing personnel.

Factors that may inhibit our efforts to commercialize our drug candidates on our own include:

- our inability to recruit and retain adequate numbers of effective sales and marketing personnel;
- the inability of sales personnel to obtain access to physicians or persuade adequate numbers of physicians to prescribe any future drugs;
- the lack of complementary drugs to be offered by sales personnel, which may put us at a competitive disadvantage relative to companies with more extensive product lines; and
- unforeseen costs and expenses associated with creating an independent sales and marketing organization.

If we enter into arrangements with third parties to perform sales, marketing and distribution services, our drug revenues or the profitability of these drug revenues to us are likely to be lower than if we were to market and sell any drug candidates that we develop ourselves. In addition, we may not be successful in entering into arrangements with third parties to sell and market our drug candidates or may be unable to do so on terms that are favorable to us. We likely will have little control over such third parties, and any of them may fail to devote the necessary resources and attention to sell and market our drug candidates effectively. If we do not establish sales and marketing capabilities successfully, either on our own or in collaboration with third parties, we will not be successful in commercializing our drug candidates. Further, our business, results of operations, financial condition and prospects will be materially adversely affected.

Our relationships with customers and third-party payors will be subject to applicable anti-kickback, fraud and abuse and other healthcare laws and regulations, which could expose us to criminal sanctions, civil penalties, exclusion from government healthcare programs, contractual damages, reputational harm and diminished profits and future earnings.

Although we do not currently have any drugs on the market, once we begin commercializing our drug candidates, we will be subject to additional healthcare statutory and regulatory requirements and enforcement by the federal government and the states and foreign governments in which we conduct our business. Healthcare providers, physicians and third-party payors play a primary role in the recommendation and prescription of any drug candidates for which we obtain marketing approval. Our future arrangements with third-party payors and customers may expose us to broadly applicable fraud and abuse and other healthcare laws and regulations that may constrain the business or financial arrangements and relationships through which we market, sell and distribute our drug candidates for which we obtain marketing approval. Restrictions under applicable federal and state healthcare laws and regulations include the following:

- the federal Anti-Kickback Statute prohibits, among other things, persons from knowingly and willfully soliciting, offering, receiving or providing remuneration, directly or indirectly, in cash or in kind, to induce or reward either the referral of an individual for, or the purchase, order or recommendation of, any good or service, for which payment may be made under federal and state healthcare programs such as Medicare and Medicaid. A person or entity does not need to have actual knowledge of the statute or specific intent to violate it in order to have committed a violation;
- the federal False Claims Act imposes civil penalties, including through civil whistleblower or qui tam actions, against individuals or entities for, among other things, knowingly presenting, or causing to be presented, to the federal government, claims for payment that are false or fraudulent or making a false statement to avoid, decrease or conceal an obligation to pay money to the federal government. In addition, the government may assert that a claim including items and services resulting from a violation of the federal Anti-Kickback Statute constitutes a false or fraudulent claim for purposes of the False Claims Act;
- the federal Health Insurance Portability and Accountability Act of 1996, or HIPAA, imposes criminal and civil liability for executing a scheme to defraud any healthcare benefit program, or knowingly and willfully falsifying, concealing or covering up a material fact or making any materially false statement in connection with the delivery of or payment for healthcare benefits, items or services; similar to the federal Anti-Kickback Statute, a person or entity does not need to have actual knowledge of the statute or specific intent to violate it in order to have committed a violation;
- the federal physician payment transparency requirements, sometimes referred to as the "Sunshine Act" under the Affordable Care Act require manufacturers of drugs, devices, biologics and medical supplies that are reimbursable under Medicare, Medicaid, or the Children's Health Insurance Program to report to the Department of Health and Human Services information related to physician payments and other transfers of value and the ownership and investment interests of such physicians and their immediate family members;
- HIPAA, as amended by the Health Information Technology for Economic and Clinical Health Act of 2009 and its implementing regulations,
 which also imposes obligations on certain covered entity healthcare providers, health plans, and healthcare clearinghouses as well as their
 business associates that perform certain services involving the use or disclosure of individually identifiable health information, including
 mandatory contractual terms, with respect to safeguarding the privacy, security and transmission of individually identifiable health information;
- federal consumer protection and unfair competition laws, which broadly regulate marketplace activities and activities that potentially harm consumers; and
- analogous state laws and regulations, such as state anti-kickback and false claims laws that may apply to sales or marketing arrangements and claims involving healthcare items or services reimbursed by non-governmental third-party payors, including private insurers; and some state laws require pharmaceutical companies to comply with the pharmaceutical industry's voluntary compliance guidelines and the relevant compliance guidance promulgated by the federal government in addition to requiring drug manufacturers to report information related to payments to physicians and other health care providers or marketing expenditures, and state laws governing the privacy and security of health information in certain circumstances, many of which differ from each other in significant ways and often are not preempted by HIPAA, thus complicating compliance efforts.

Ensuring that our future business arrangements with third parties comply with applicable healthcare laws and regulations could involve substantial costs. It is possible that governmental authorities will conclude that our business practices do not comply with current or future statutes, regulations or case law involving applicable fraud and abuse or other healthcare laws and regulations. If our operations, including anticipated activities to be conducted by our sales team, were to be found to be in violation of any of these laws or any other governmental regulations that may apply to us, we may be subject to significant civil, criminal and administrative penalties, damages, fines, exclusion from government-funded healthcare programs, such as Medicare and Medicaid, and the curtailment or restructuring of our operations. If any of the physicians or other providers or entities with whom we expect to do business is found to be not in compliance with applicable laws, they may be subject to criminal, civil or administrative sanctions, including exclusions from government-funded healthcare programs.

If we fail to adequately understand and comply with the local laws and customs as we expand into new international markets, these operations may incur losses or otherwise adversely affect our business and res ults of operations.

We expect to operate a portion of our business in certain countries through subsidiaries or through supply and marketing arrangements. In those countries where we have limited experience in operating subsidiaries and in reviewing equity investees, we will be subject to additional risks related to complying with a wide variety of national and local laws, including restrictions on the import and export of certain intermediates, drugs, technologies and multiple and possibly overlapping tax laws. In addition, we may face competition in certain countries from companies that may have more experience with operations in such countries or with international operations generally. We may also face difficulties integrating new facilities in different countries into our existing operations, as well as integrating employees hired in different countries into our existing corporate culture. If we do not effectively manage our operations in these subsidiaries and review equity investees effectively, or if we fail to manage our alliances, we may lose money in these countries and it may adversely affect our business and results of our operations. In all interactions with foreign regulatory authorities, we are exposed to liability risks under the Foreign Corrupt Practices Act or similar anti-bribery laws.

Any product for which we obtain marketing approval could be subject to restrictions or withdrawal from the market and we may be subject to penalties if we fail to comply with regulatory requirements or if we experience unanticipated problems with products.

Any product for which we obtain marketing approval, along with the manufacturing processes and facilities, post-approval clinical data, labeling, advertising and promotional activities for such product, will be subject to continual requirements of, and review by, the FDA and comparable regulatory authorities. These requirements include submissions of safety and other post-marketing information and reports, registration requirements, current Good Manufacturing Practice ("cGMP") requirements relating to quality control, quality assurance and corresponding maintenance of records and documents, requirements regarding the distribution of samples to physicians and recordkeeping, and requirements regarding company presentations and interactions with healthcare professionals. Even if we obtain regulatory approval of a product, the approval may be subject to limitations on the indicated uses for which the product may be marketed, be subject to conditions of approval, or contain requirements for costly post-marketing testing and surveillance to monitor the safety and/or efficacy of the product. We also may be subject to state laws and registration requirements covering the distribution of drug products. Later discovery of previously unknown problems with products, manufacturers or manufacturing processes, or failure to comply with regulatory requirements, may result in actions such as:

- restrictions on product manufacturing, distribution or use;
- restrictions on the labeling or marketing of a product;
- requirements to conduct post-marketing studies or clinical trials;
- warning letters;
- withdrawal of the products from the market;
- refusal to approve pending applications or supplements to approved applications that we or our subsidiaries submit;
- recalls:
- fines:
- suspension or withdrawal of marketing or regulatory approvals;
- refusal to permit the import or export of products;
- product seizure or detentions;
- injunctions or the imposition of civil or criminal penalties; and
- adverse publicity.

If we, or our respective suppliers, third-party contractors, clinical investigators or collaborators are slow to adapt, or are unable to adapt, to changes in existing regulatory requirements or adoption of new regulatory requirements or policies, we, our subsidiaries, or our respective collaborators may be subject to the actions listed above, including losing marketing approval for products, resulting in decreased revenue from milestones, product sales or royalties.

We will need to obtain FDA approval of any proposed product brand names, and any failure or delay associated with such approval may adversely impact our business.

A pharmaceutical product candidate cannot be marketed in the United States or other countries until we have completed a rigorous and extensive regulatory review processes, including approval of a brand name. Any brand names we intend to use for ublituximab, umbralisib or any future product candidates will require approval from the FDA regardless of whether we have secured a formal trademark registration from the United States Patent and Trademark Office ("USPTO"). The FDA typically conducts a review of proposed product brand names, including an evaluation of potential for confusion with other product names. The FDA may also object to a product brand name if it believes the name inappropriately implies medical claims. If the FDA objects to any of our proposed product brand names, we may be required to adopt an alternative brand name for ublituximab, umbralisib or any future product candidates. If we adopt an alternative brand name, we would lose the benefit of our existing trademark applications for such product candidate and may be required to expend significant additional resources in an effort to identify a suitable product brand name that would qualify under applicable trademark laws, not infringe the existing rights of third parties and be acceptable to the FDA. We may be unable to build a successful brand identity for a new trademark in a timely manner or at all, which would limit our ability to commercialize ublituximab, umbralisib, or any future product candidates. We do not currently have an agreed upon brand name for umbralisib, and no assurance can be given that we will obtain one in a timely fashion. Any delay in obtaining a brand name for umbralisib or any other of our drug candidates could delay approval and/or commercialization and have a negative impact on our launch and future prospects for umbralisib or any other such drug candidates.

Risks Related to Our Dependence on Third Parties

We rely on third parties to generate clinical, preclinical and quality data necessary to support the regulatory applications needed to conduct clinical trials and file for marketing approval. We rely on third parties to help conduct our planned clinical trials. If these third parties do not perform their services as required, we may not be able to obtain regulatory approval for or commercialize our product candidates when expected or at all.

In order to submit and maintain an Investigational New Drug application ("IND"), BLA, or NDA to the FDA, it is necessary to submit all information on the clinical, non-clinical, chemistry, manufacturing, controls and quality aspects of the product candidate. We rely on our third party contractors and our licensing partners to provide portions of this data. If we are unable to obtain this data, or the data is not sufficient to meet the regulatory requirements, we may experience significant delays in our development programs. While we maintain an active IND for ublituximab and umbralisib enabling the conduct of studies in the FDA's Division of Hematology and Oncology, and an active IND for ublituximab under the FDA's Division of Neurology, there can be no assurance that the FDA will allow us to continue the development of our product candidates in those divisions where we maintain an active IND.

Additionally, we use CRO's to assist in the conduct of our current clinical trials and expect to use such services for future clinical trials and we rely upon medical institutions, clinical investigators and contract laboratories to conduct our trials in accordance with our clinical protocols and appropriate regulations. Our current and future CROs, investigators and other third parties play a significant role in the conduct of our trials and the subsequent collection and analysis of data from the clinical trials. There is no guarantee that any CROs, investigators and other third parties will devote adequate time and resources to our clinical trials or perform as contractually required. If any third parties upon whom we rely for administration and conduct of our clinical trials fail to meet expected deadlines, fail to adhere to its clinical protocols or otherwise perform in a substandard manner, our clinical trials may be extended, delayed or terminated, and we may not be able to commercialize our product candidates. In addition to the third parties identified above, we are also heavily reliant on the conduct of our patients enrolled to our studies by our third-party investigators. We rely on our clinical trial sites and investigators to properly identify and screen qualified candidates for our clinical trials, and for them to ensure participants adhere to our clinical protocol requirements. The majority of our clinical trial conduct occurs in the out-patient setting, where patients are expected to continue to adhere to our study protocol specified requirements. The ability of our enrolled patients to properly identify, document, and report adverse events; take protocol specified study drugs at the correct quantity, time, and setting, as applicable; avoid contraindicated medications; and comply with other protocol specified procedures such as returning to the trial site for scheduled laboratory and disease assessments, is wholly out of our control. Deviations from protocol procedures, such as those identified previously, could materially affect the quality of our clinical trial data, and therefore ultimately affect our ability to develop and commercialize our drug candidates. If any of our clinical trial sites terminates for any reason, we may experience the loss of follow-up information on patients enrolled in our ongoing clinical trials unless we are able to transfer the care of those patients to another qualified clinical trial site. If any of our clinical trial sites are required by the FDA or IRB to close down due to data management or patient management or any other issues we may lose patients. In our MS Phase 2 trial, during routine monitoring and site audits, significant Good Clinical Practice (GCP) violations and other noncompliance issues were identified at one of our US-based large academic sites. The investigator left the institution; shortly thereafter the site terminated their participation in our study, before all data could be source document verified. While we do not believe this will have any effect on the overall results of the MS Phase 2 trial, sensitivity analyses excluding data from this site will be performed and no assurance can be given that the results were not affected.

Whether conducted through a CRO or through our internal staff, we are solely responsible for ensuring that each of our clinical trials are conducted in accordance with the applicable protocol, legal and regulatory requirements and scientific standards, and our reliance on CROs will not relieve us of our regulatory responsibilities. For any violations of laws and regulations during the conduct of our clinical trials, we could be subject to warning letters or enforcement action that may include civil penalties up to and including criminal prosecution. We and our CROs are required to comply with regulations, including GCP Guidelines for conducting, monitoring, recording and reporting the results of clinical trials to ensure that the data and results are scientifically credible and accurate, and that the trial patients are adequately informed of the potential risks of participating in clinical trials and their rights are protected. These regulations are enforced by the FDA, the Competent Authorities of the Member States of the European Economic Area and comparable foreign regulatory authorities for any drugs in clinical development. The FDA enforces GCP regulations through periodic inspections of clinical trial sponsors, principal investigators and trial sites. If we or our CROs fail to comply with applicable GCPs, the clinical data generated in our clinical trials may be deemed unreliable and the FDA or comparable foreign regulatory authorities may require us to perform additional clinical trials before approving our marketing applications. We cannot assure you that, upon inspection, the FDA will determine that our current or future clinical trials comply with GCPs. In addition, our clinical trials must be conducted with drug candidates produced under cGMPs regulations. Our failure or the failure of our CROs to comply with these regulations may require us to repeat clinical trials, which would delay the regulatory approval process and could also subject us to enforcement action. We also are required to register mo

Although we intend to design the clinical trials for our drug candidates, CROs play an important role in the conduct of our clinical trials, especially outside of the United States. As a result, many important aspects of our development programs, including their conduct and timing, will be outside of our direct control. Our reliance on third parties to conduct current or future clinical trials will also result in less direct control over the management of data developed through clinical trials than would be the case if we were relying entirely upon our own staff. Communicating with outside parties can also be challenging, potentially leading to mistakes as well as difficulties in coordinating activities. Outside parties may:

- have staffing difficulties;
- fail to comply with contractual obligations;
- experience regulatory compliance issues;
- undergo changes in priorities or become financially distressed; or
- form relationships with other entities, some of which may be our competitors.

These factors may materially adversely affect the willingness or ability of third parties to conduct our clinical trials and may subject us to unexpected cost increases that are beyond our control. If the CROs do not perform clinical trials in a satisfactory manner, breach their obligations to us or fail to comply with regulatory requirements, the development, regulatory approval and commercialization of our drug candidates may be delayed, we may not be able to obtain regulatory approval and commercialize our drug candidates, or our development program may be materially and irreversibly harmed. If we are unable to rely on clinical data collected by our CROs, we could be required to repeat, extend the duration of, or increase the size of any clinical trials we conduct and this could significantly delay commercialization and require significantly greater expenditures.

If any of our relationships with these third-party CROs terminate, we may not be able to enter into arrangements with alternative CROs. If CROs do not successfully carry out their contractual duties or obligations or meet expected deadlines, if they need to be replaced or if the quality or accuracy of the clinical data they obtain is compromised due to the failure to adhere to our clinical protocols, regulatory requirements or for other reasons, any clinical trials such CROs are associated with may be extended, delayed or terminated, and we may not be able to obtain regulatory approval for or successfully commercialize our drug candidates. As a result, we believe that our financial results and the commercial prospects for our drug candidates in the subject indication would be harmed, our costs could increase and our ability to generate revenue could be delayed.

We contract with third parties for the manufacture of our drug candidates for pre-clinical development and clinical trials, and we expect to continue to do so for commercialization. This reliance on third parties increases the risk that we will not have sufficient quantities of our drug candidates or drugs or such quantities at an acceptable cost, which could delay, prevent or impair our development or commercialization efforts.

We do not currently own or operate, nor do we have any plans to establish in the future, any manufacturing facilities or personnel. We rely, and expect to continue to rely, on third parties for the manufacture of our drug candidates for pre-clinical development and clinical testing, as well as for the commercial manufacture of our drugs if any of our drug candidates receive marketing approval. This reliance on third parties increases the risk that we will not have sufficient quantities of our drug candidates or drugs or such quantities at an acceptable cost or quality, which could delay, prevent or impair our development or commercialization efforts.

The facilities used by our contract manufacturers to manufacture our drug candidates must be approved by the FDA pursuant to inspections that will be conducted after we submit our marketing applications to the FDA. We do not control the manufacturing process of, and will be completely dependent on, our contract manufacturers for compliance with cGMPs in connection with the manufacture of our drug candidates. If our contract manufacturers cannot successfully manufacture material that conforms to our specifications and the strict regulatory requirements of the FDA or others, they will not be able to secure and/or maintain regulatory approval for their manufacturing facilities. In addition, we have no control over the ability of our contract manufacturers to maintain adequate quality control, quality assurance and qualified personnel. If the FDA or a comparable foreign regulatory authority does not approve these facilities for the manufacture of our drug candidates or if it withdraws any such approval in the future, we may need to find alternative manufacturing facilities, which would significantly impact our ability to develop, obtain regulatory approval for or market our drug candidates, if approved. Further, our failure, or the failure of our third-party manufacturers, to comply with applicable regulations could result in sanctions being imposed on us, including clinical holds, fines, injunctions, civil penalties, delays, suspension or withdrawal of approvals, license revocation, seizures or recalls of drug candidates or drugs, if approved, operating restrictions and criminal prosecutions, any of which could significantly and adversely affect our business and supplies of our drug candidates.

We do not have any long-term supply agreements with all our contract manufacturers, and in those instances where we do not, we purchase our required drug supply, including the drug product and drug substance on a purchase order basis. In addition, we may be unable to establish or maintain any agreements with third-party manufacturers or to do so on acceptable terms. No assurance can be given that a long-term, scalable manufacturer can be identified or that they can make clinical and commercial supplies of our product candidates that meets the product specifications of previously manufactured batches, or is of a sufficient quality, or at an appropriate scale and cost to make it commercially feasible. If they are unable to do so, it could have a material adverse impact on our business.

Even if we are able to establish and maintain agreements with third-party manufacturers, reliance on third-party manufacturers entails additional risks, including:

- reliance on the third party for regulatory compliance and quality assurance;
- the possible breach of the manufacturing agreement by the third party;
- the possible misappropriation of our proprietary information, including our trade secrets and know-how; and
- the possible termination or nonrenewal of the agreement by the third party at a time that is costly or inconvenient for us.

Moreover, our current long-term supply agreements and, we would expect all future long-term supply agreements would contain certain minimum purchases in what are commonly referred to as "take or pay" provisions. To the extent our demand does not meet the minimum supply required amounts, we would be forced to pay more than desired. This could create a situation where we are spending more than required and could impact our on-going operations and entail curtailing other important research and development or commercialization efforts. All of which could have a material adverse effect on the Company.

Our drug candidates and any drugs that we may develop may compete with other drug candidates and approved drugs for access to manufacturing facilities. There are a limited number of manufacturers that operate under cGMP regulations and that might be capable of manufacturing for us.

Any performance failure on the part of our existing or future manufacturers could delay clinical development or marketing approval. If our current contract manufacturers cannot perform as agreed, we may be required to replace such manufacturers causing additional costs and delays in identifying and qualifying any such replacement.

Our current and anticipated future dependence upon others for the manufacture of our drug candidates or drugs could result in significant delays or gaps in availability of such drug candidates or drugs and may adversely affect our future profit margins and our ability to commercialize any drugs that receive marketing approval on a timely and competitive basis.

We also expect to rely on other third parties to store and distribute drug supplies for our clinical trials. Any performance failure on the part of our distributors could delay clinical development or marketing approval of any future product candidates or commercialization of our products, producing additional losses and depriving us of potential product revenue.

In addition, we do not have the capability to package finished products for distribution to hospitals and other customers. Prior to commercial launch, we intend to enter into agreements with one or more alternate fill/finish pharmaceutical product suppliers so that we can ensure proper supply chain management once we are authorized to make commercial sales of our product candidates. If we receive marketing approval from the FDA, we intend to sell pharmaceutical product finished and packaged by such suppliers. We have not entered into long-term agreements with our fill/finish suppliers, and we may be unable to enter into such an agreement or do so on commercially reasonable terms, which could have a material adverse impact upon our business.

The third parties upon whom we rely for the supply of the active pharmaceutical ingredient ("API"), drug product, regulatory starting materials and intermediates for drug substance and other materials used in our drug candidates are our sole source of supply, and the loss of any of these suppliers could significantly harm our business.

The API, drug product and drug substance used in our drug candidates are currently supplied to us from single-source suppliers. Our ability to successfully develop our drug candidates, supply our drug candidates for clinical trials and to ultimately supply our commercial drugs in quantities sufficient to meet the market demand, depends in part on our ability to obtain the API, drug product and drug substance for these drugs in accordance with regulatory requirements and in sufficient quantities for clinical testing and commercialization. If any of our suppliers ceases its operations for any reason or is unable or unwilling to supply API, drug product, drug substance and other materials in sufficient quantities or on the timelines necessary to meet our needs, it could significantly and adversely affect our business, the supply of our drug candidates and our financial condition.

In most cases, our manufacturing partners are single source suppliers. It is expected that our manufacturing partners will be sole source suppliers from single site locations for the foreseeable future. Various raw materials, components, and testing services required for our products may also be single sourced. Given this, any disruption of supply from these partners could have a material, long-term impact on our ability to supply products for clinical trials or commercial sale. If our suppliers do not deliver sufficient quantities of our product candidates on a timely basis, or at all, and in accordance with applicable specifications, there could be a significant interruption of our supply, which would adversely affect clinical development and commercialization of our products. In addition, if our current or future supply of any of our product candidates should fail to meet specifications during its stability program there could be a significant interruption of our supply of drug, which would adversely affect the clinical development and commercialization of the product.

For all of our drug candidates, we plan to identify and qualify additional manufacturers and other suppliers to provide such API, drug product and drug substance prior to or following submission of an NDA to the FDA and/or an MAA to the EMA. We are not certain, however, that our single-source suppliers will be able to meet our demand for their products, either because of the nature of our agreements with those suppliers, our limited experience with those suppliers or our relative importance as a customer to those suppliers. It may be difficult for us to assess their ability to timely meet our demand in the future based on past performance. While our suppliers have generally met our demand for their products on a timely basis in the past, they may subordinate our needs in the future to their other customers.

Establishing additional or replacement suppliers for the API, drug product and drug substance used in our drug candidates, if required, may not be accomplished quickly or at all. If we are able to find a replacement supplier, such replacement supplier would need to be qualified and may require additional regulatory approval, which could result in further delay. While we seek to maintain adequate inventory of the API, drug product and drug substance used in our drug candidates, any interruption or delay in the supply of components or materials, or our inability to obtain such API, drug product and drug substance from alternate sources at acceptable prices in a timely manner could impede, delay, limit or prevent our development efforts, which could harm our business, results of operations, financial condition and prospects.

Because we have in-licensed our product candidates from third parties, any dispute with or non-performance by our licensors will adversely affect our ability to develop and commercialize the applicable product candidates.

Because we license our intellectual property from third parties and we expect to continue to in-license additional intellectual property rights, if there is any dispute between us and our licensor regarding our rights under a license agreement, our ability to develop and commercialize our product candidates may be adversely affected. Disputes may arise with the third parties from whom we license our intellectual property rights from for a variety of reasons, including:

- the scope of rights granted under the license agreement and other interpretation-related issues;
- the extent to which our technology and processes infringe on intellectual property of the licensor that is not subject to the license agreement;
- the sublicensing of patent and other rights under our collaborative development relationships and obligations associated with sublicensing;
- our diligence obligations under the license agreement and what activities satisfy those diligence obligations;
- the ownership of inventions and know-how resulting from the joint creation or use of intellectual property by our licensors and us and our partners; and
- the priority of invention of patented technology.

In addition, the agreements under which we currently license intellectual property or technology from third parties are complex, and certain provisions in such agreements may be susceptible to multiple interpretations, or may conflict in such a way that puts us in breach of one or more agreements, which would make us susceptible to lengthy and expensive disputes with one or more of our licensing partners. The resolution of any contract interpretation disagreement that may arise could narrow what we believe to be the scope of our rights to the relevant intellectual property or technology, or increase what we believe to be our financial or other obligations under the relevant agreement, either of which could have a material adverse effect on our business, financial condition, results of operations, and prospects. Moreover, if disputes over intellectual property that we have licensed prevent or impair our ability to maintain our current licensing arrangements on commercially acceptable terms, we may be unable to successfully develop and commercialize the affected product candidates, which could have a material adverse effect on our business, financial conditions, results of operations, and prospects.

If conflicts arise between us and our future collaborators or strategic partners, these parties may act in a manner adverse to us and could limit our ability to implement our strategies.

If conflicts arise between our future corporate or academic collaborators or strategic partners and us, the other party may act in a manner adverse to us and could limit our ability to implement our strategies. Future collaborators or strategic partners, may develop, either alone or with others, products in related fields that are competitive with the products or potential products that are the subject of these collaborations. Competing products, either developed by the collaborators or strategic partners or to which the collaborators or strategic partners have rights, may result in the withdrawal of partner support for any future product candidates. Our current or future collaborators or strategic partners may preclude us from entering into collaborations with their competitors, fail to obtain timely regulatory approvals, terminate their agreements with us prematurely, or fail to devote sufficient resources to the development and commercialization of products. Any of these developments could harm any future product development efforts.

We may seek to establish additional collaborations, and, if we are not able to establish them on commercially reasonable terms, we may have to alter our development and commercialization plans.

Our drug development programs and the potential commercialization of our drug candidates will require substantial additional cash to fund expenses. For some of our drug candidates, we may decide to collaborate with additional pharmaceutical and biotechnology companies for the development and potential commercialization of those drug candidates.

We face significant competition in seeking appropriate collaborators. Whether we reach a definitive agreement for a collaboration will depend, among other things, upon our assessment of the collaborator's resources and expertise, the terms and conditions of the proposed collaboration, and the proposed collaborator's evaluation of a number of factors. Those factors may include the design or results of our clinical trials, the likelihood of approval by the FDA or similar regulatory authorities outside the United States, the potential market for the subject drug candidate, the costs and complexities of manufacturing and delivering such drug candidate to patients, the potential of competing drugs, the existence of uncertainty with respect to our ownership of technology, which can exist if there is a challenge to such ownership without regard to the merits of the challenge and industry and market conditions generally. The collaborator may also consider alternative drug candidates or technologies for similar indications that may be available to collaborate on and whether such a collaboration could be more attractive than the one with us for our drug candidate. The terms of any additional collaborations or other arrangements that we may establish may not be favorable to us.

We may be restricted under our collaboration agreements from entering into future agreements on certain terms with potential collaborators. Collaborations are complex and time-consuming to negotiate and document. In addition, there have been a significant number of recent business combinations among large pharmaceutical companies that have resulted in a reduced number of potential future collaborators.

We may not be able to negotiate additional collaborations on a timely basis, on acceptable terms, or at all. If we are unable to do so, we may have to curtail the development of the drug candidate for which we are seeking to collaborate, reduce or delay its development program or one or more of our other development programs, delay its potential commercialization or reduce the scope of any sales or marketing activities, or increase our expenditures and undertake development or commercialization activities at our own expense. If we elect to increase our expenditures to fund development or commercialization activities on our own, we may need to obtain additional capital, which may not be available to us on acceptable terms or at all. If we do not have sufficient funds, we may not be able to further develop our drug candidates or bring them to market and generate drug revenue.

Any future collaborations that we enter into may not be successful. The success of our collaboration arrangements will depend heavily on the efforts and activities of our collaborators. Collaborators generally have significant discretion in determining the efforts and resources that they will apply to these collaborations. Disagreements between parties to a collaboration arrangement regarding clinical development and commercialization matters can lead to delays in the development process or commercializing the applicable drug candidate and, in some cases, termination of the collaboration arrangement. These disagreements can be difficult to resolve if neither of the parties has final decision-making authority. Collaborations with pharmaceutical or biotechnology companies and other third parties often are terminated or allowed to expire by the other party. Any termination or expiration of any future collaboration agreement could adversely affect us financially or harm our business reputation.

Risks Relating to Our Intellectual Property

Our success depends upon our ability to obtain and protect our intellectual property and proprietary technologies and if the scope of our patent protection obtained is not sufficiently broad, our competitors could develop and commercialize technology and drugs similar or identical to ours, and our ability to successfully commercialize our technology and drugs may be impaired.

Our commercial success in part depends on obtaining and maintaining patent protection and trade secret protection in the United States and other countries with respect to our product candidates or any future product candidate that we may license or acquire, their formulations and uses and the methods we use to manufacture them, as well as successfully defending these patents against third-party challenges. We seek to protect our proprietary and intellectual property position by filing patent applications in the United States and abroad related to our novel technologies and product candidates, and by maintenance of our trade secrets through proper procedures.

We will only be able to protect our technologies from unauthorized use by third parties to the extent that valid and enforceable patents or trade secrets cover them in the market they are being used or developed. The degree of patent protection we require to successfully commercialize our drug candidates may be unavailable or severely limited in some cases and may not adequately protect our rights or permit us to gain or keep any competitive advantage. We cannot provide any assurances that any of our patents have, or that any of our pending patent applications that mature into issued patents will include, claims with a scope sufficient to protect any of our drug candidates. In addition, the laws of foreign countries may not protect our rights to the same extent as the laws of the United States.

Furthermore, patents have a limited lifespan. In the United States, the natural expiration of a patent is generally twenty years after it is filed. Various extensions may be available; however, the life of a patent, and the protection it affords, is limited. Given the amount of time required for the development, testing and regulatory review of new drug candidates, patents protecting such candidates might expire before or shortly after such candidates are commercialized. As a result, our owned patent portfolio may not provide us with adequate and continuing patent protection sufficient to exclude others from commercializing drugs similar or identical to our drug candidates, including generic versions of such drugs.

Currently, the composition of matter patent for ublituximab and umbralisib are granted in both the United States and EU, among other countries. A method of use patent covering the combination of ublituximab and umbralisib has also been granted in the US, EU, Japan, and several other territories. Additionally, several method of use patents for ublituximab and umbralisib in various indications and settings have also been applied for but have not yet been issued, or have been issued in certain territories but not under all jurisdictions in which such applications have been filed. No patents to date have been issued for TG-1501, TG-1701 and TG-1801 or for our pre-clinical product candidates. There can be no guarantee that any of these patents for which an application has already been filed, nor any patents filed in the future for our product candidates will be granted in any or all jurisdictions in which there were filed, or that all claims initially included in such patent applications will be allowed in the final patent that is issued. The patent application process is subject to numerous risks and uncertainties, and there can be no assurance that we or our partners will be successful in protecting our product candidates by obtaining and defending patents, or what the scope of an issued patent may ultimately be.

These risks and uncertainties include the following:

- the patent applications that we or our partners file may not result in any patents being issued;
- patents that may be issued or in-licensed may be challenged, invalidated, modified, revoked or circumvented, or otherwise may not provide any competitive advantage;
- as of March 16, 2013, the United States converted from a "first to invent" to a "first to file" system. If we do not win the filing race, we will not be entitled to inventive priority;
- our competitors, many of which have substantially greater resources than we do, and many of which have made significant investments in competing technologies, may seek, or may already have obtained, patents that will limit, interfere with, or eliminate its ability to file new patent applications or make, use, and sell our potential products either in the United States or in international markets;
- there may be significant pressure on the United States government and other international governmental bodies to limit the scope of patent
 protection both inside and outside the United States for disease treatments that prove successful as a matter of public policy regarding worldwide
 health concerns; and
- countries other than the United States may have less restrictive patent laws than those upheld by United States courts, allowing foreign
 competitors the ability to exploit these laws to create, develop, and market competing products.

If patents are not issued that protect our product candidates, it could have a material adverse effect on our financial condition and results of operations.

In addition, the patent prosecution process is expensive and time-consuming, and we may not be able to file and prosecute all necessary or desirable patent applications at a reasonable cost or in a timely manner. Further, with respect to some of the pending patent applications covering our drug candidates, prosecution has yet to commence. Patent prosecution is a lengthy process, during which the scope of the claims initially submitted for examination by the U.S. Patent and Trademark Office, or USPTO, have been significantly narrowed by the time they issue, if at all. It is also possible that we will fail to identify any patentable aspects of our research and development output and methodology, and, even if we do, an opportunity to obtain patent protection may have passed. Given the uncertain and time-consuming process of filing patent applications and prosecuting them, it is possible that our product(s) or process(es) originally covered by the scope of the patent application may have changed or been modified, leaving our product(s) or process(es) without patent protection. Moreover, in some circumstances, we do not have the right to control the preparation, filing and prosecution of patent applications, or to maintain the patents, covering technology that we license from third parties. Therefore, these patents and applications may not be prosecuted and enforced in a manner consistent with the best interests of our business. If our licensors or we fail to appropriately prosecute and maintain patent protection or trade secret protection for one or more product candidates or any future product candidate we may license or acquire, our ability to develop and commercialize these product candidates may be adversely affected and we may not be able to prevent competitors from making, using and selling competing products. This failure to properly protect the intellectual property rights relating to these product candidates could impair our ability to compete in the market and adversely affect our ability to generate revenues and achieve profitability, which would have a material adverse effect on our financial condition and results of operations. Furthermore, should we enter into other collaborations, including out-licensing or partnerships, we may be required to consult with or cede control to collaborators regarding the prosecution, maintenance and enforcement of licensed patents. Therefore, these patents and applications may not be prosecuted and enforced in a manner consistent with the best interests of our business.

The patent position of biotechnology and pharmaceutical companies generally is highly uncertain, involves complex legal and factual questions, and has in recent years been the subject of much litigation. In addition, no consistent policy regarding the breadth of claims allowed in pharmaceutical or biotechnology patents has emerged to date in the United States. The patent situation outside the United States is even more uncertain. The laws of foreign countries may not protect our rights to the same extent as the laws of the United States, and we may fail to seek or obtain patent protection in all major markets. For example, European patent law restricts the patentability of methods of treatment of the human body more than United States law does. Our pending and future patent applications may not result in patents being issued which protect our technology or products, in whole or in part, or which effectively prevent others from commercializing competitive technologies and products. Changes in either the patent laws or interpretation of the patent laws in the United States and other countries may diminish the value of our patents or narrow the scope of our patent protection. For example, the federal courts of the United States have taken an increasingly dim view of the patent eligibility of certain subject matter, such as naturally occurring nucleic acid sequences, amino acid sequences and certain methods of utilizing same, which include their detection in a biological sample and diagnostic conclusions arising from their detection. Such subject matter, which had long been a staple of the biotechnology and biopharmaceutical industry to protect their discoveries, is now considered, with few exceptions, ineligible in the first instance for protection under the patent laws of the United States. Accordingly, we cannot predict the breadth of claims that may be allowed or enforced in our patents or in those licensed from a third-party.

In addition, U.S. patent laws may change, which could prevent or limit us or our subsidiaries from filing patent applications or patent claims to protect products and/or technologies or limit the exclusivity periods that are available to patent holders, as well as affect the validity, enforceability, or scope of issued patents. For example, on September 16, 2011, the Leahy-Smith America Invents Act was signed into law. The Leahy-Smith Act includes a number of significant changes to United States patent law. These include changes to transition from a "first-to-invent" system to a "first-to-file" system and to the way issued patents are challenged. The formation of the Patent Trial and Appeal Board now provides a quicker and less expensive process for challenging issued patents.

We may be subject to a third-party pre-issuance submission of prior art to the USPTO, or become involved in opposition, derivation, reexamination, inter parties review, post-grant review or interference proceedings challenging our patent rights or the patent rights of others. The costs of these proceedings could be substantial and it is possible that our efforts to establish priority of invention would be unsuccessful, resulting in a material adverse effect on our United States patent position. An adverse determination in any such submission, patent office trial, proceeding or litigation could reduce the scope of, render unenforceable, or invalidate, our patent rights, allow third parties to commercialize our technology or products and compete directly with us, without payment to us, or result in our inability to manufacture or commercialize products without infringing third-party patent rights. In addition, if the breadth or strength of protection provided by our patents and patent applications is threatened, it could dissuade companies from collaborating with us to license, develop or commercialize current or future product candidates.

The issuance of a patent does not foreclose challenges to its inventorship, scope, validity or enforceability. Therefore, our owned and licensed patents may be challenged in the courts or patent offices in the United States and abroad. Such challenges may result in loss of exclusivity or in patent claims being narrowed, invalidated or held unenforceable, in whole or in part, which could limit our ability to stop others from using or commercializing similar or identical technology and products, or limit the duration of the patent protection of our technology and products. Given the amount of time required for the development, testing and regulatory review of new product candidates, patents protecting such product candidates might expire before or shortly after such product candidates are commercialized. As a result, our owned and licensed patent portfolio may not provide us with sufficient rights to exclude others from commercializing products similar or identical to ours.

Even if our patent applications issue as patents, and they are unchallenged, our issued patents and our pending patents, if issued, may not provide us with any meaningful protection or prevent competitors from designing around our patent claims to circumvent our owned or licensed patents by developing similar or alternative technologies or drugs in a non-infringing manner. For example, a third party may develop a competitive drug that provides benefits similar to one or more of our drug candidates but that has a different composition that falls outside the scope of our patent protection. If the patent protection provided by the patents and patent applications we hold or pursue with respect to our drug candidates is not sufficiently broad to impede such competition, our ability to successfully commercialize our drug candidates could be negatively affected, which would harm our business.

In addition, we may in the future be subject to claims by our former employees or consultants asserting an ownership right in our patents or patent applications, as a result of the work they performed on our behalf. Although we generally require all of our employees, consultants and advisors and any other third parties who have access to our proprietary know-how, information or technology to assign or grant similar rights to their inventions to us, we cannot be certain that we have executed such agreements with all parties who may have contributed to our intellectual property, nor can we be certain that our agreements with such parties will be upheld in the face of a potential challenge, or that they will not be breached, for which we may not have an adequate remedy. An adverse determination in any such submission or proceeding may result in loss of exclusivity or freedom to operate or in patent claims being narrowed, invalidated or held unenforceable, in whole or in part, which could limit our ability to stop others from using or commercializing similar or identical technology and drugs, without payment to us, or could limit the duration of the patent protection covering our technology and drug candidates. Such challenges may also result in our inability to manufacture or commercialize our drug candidates without infringing third-party patent rights. In addition, if the breadth or strength of protection provided by our patents and patent applications is threatened, it could dissuade companies from collaborating with us to license, develop or commercialize current or future drug candidates.

Patent protection and other intellectual property protection are crucial to the success of our business and prospects, and there is a substantial risk that such protections will prove inadequate.

Obtaining and maintaining patent protection depends on compliance with various procedural, document submission, fee payment and other requirements imposed by governmental patent agencies, and our patent protection could be reduced or eliminated for non-compliance with these requirements.

The USPTO and various foreign governmental patent agencies require compliance with a number of procedural, documentary, fee payment and other similar provisions during the patent application process. In addition, periodic maintenance fees on issued patents often must be paid to the USPTO and foreign patent agencies over the lifetime of the patent. While an unintentional lapse can in many cases be cured by payment of a late fee or by other means in accordance with the applicable rules, there are situations in which noncompliance can result in abandonment or lapse of the patent or patent application, resulting in partial or complete loss of patent rights in the relevant jurisdiction.

Non-compliance events that could result in abandonment or lapse of a patent or patent application include, but are not limited to, failure to respond to official actions within prescribed time limits, non-payment of fees and failure to properly legalize and submit formal documents. If we fail to maintain the patents and patent applications covering our drugs or procedures, we may not be able to stop a competitor from marketing drugs that are the same as or similar to our drug candidates, which would have a material adverse effect on our business.

If we or our partners are sued for infringing intellectual property rights of third parties, it will be costly and time consuming, and an unfavorable outcome in that litigation would have a material adverse effect on our business.

Our commercial success depends upon our ability and the ability of our collaborators to develop, manufacture, market and sell our drug candidates and use our proprietary technologies without infringing the proprietary rights and intellectual property of third parties. The biotechnology and pharmaceutical industries are characterized by extensive and frequent litigation regarding patents and other intellectual property rights. We may in the future become party to, or threatened with, adversarial proceedings or litigation regarding intellectual property rights with respect to our drug candidates and technology, including interference proceedings before the USPTO.

Our competitors or other third parties may assert infringement claims against us, alleging that our drugs are covered by their patents. Given the vast number of patents in our field of technology, we cannot be certain that we do not infringe existing patents or that we will not infringe patents that may be granted in the future. Numerous United States and foreign issued patents and pending patent applications, which are owned by third parties, exist in the fields in which we are developing products, some of which may be directed at claims that overlap with the subject matter of our intellectual property. In addition, because patent applications can take many years to issue, there may be currently pending applications, unknown to us, which may later result in issued patents that our product candidates or proprietary technologies may infringe. Similarly, there may be issued patents relevant to our product candidates of which we are not aware. Publications of discoveries in the scientific literature often lag behind the actual discoveries, and patent applications in the United States and other jurisdictions are typically not published until 18 months after a first filing, or in some cases not at all. Therefore, we cannot know with certainty whether we or our licensors were the first to make the inventions claimed in patents or pending patent applications that we own or licensed, or that we or our licensors were the first to file for patent protection of such inventions.

We are aware of certain patents that may pose issues for our commercialization of our drug candidates. For example, Roche, Biogen Idec, and Genentech hold patents for the use of anti-CD20 antibodies utilized in the treatment of CLL in the United States which are expected to expire in November of 2019. While these patents have been challenged, to the best of our knowledge, those matters were settled in a way that permitted additional anti-CD20 antibodies to be marketed for CLL. If those patents are still valid and enforced at the time we are intending to launch ublituximab, then we will need to either prevail in a litigation to challenge those patents or negotiate a settlement agreement with the patent holders. If we decide to initiate proceedings to challenge the validity of these patents in the future, we may be unsuccessful, as courts or patent offices in the United States and abroad could uphold the validity of any such patents. If we were to challenge the validity of any issued United States patent in court, we would need to overcome a statutory presumption of validity that attaches to every United States patent. This means that in order to prevail, we would have to present clear and convincing evidence as to the invalidity of the patent's claims. If we are unable to do so, we may be forced to delay the launch of ublituximab or launch at the risk of litigation for patent infringement, which may have a material adverse effect on our business and results of operations.

If a third party claims that we or any collaborators of ours infringe their intellectual property rights, we may have to defend litigation or administrative proceedings which may be costly whether we win or lose, and which could result in a substantial diversion of our financial and management resources. If we are found to infringe a third party's intellectual property rights, we could be required to obtain a license from such third party to continue developing and marketing our drug candidates and technology. However, we may not be able to obtain any required license on commercially reasonable terms or at all. Even if we were able to obtain such a license, it could be granted on non-exclusive terms, thereby providing our competitors and other third parties access to the same technologies licensed to us. Without such a license, we could be forced, including by court order, to cease developing and commercializing the infringing technology or drug candidates. In addition, we could be found liable for monetary damages, including treble damages and attorneys' fees if we are found to have willfully infringed such third-party patent rights. A finding of infringement could prevent us from commercializing our drug candidates or force us to cease some of our business operations, which could materially harm our business.

No assurance can be given that patents issued to third parties do not exist, have not been filed, or could not be filed or issued, which contain claims covering its products, technology or methods that may encompass all or a portion of our products and methods. Given the number of patents issued and patent applications filed in our technical areas or fields, we believe there is a risk that third parties may allege they have patent rights encompassing our products or methods.

Other product candidates that we may in-license or acquire could be subject to similar risks and uncertainties.

We may need to license cert ain intellectual property from third parties, and such licenses may not be available or may not be available on commercially reasonable terms.

A third party may hold intellectual property, including patent rights that are important or necessary to the development and commercialization of our products. It may be necessary for us to use the patented or proprietary technology of third parties to commercialize our products, in which case we would be required to obtain a license from these third parties, whom may or may not be interested in granting such a license, on commercially reasonable terms, or our business could be harmed, possibly materially. For example, we engage extensively with third parties, including academic institutions, to conduct non-clinical and clinical research on our product candidates. While we seek to ensure all material transfer and service agreements governing this research provide us with favorable terms covering newly generated intellectual property, a general principle under which much of this research with academic institutions is conducted provides third party ownership of newly generated intellectual property, with an exclusive option available for us to obtain a license to such intellectual property. Through the conduct of this research, it is possible that valuable intellectual property could be developed by a third party, which we will then need to license in order to better develop or commercialize our products. No assurance can be given that we will be able to successfully negotiate such a license on commercially reasonable terms, or at all. Further, should we fail to successfully negotiate a license to such intellectual property, most institutions are then free to license such intellectual property to any other third party, including potentially direct competitors of ours. Should we fail to adequately secure a license to any newly generated intellectual property, our ability to successfully develop or commercialize our products may be hindered, possibly materially.

We may be involved in lawsuits to protect or enforce our patents or the patents of our licensors, which could be expensive, time consuming and unsuccessful.

Competitors may infringe our patents or the patents of our licensors. To counter infringement or unauthorized use, we may be required to file infringement claims, which typically are very expensive, time-consuming and disruptive of day-to-day business operations. Any claims we assert against accused infringers could provoke these parties to assert counterclaims against us alleging invalidity of our or certain of our subsidiaries' patents or that we infringe their patents; or provoke those parties to petition the USPTO to institute inter parties review against the asserted patents, which may lead to a finding that all or some of the claims of the patent are invalid. In addition, in an infringement proceeding, a court may decide that a patent of ours or our licensors is not valid or is unenforceable, or may refuse to stop the other party from using the technology at issue on the grounds that our patents do not cover the technology in question. An adverse result in any litigation or defense proceedings could put one or more of our pending patents at risk of being invalidated, held unenforceable, or interpreted narrowly.

In patent litigation in the United States, defendant counterclaims challenging the validity, enforceability or scope of asserted patents are commonplace. In addition, third parties may initiate legal proceedings against us to assert such challenges to our intellectual property rights. The outcome of any such proceeding is generally unpredictable. Grounds for a validity challenge could be an alleged failure to meet any of several statutory requirements, including lack of novelty, obviousness or non-enablement. Patents may be unenforceable if someone connected with prosecution of the patent withheld relevant information from the USPTO or made a misleading statement during prosecution. It is possible that prior art of which we and the patent examiner were unaware during prosecution exists, which could render our patents invalid. Moreover, it is also possible that prior art may exist that we are aware of but do not believe is relevant to our current or future patents, but that could nevertheless be determined to render our patents invalid.

Competing drugs may also be sold in other countries in which our patent coverage might not exist or be as strong. If we lose a foreign patent lawsuit, alleging our infringement of a competitor's patents, we could be prevented from marketing our drugs in one or more foreign countries. Any of these outcomes would have a materially adverse effect on our business.

In addition, because of the substantial amount of discovery required in connection with intellectual property litigation, there is a risk that some of our confidential information could be compromised by disclosure during this type of litigation. Furthermore, adverse results on United States patents may affect related patents in our global portfolio. The adverse result could also put related pending patent applications at risk of not issuing. Additionally, there could be public announcements of the results of hearings, motions or other interim proceedings or developments. If securities analysts or investors perceive these results to be negative, it could have a substantial adverse effect on the price of our common stock.

Interference proceedings provoked by third parties or brought by the USPTO may be necessary to determine the priority of inventions with respect to our patents or pending patent applications or those of our collaborators or licensors. An unfavorable outcome could require us to cease using the related technology or to attempt to license rights to it from the prevailing party. The costs of these proceedings could be substantial. As a result, the issuance, scope, validity, enforceability and commercial value of our or any of our respective licensors' patent rights are highly uncertain. Our business could be harmed if the prevailing party does not offer us a license on commercially reasonable terms. Litigation or interference proceedings may fail and, even if successful, may result in substantial costs and distract our management and other employees. We may not be able to prevent, alone or with our licensors, misappropriation of our trade secrets or confidential information, particularly in countries where the laws may not protect those rights as fully as in the United States.

We may not have sufficient financial or other resources to adequately conduct such litigation or proceedings. Some of our competitors may be able to sustain the costs of such litigation or proceedings more effectively than we can because of their greater financial resources and more mature and developed intellectual property portfolios. Accordingly, despite our efforts, we may not be able to prevent third parties from infringing upon or misappropriating or from successfully challenging our intellectual property rights. Uncertainties resulting from the initiation and continuation of patent litigation or other proceedings could have a material adverse effect on our ability to compete in the marketplace.

If we are unable to protect the confidentiality of our trade secrets, our business and competitive position may be harmed.

In addition to the protection afforded by patents, we rely upon unpatented trade secret protection, unpatented know-how and continuing technological innovation to develop and maintain our competitive position. With respect to the building of our proprietary compound library, we consider trade secrets and know-how to be our primary intellectual property. We seek to protect our proprietary technology and processes, in part, by entering into confidentiality agreements with our collaborators, scientific advisors, employees and consultants, and invention assignment agreements with our consultants and employees. We may not be able to prevent the unauthorized disclosure or use of our technical know-how or other trade secrets by the parties to these agreements, however, despite the existence generally of confidentiality agreements and other contractual restrictions. Monitoring unauthorized uses and disclosures is difficult, and we do not know whether the steps we have taken to protect our proprietary technologies will be effective. If any of the collaborators, scientific advisors, employees and consultants who are parties to these agreements breaches or violates the terms of any of these agreements, we may not have adequate remedies for any such breach or violation, and we could lose our trade secrets as a result. Enforcing a claim that a third party illegally obtained and is using our trade secrets, like patent litigation, is expensive and time-consuming, and the outcome is unpredictable. In addition, courts outside the United States are sometimes less willing to protect trade secrets.

Our trade secrets could otherwise become known or be independently discovered by our competitors. Competitors could purchase our drug candidates and attempt to replicate some or all of the competitive advantages we derive from our development efforts, willfully infringe our intellectual property rights, design around our protected technology or develop their own competitive technologies that fall outside of our intellectual property rights. If any of our trade secrets were to be lawfully obtained or independently developed by a competitor, we would have no right to prevent them, or those to whom they communicate it, from using that technology or information to compete with us. If our trade secrets are not adequately protected so as to protect our market against competitors' drugs, our competitive position could be adversely affected, as could our business.

We may be subject to damages resulting from claims that we or our employees have wrongfully used or disclosed alleged trade secrets of our competitors or are in breach of non-competition or non-solicitation agreements with our competitors.

We could in the future be subject to claims that we or our employees have inadvertently or otherwise used or disclosed alleged trade secrets or other proprietary information of former employers or competitors. Although we try to ensure that our employees and consultants do not use the intellectual property, proprietary information, know-how or trade secrets of others in their work for us, we may in the future be subject to claims that we caused an employee to breach the terms of his or her non-competition or non-solicitation agreement, or that we or these individuals have, inadvertently or otherwise, used or disclosed the alleged trade secrets or other proprietary information of a former employer or competitor. Litigation may be necessary to defend against these claims. Even if we are successful in defending against these claims, litigation could result in substantial costs and could be a distraction to management. If our defenses to these claims fail, in addition to requiring us to pay monetary damages, a court could prohibit us from using technologies or features that are essential to our drug candidates, if such technologies or features are found to incorporate or be derived from the trade secrets or other proprietary information of the former employers. An inability to incorporate such technologies or features would have a material adverse effect on our business, and may prevent us from successfully commercializing our drug candidates. In addition, we may lose valuable intellectual property rights or personnel as a result of such claims. Moreover, any such litigation or the threat thereof may adversely affect our ability to hire employees or contract with independent sales representatives. A loss of key personnel or their work product could hamper or prevent our ability to commercialize our drug candidates, which would have an adverse effect on our business, results of operations and financial condition.

Risks Related to Employee Matters, Managing Growth and Other Risks Related to Our Business

If we fail to attract and keep key management and clinical development personnel, we may be unable to successfully develop or commercialize our product candidates.

We are highly dependent on the research and development, clinical, business development, financial and legal expertise of our senior management team as well as the other principal members of our management, scientific and clinical team. Although we have entered into employment agreement with our chief executive officer and employment letters with our senior managers, each of our executive officers may terminate their employment with us at any time. We do not maintain "key person" insurance for any of our executives or other employees. In addition, we rely on consultants and advisors, including scientific and clinical advisors, to assist us in formulating our research and development and commercialization strategy. Our consultants and advisors may be employed by employers other than us and may have commitments under consulting or advisory contracts with other entities that may limit their availability to us. If we are unable to continue to attract and retain high quality personnel, our ability to pursue our growth strategy will be limited.

We expect to continue hiring qualified development personnel. Recruiting and retaining qualified scientific, clinical, manufacturing and sales and marketing personnel will be critical to our success. The loss of the services of our chief executive officer or other key employees could impede the achievement of our research, development and commercialization objectives and seriously harm our ability to successfully implement our business strategy. Furthermore, replacing key employees may be difficult and may take an extended period of time because of the limited number of individuals in our industry with the breadth of skills and experience required to successfully develop, gain regulatory approval of and commercialize drugs. Competition to hire from this limited pool is intense, and we may be unable to hire, train, retain or motivate these key personnel on acceptable terms given the competition among numerous pharmaceutical and biotechnology companies for similar personnel. Failure to succeed in clinical trials may make it more challenging to recruit and retain qualified medical and scientific personnel. If we are not able to attract and retain the necessary personnel to accomplish our business objectives, we may experience constraints that will impede significantly the achievement of our development objectives, our ability to raise additional capital, and our ability to implement our business strategy.

We will need to develop and expand our Company, and we may encounter difficulties in managing this development and expansion, which could disrupt our operations.

As of August 2, 2019, we had 109 full-time employees, and we expect to continue to increase our number of employees and expand the scope of our operations. Our management and medical and scientific personnel, systems and facilities currently in place may not be adequate to support our future growth. To manage our anticipated future growth, we must continue to implement and improve our managerial, operational and financial systems, expand our facilities and continue to recruit and train additional qualified personnel. Also, our management may need to divert a disproportionate amount of its attention away from its day-to-day activities and devote a substantial amount of time to managing these development activities. Due to our limited resources, we may not be able to effectively manage the expansion of our operations or recruit and train additional qualified personnel. This may result in weaknesses in our infrastructure, give rise to operational mistakes, loss of business opportunities, loss of employees and reduced productivity among remaining employees. To accommodate growth, additional physical expansion of our operations in the future may lead to significant costs, including capital expenditures, and may divert financial resources from other projects, such as the development of our drug candidates. If our management is unable to effectively manage our expected development and expansion, our expenses may increase more than expected, our ability to generate or increase our revenue could be reduced and we may not be able to implement our business strategy. Our future financial performance and our ability to commercialize our drug candidates, if approved, and compete effectively will depend, in part, on our ability to effectively manage the future development and expansion of our company.

Additionally, to help manage the expanding needs, we may utilize the services of outside vendors or consultants to perform tasks including clinical trial management, statistics and analysis, regulatory affairs, formulation development, chemistry, manufacturing, controls, and other pharmaceutical development functions. Our growth strategy may also entail expanding our group of contractors or consultants to implement these tasks going forward. Because we rely on a substantial number of consultants, effectively outsourcing many key functions of our business, we will need to be able to effectively manage these consultants to ensure that they successfully carry out their contractual obligations and meet expected deadlines. However, if we are unable to effectively manage our outsourced activities or if the quality or accuracy of the services provided by consultants is compromised for any reason, our clinical trials may be extended, delayed or terminated, and we may not be able to obtain regulatory approval for our product candidates or otherwise advance its business. There can be no assurance that we will be able to manage our existing consultants or find other competent outside contractors and consultants on economically reasonable terms, or at all. If we are not able to effectively expand our organization by hiring new employees and expanding our groups of consultants and contractors, we may be unable to successfully implement the tasks necessary to further develop and commercialize our product candidates and, accordingly, may not achieve our research, development and commercialization goals.

Unfavorable global economic conditions could adversely affect our business, financial condition or results of operations.

Our results of operations could be adversely affected by general conditions in the global economy and in the global financial markets. For example, the global financial crisis caused extreme volatility and disruptions in the capital and credit markets. A severe or prolonged economic downturn, such as the global financial crisis, could result in a variety of risks to our business, including, weakened demand for our drug candidates and our ability to raise additional capital when needed on acceptable terms, if at all. A weak or declining economy could also strain our suppliers, possibly resulting in supply disruption, or cause our customers to delay making payments for our services.

Following its June 23, 2016 vote to leave the European Union, on March 29, 2017, the United Kingdom invoked Article 50 of the Lisbon Treaty and formally began the process of exiting the European Union. Although Brexit has already and may continue to adversely affect European and/or worldwide economic or market, political or regulatory conditions and may contribute to instability in the global financial markets, political institutions and regulatory agencies, the resulting immediate changes in foreign currency exchange rates have had a limited overall impact due to natural hedging. The long-term impact of Brexit, including on our business and our industry, will depend on the terms that are negotiated in relation to the United Kingdom's future relationship with the European Union, and we are closely monitoring the Brexit developments in order to determine, quantify and proactively address changes as they become clear. Despite the Brexit developments, we do not expect macroeconomic conditions to have a significant impact on our liquidity needs, financial condition or results of operations.

We or the third parties upon whom we depend may be adversely affected by earthquakes or other natural disasters and our business continuity and disaster recovery plans may not adequately protect us from a serious disaster.

Earthquakes or other natural disasters could severely disrupt our operations, and have a material adverse effect on our business, results of operations, financial condition and prospects. If a natural disaster, power outage or other event occurred that prevented us from using all or a significant portion of our headquarters, that damaged critical infrastructure, such as the manufacturing facilities of our third-party contract manufacturers, or that otherwise disrupted operations, it may be difficult or, in certain cases, impossible for us to continue our business for a substantial period of time. The disaster recovery and business continuity plans we have in place may prove inadequate in the event of a serious disaster or similar event. We may incur substantial expenses as a result of the limited nature of our disaster recovery and business continuity plans, which, could have a material adverse effect on our business.

Our internal computer systems, or those of our third-party CROs or other contractors or consultants, may fail or suffer security breaches, which could result in a material disruption of our drug candidates' development programs.

Despite the implementation of security measures, our internal computer systems and those of our third-party CROs and other contractors and consultants are vulnerable to damage from computer viruses, unauthorized access, natural disasters, terrorism, war and telecommunication and electrical failures. While we have not experienced any such system failure, accident, or security breach to date, if such an event were to occur and cause interruptions in our operations, it could result in a material disruption of our programs. For example, the loss of clinical trial data for our drug candidates could result in delays in our regulatory approval efforts and significantly increase our costs to recover or reproduce the data. To the extent that any disruption or security breach results in a loss of or damage to our data or applications or other data or applications relating to our technology or drug candidates, or inappropriate disclosure of confidential or proprietary information, we could incur liabilities and the further development of our drug candidates could be delayed.

Our employees, principal investigators, CROs and consultants may engage in misconduct or other improper activities, including non-compliance with regulatory standards and requirements and insider trading, which could have a material adverse effect on our business.

We are exposed to the risk that our employees, principal investigators, CROs and consultants may engage in fraudulent conduct or other illegal activity. Misconduct by these parties could include intentional failures to comply with FDA regulations, provide accurate information to the FDA, comply with manufacturing standards we have established, comply with federal and state healthcare fraud and abuse laws and regulations, report financial information or data accurately or disclose unauthorized activities to us. In particular, sales, marketing and business arrangements in the healthcare industry are subject to extensive laws and regulations intended to prevent fraud, misconduct, kickbacks, self-dealing and other abusive practices. These laws and regulations may restrict or prohibit a wide range of pricing, discounting, marketing and promotion, sales commission, customer incentive programs and other business arrangements. Activities subject to these laws also involve the improper use of information obtained in the course of clinical trials or creating fraudulent data in our pre-clinical studies or clinical trials, which could result in regulatory sanctions and cause serious harm to our reputation. We have adopted a code of ethics applicable to all of our employees, but it is not always possible to identify and deter misconduct by employees and other third parties, and the precautions we take to detect and prevent this activity may not be effective in controlling unknown or unmanaged risks or losses or in protecting us from governmental investigations or other actions or lawsuits stemming from a failure to comply with these laws or regulations. In addition, we are subject to the risk that a person could allege such fraud or other misconduct, even if none occurred. If any such actions are instituted against us, and we are not successful in defending ourselves or asserting our rights, those actions could have a significant impact on our business, including the imposition of civil, criminal and administrative penalties, damages, monetary fines, possible exclusion from participation in Medicare, Medicaid and other federal healthcare programs, contractual damages, reputational harm, diminished profits and future earnings, and curtailment of our operations, any of which could adversely affect our ability to operate our business and our results of operations.

We may acquire businesses or drugs, or form strategic alliances, in the future, and we may not realize the benefits of such acquisitions.

We may acquire additional businesses or drugs, form strategic alliances or create joint ventures with third parties that we believe will complement or augment our existing business. If we acquire businesses with promising markets or technologies, we may not be able to realize the benefit of acquiring such businesses if we are unable to successfully integrate them with our existing operations and company culture. We may encounter numerous difficulties in developing, manufacturing and marketing any new drugs resulting from a strategic alliance or acquisition that delay or prevent us from realizing their expected benefits or enhancing our business. We cannot assure you that, following any such acquisition, we will achieve the expected synergies to justify the transaction.

We may be subject to adverse legislative or regulatory tax changes that could negatively impact our financial condition.

The rules dealing with U.S. federal, state and local income taxation are constantly under review by persons involved in the legislative process and by the IRS and the U.S. Treasury Department. Changes to tax laws (which changes may have retroactive application) could adversely affect our stockholders or us. In recent years, many such changes have been made and changes are likely to continue to occur in the future. We cannot predict whether, when, in what form, or with what effective dates, tax laws, regulations and rulings may be enacted, promulgated or decided, which could result in an increase in our, or our stockholders', tax liability or require changes in the manner in which we operate in order to minimize increases in our tax liability.

On December 22, 2017, the Tax Cuts and Jobs Act, or TCJA, was enacted. The TCJA significantly reforms the Internal Revenue Code of 1986, as amended. The TCJA, among other things, includes changes to U.S. federal tax rates, imposes significant additional limitations on the deductibility of interest and net operating loss carryforwards and allows for the expensing of capital expenditures. Our net deferred tax assets and liabilities were revalued as of December 31, 2017 at the newly enacted U.S. corporate rate, and the impact was recognized in our tax expense in the year of enactment but was offset by a corresponding reduction to the valuation allowance. We continue to examine the impact this tax reform legislation may have on our business. The impact of this tax reform is uncertain and could be adverse.

Our tax position could be affected by recent changes in United States federal income tax laws.

On December 22, 2017, legislation commonly referred to as the "Tax Cuts and Jobs Act" was signed into law and is generally effective after December 31, 2017. The Tax Cuts and Jobs Act makes significant changes to the United States federal income tax rules for taxation of individuals and business entities. Most of the changes applicable to individuals are temporary and apply only to taxable years beginning after December 31, 2017 and before January 1, 2026. For corporations, the Tax Cuts and Jobs Act reduces the top corporate income tax rate to 21% and repeals the corporate alternative minimum tax, limits the deduction for net interest expense, limits the deduction for net operating losses and eliminates net operating loss carrybacks, modifies or repeals many business deductions and credits, shifts the United States toward a more territorial tax system, and imposes new taxes to combat erosion of the United States federal income tax base. The Tax Cuts and Jobs Act makes numerous other large and small changes to the federal income tax rules that may affect potential investors and may directly or indirectly affect us. We continue to examine the impact this tax reform legislation may have on our business. However, the effect of the Tax Cuts and Jobs Act on us, whether adverse or favorable, is uncertain, and may not become evident for some period of time. This document does not discuss such legislation or the manner in which it might affect us or purchasers of our common stock. Prospective investors are urged to consult with their legal and tax advisors with respect to the Tax Cuts and Jobs Act and any other regulatory or administrative developments and proposals, and their potential effects on them based on their unique circumstances.

Risks Related to Our Common Stock and Being a Publicly-Traded Company

Our stock price is, and we expect it to remain, volatile, which could limit investors' ability to sell stock at a profit.

The trading price of our common stock is likely to be highly volatile and subject to wide fluctuations in price in response to various factors, many of which are beyond our control. These factors include:

- publicity regarding actual or potential clinical results relating to products under development by our competitors or us;
- delay or failure in initiating, completing or analyzing nonclinical or clinical trials or the unsatisfactory design or results of these trials;
- achievement or rejection of regulatory approvals by our competitors or us;
- announcements of technological innovations or new commercial products by our competitors or us;
- developments concerning proprietary rights, including patents;
- developments concerning our collaborations;
- regulatory developments in the United States and foreign countries;
- economic or other crises and other external factors;
- period-to-period fluctuations in our revenues and other results of operations;
- changes in financial estimates by securities analysts; and
- sales of our common stock by us.

We will not be able to control many of these factors, and we believe that period-to-period comparisons of our financial results will not necessarily be indicative of our future performance.

In addition, the stock market in general, and the market for biotechnology companies in particular, has experienced extreme price and volume fluctuations that may have been unrelated or disproportionate to the operating performance of individual companies. These broad market and industry factors may seriously harm the market price of our common stock, regardless of our operating performance.

Our executive officers, directors, principal stockholders and their affiliates maintain the ability to exercise significant influence over our company and all matters submitted to stockholders for approval.

Our executive officers, directors and stockholders who own more than 5% of our outstanding common stock, together with their affiliates and related persons, beneficially own shares of common stock representing a significant percentage of our capital stock. As a result, if these stockholders were to choose to act together, they would be able to influence our management and affairs and the outcome of matters submitted to our stockholders for approval, including the election of directors and any sale, merger, consolidation, or sale of all or substantially all of our assets. This concentration of voting power could delay or prevent an acquisition of our company on terms that other stockholders may desire. In addition, this concentration of ownership might adversely affect the market price of our common stock by:

- delaying, deferring or preventing a change of control of us;
- impeding a merger, consolidation, takeover or other business combination involving us; or
- discouraging a potential acquirer from making a tender offer or otherwise attempting to obtain control of us.

Because we do not anticipate paying any cash dividends on our capital stock in the foreseeable future, capital appreciation, if any, will be the sole source of gain for our stockholders.

We have never declared or paid cash dividends on our capital stock. We currently intend to retain all of our future earnings, if any, to finance the growth and development of our business. In addition, under the Loan Agreement, we are currently restricted from paying cash dividends, and we expect these restrictions to continue in the future. In addition, the terms of any future debt agreements may preclude us from paying dividends. As a result, capital appreciation, if any, of our common stock will be the sole source of gain for our stockholders for the foreseeable future.

Certain anti-takeover provisions in our charter documents and Delaware law could make a third-party acquisition of us difficult. This could limit the price investors might be willing to pay in the future for our common stock.

Provisions in our amended and restated certificate of incorporation and restated bylaws could have the effect of making it more difficult for a third party to acquire, or of discouraging a third party from attempting to acquire, or control us. These factors could limit the price that certain investors might be willing to pay in the future for shares of our common stock. Our amended and restated certificate of incorporation allows us to issue preferred stock without the approval of our stockholders. The issuance of preferred stock could decrease the amount of earnings and assets available for distribution to the holders of our common stock or could adversely affect the rights and powers, including voting rights, of such holders. In certain circumstances, such issuance could have the effect of decreasing the market price of our common stock. Our restated bylaws eliminate the right of stockholders to call a special meeting of stockholders, which could make it more difficult for stockholders to effect certain corporate actions. Any of these provisions could also have the effect of delaying or preventing a change in control.

On July 18, 2014, the Board of Directors declared a distribution of one right for each outstanding share of common stock. The rights may have certain anti-takeover effects. The rights will cause substantial dilution to a person or group that attempts to acquire us on terms not approved by the Board of Directors unless the offer is conditioned on a substantial number of rights being acquired. However, the rights should not interfere with any merger, statutory share exchange or other business combination approved by the Board of Directors since the rights may be terminated by us upon resolution of the Board of Directors. Thus, the rights are intended to encourage persons who may seek to acquire control of the Company to initiate such an acquisition through negotiations with the Board of Directors. However, the effect of the rights may be to discourage a third party from making a partial tender offer or otherwise attempting to obtain a substantial equity position in the equity securities of, or seeking to obtain control of, the Company. To the extent any potential acquirers are deterred by the rights, the rights may have the effect of preserving incumbent management in office.

An active trading market for our common stock may not be sustained, and investors may not be able to resell their shares at or above the price they paid.

Although we have listed our common stock on The Nasdaq Capital Market, an active trading market for our shares may not be sustained. In the absence of an active trading market for our common stock, investors may not be able to sell their common stock at or above the price at which they acquired their shares or at the time that they would like to sell. An inactive trading market may also impair our ability to raise capital to continue to fund operations by selling shares and may impair our ability to acquire other companies or technologies by using our shares as consideration.

If equity research analysts do not publish research or reports about our business or if they publish negative evaluations of or downgrade our common stock, the price of our common stock could decline.

The trading market for our common stock relies in part on the research and reports that equity research analysts publish about us or our business. We do not control these analysts. We may never obtain research coverage by industry or financial analysts. If no or few analysts commence coverage of us, the trading price of our stock would likely decrease. Even if we do obtain analyst coverage, if one or more of the analysts covering our business downgrade their evaluations of our common stock, the price of our common stock could decline. If one or more of these analysts cease to cover our common stock, we could lose visibility in the market for our common stock, which in turn could cause our common stock price to decline.

We incur significant increased costs as a result of operating as a public company, and our management is required to devote substantial time to compliance initiatives.

As a public company, we incur significant legal, accounting and other expenses under the Sarbanes-Oxley Act of 2002, as well as rules subsequently implemented by the Securities and Exchange Commission ("SEC"), and the rules of any stock exchange on which we may become listed. These rules impose various requirements on public companies, including requiring establishment and maintenance of effective disclosure and financial controls and appropriate corporate governance practices. Our team has devoted and will continue to devote a substantial amount of time to these compliance initiatives. Moreover, these rules and regulations increase our legal and financial compliance costs and make some activities more time-consuming and costly. For example, these rules and regulations make it more difficult and more expensive for us to obtain director and officer liability insurance, and we may be required to accept reduced policy limits and coverage or incur substantially higher costs to obtain the same or similar coverage. As a result, it may be more difficult for us to attract and retain qualified persons to serve on our Board of Directors, our Board committees or as executive officers.

The Sarbanes-Oxley Act of 2002 requires, among other things, that we maintain effective internal control over financial reporting and disclosure controls and procedures. As a result, we are required to periodically perform an evaluation of our internal control over financial reporting to allow management to report on the effectiveness of those controls, as required by Section 404 of the Sarbanes-Oxley Act. Additionally, our independent auditors are required to perform a similar evaluation and report on the effectiveness of our internal control over financial reporting. These efforts to comply with Section 404 will require the commitment of significant financial and managerial resources. While we anticipate maintaining the integrity of our internal control over financial reporting and all other aspects of Section 404, we cannot be certain that a material weakness will not be identified when we test the effectiveness of our control systems in the future. If a material weakness is identified, we could be subject to sanctions or investigations by the SEC or other regulatory authorities, which would require additional financial and management resources, costly litigation or a loss of public confidence in our internal control, which could have an adverse effect on the market price of our stock.

Volatility in the price of our common stock may subject us to securities litigation, which could cause us to incur substantial costs and divert management's attention, financial resources and other company assets.

In the past, securities class action litigation has often been brought against a company following periods of volatility in the market price of its securities. This risk is especially relevant for us because pharmaceutical companies have experienced significant stock price volatility in recent years. Past lawsuits and any future lawsuits to which we may become a party are subject to inherent uncertainties and will likely be expensive and time-consuming to investigate, defend and resolve, and will divert our management's attention and financial and other resources. The outcome of litigation is necessarily uncertain, and we could be forced to expend significant resources in the defense of these and other suits, and we may not prevail. Any litigation to which we are a party may result in an onerous or unfavorable judgment that may not be reversed upon appeal or in payments of substantial monetary damages or fines, or we may decide to settle this or other lawsuits on similarly unfavorable terms, which could adversely affect our business, financial condition, results of operations or stock price. See Part II. Item I. Legal Proceedings above for additional information.

Future sales of our common stock, including by us or our directors and executive officers or shares issued upon the exercise of currently outstanding options, could cause our stock price to decline.

A substantial portion of our outstanding common stock can be traded without restriction at any time. In addition, a portion of our outstanding common stock is currently restricted as a result of federal securities laws, but can be sold at any time subject to applicable volume limitations. As such, sales of a substantial number of shares of our common stock in the public market could occur at any time. These sales, or the perception in the market that the holders of a large number of shares intend to sell shares, by us or others, could reduce the market price of our common stock or impair our ability to raise adequate capital through the sale of additional equity securities. In addition, we have a significant number of shares that are subject to outstanding options. The exercise of these options and the subsequent sale of the underlying common stock could cause a further decline in our stock price. These sales also might make it difficult for us to sell equity securities in the future at a time and at a price that we deem appropriate. We cannot predict the number, timing or size of future issuances or the effect, if any, that any future issuances may have on the market price for our common stock.

Our ability to utilize our net operating loss carryforwards and certain other tax attributes may be limited.

Under Section 382 of the Internal Revenue Code of 1986, as amended, if a corporation undergoes an "ownership change" (generally defined as a greater than 50% change (by value) in the ownership of its equity over a three year period), the corporation's ability to use its pre-change net operating loss carryforwards and certain other pre-change tax attributes to offset its post-change income may be limited. We may have experienced such ownership changes in the past, and we may experience ownership changes in the future as a result of shifts in our stock ownership, some of which are outside our control. As of December 31, 2018, we had federal net operating loss carryforwards of approximately \$550.6 million, and our ability to utilize those net operating loss carryforwards could be limited by an "ownership change" as described above, which could result in increased tax liability to us. In addition, pursuant to the TCJA, we may not use net operating loss carry-forwards to reduce our taxable income in any year by more than 80%, and we may not carry back any net operating losses to prior years. These new rules apply regardless of the occurrence of an ownership change.

ITEM 6. EXHIBITS

The exhibits listed on the Exhibit Index are included with this report.

- 10.1 Amended and Restated Collaboration Agreement by and between TG Therapeutics, Inc. and Checkpoint Therapeutics, Inc., dated June 19, 2019.*
- 10.2 Master Services Agreement between Samsung Biologics Co., Ltd. and TG Therapeutics, Inc., effective February 21, 2018.
- 31.1 Certification of Chief Executive Officer pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, dated August 9, 2019.
- 31.2 Certification of Chief Financial Officer pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, dated August 9, 2019.
- 32.1 Certification of Chief Executive Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, dated August 9, 2019.
- 32.2 Certification of Chief Financial Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, dated August 9, 2019.
- The following financial information from the Company's Quarterly Report on Form 10-Q for the period ended June 30, 2019, formatted in Inline Extensible Business Reporting Language (iXBRL): (i) the Condensed Consolidated Balance Sheets, (ii) the Condensed Consolidated Statements of Operations, (iii) the Condensed Consolidated Statements of Stockholders' Equity, (iv) the Condensed Consolidated Statements of Cash Flows, and (v) Notes to the Condensed Consolidated Financial Statements (filed herewith).

*Certain portions of this exhibit have been omitted pursuant to Item 601(b)(10) of Regulation S-K.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

TG THERAPEUTICS, INC.

Date: August 9, 2019

By: /s/ Sean A. Power

Sean A. Power Chief Financial Officer

Principal Financial and Accounting Officer

AMENDED AND RESTATED COLLABORATION AGREEMENT

THIS AMENDED AND RESTATED COLLABORATION AGREEMENT (the "Agreement") is dated as of June 19, 2019 by and between Checkpoint Therapeutics, Inc., a Delaware corporation organized having its place of business at 2 Gansevoort Street, New York, NY 10014 ("CTI"), and TG Therapeutics, Inc. located at 2 Gansevoort Street, New York, NY 10014 ("TGTX"). CTI, on the one hand, and TGTX, on the other hand, shall each be referred to herein as a "Party" or, collectively, as the "Parties."

RECITALS:

WHEREAS, CTI is party to that certain license agreement (the "License Agreement") dated March 3, 2015 with Dana Farber Cancer Institute ("DFCI");

WHEREAS, DFCI is the owner of certain rights in the DFCI Technology; and

WHEREAS, DFCI has licensed rights to the DFCI Technology to CTI; and

WHEREAS, CTI is permitted to extend the rights granted to it under the DFCI Technology to Affiliates (as defined in the License Agreement); and

WHEREAS, CTI has developed and licensed the rights to Additional PD-L1 Intellectual Property, including rights under CTI Patents (as defined, below) and that certain collaboration agreement ("Collaboration Agreement") dated January 22, 2019 with Adimab LLC ("Adimab"), a true and correct copy of which is attached hereto as Exhibit A; and

WHEREAS, TGTX, an Affiliate of CTI, is engaged in the research, development, manufacturing and commercialization of pharmaceutical products, and TGTX is interested in developing and commercializing products based on the DFCI Patents and Additional PD-L1 Intellectual Property (together, the "Licensed Technology"); and

WHEREAS, the Parties signed a Collaboration Agreement (the "Original Agreement") on March 3, 2015 (the "Effective Date"), and the Parties wish to amend such Original Agreement as provided herein; and

WHEREAS, CTI desires to continue to collaborate with TGTX and extend to TGTX the rights granted to it under the Licensed Technology in order to benefit the public by disseminating the results of its research via the commercial development, manufacture, distribution and use of Licensed Products (as defined below); and

WHEREAS, TGTX desires to continue to collaborate with CTI and to exercise the rights granted to CTI, on an exclusive basis, so that it can exclusively use, develop and commercialize the Licensed Technology in and for a defined field of use; and

WHEREAS, in the event TGTX is no longer an Affiliate of CTI, TGTX and CTI intend for the rights extended to TGTX hereunder to continue as a Sublicense (as defined in the License Agreement) as permitted by Section 2.3 of the License Agreement and Section 3.2(b)(ii) of the Collaboration Agreement with Adimab.

NOW, THEREFORE, in consideration of the foregoing and of the various promises and undertakings set forth herein, the Parties agree as follows:

ARTICLE I DEFINITIONS

Unless otherwise specifically provided herein, the following terms shall have the following meanings:

- 1.1 "Affiliate" means a Person or entity that controls, is controlled by or is under common control with a Party, but only for so long as such control exists. For the purposes of this Section 1.1, the word "control" (including, with correlative meaning, the terms "controlled by" or "under common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct the management and policies of such Person or entity, whether by the ownership of at least 50% of the voting stock of such entity, or by contract or otherwise. TGTX and CTI acknowledge and agree that TGTX is an Affiliate of CTI.
- 1.2 "Antibody" means any antibody, any gene expressing such an antibody, any hybridoma producing such antibody, or any fragment, variant, derivative or construct thereof, or antibody fusion protein produced therefrom (including PEDgylated or multimeric versions thereof, whether polyclonal, monoclonal, multi-specific antibodies (e.g., bi-specific antibodies), human, humanized, chimeric, murine, synthetic, or from any other source), including without limitation (a) the full immunoglobin molecules (e.g., the IgG, IgM, IgE, IgA, and IgD molecules), and (b) the antigen binding portions including Fab, Fab', F(ab')2, Fv, dAb, and CDR fragments, chimeric antibodies, diabodies, polypeptides, linear antibodies and single-chain antibodies (scFv) that contain any portion of an immunoglobulin that is sufficient to confer specific binding to an antigen.
- 1.3 "Autoimmune Diseases" means any disease which results from a loss of immune tolerance to self-antigens, including without limitation multiple sclerosis, rheumatoid arthritis, systemic lupus erythematosus, sjogren syndrome, celiac disease, Graves' disease, myasthenia gravis, Type I diabetes, idiopathic thrombocytopenic purpura, pemphigus vulgaris, among others, including any presentation or manifestation thereof.
- 1.4 "Calendar Quarter" means each three-month period commencing January 1, April 1, July 1 or October 1, provided however that (a) the first Calendar Quarter of the Term shall extend from the Effective Date to the end of the first full Calendar Quarter thereafter, and (b) the last Calendar Quarter of the Term shall end upon the termination or expiration of this Agreement.
- 1.5 "Calendar Year" means the period beginning on the 1st of January and ending on the 31st of December of the same year, provided however that (a) the first Calendar Year of the Term shall commence on the Effective Date and end on December 31 of the same calendar year as the Effective Date, and (b) the last Calendar Year of the Term shall commence on January 1 of the Calendar Year in which this Agreement terminates or expires and end on the date of termination or expiration of this Agreement.
- 1.6 "Combination Product" means a product (a) containing a Licensed Product together with one or more other active ingredients, or (b) with one or more products, devices, pieces of equipment or components, but sold for an integrated price (e.g., with the purchase of one product the customer gets a coupon for the other) or for a single price.
- 1.7 "Commercialization" or "Commercialize" means any and all activities undertaken at any time for a particular Licensed Product and that relate to the manufacturing, marketing, promoting, distributing, importing or exporting for sale, offering for sale, and selling of the Licensed Product, and interacting with Regulatory Authorities regarding the foregoing.
- 1.8 "Commercially Reasonable Efforts" means, with respect to the efforts to be expended by a Party or such Party's applicable Affiliate with respect to any objective, such reasonable, diligent, and good faith efforts normally used to accomplish a similar objective under similar circumstances by a similarly-situated company. Commercially Reasonable Efforts will not mean that a Party commits that it or such Party's applicable Affiliate will actually accomplish the applicable task.
- 1.9 "Controlled" means, with respect to (a) DFCI Patents, (b) Know-How, (c) Antibodies, (d) DFCI Materials, or (e) Additional PD-L1 Intellectual Property, that a Party or one of its Affiliates owns or has a license or sublicense to such DFCI Patents, Know-How, Antibodies, DFCI Material (or in the case of DFCI Material, has the right to physical possession of such material), or Additional PD-L1 Intellectual Property, and has the ability to grant a license or sublicense to, or assign its right, title and interest in and to, such DFCI Patents, Know-How, Antibodies, DFCI Material, or Additional PD-L1 Intellectual Property as provided for in this Agreement without violating the terms of any agreement or other arrangement with any Third Party.
- 1.10 "Covered" means, with respect to a Licensed Product, that the practicing, manufacturing, importing, using, selling, or offering for sale of such Licensed Product would, but for ownership of or a license granted hereunder under Additional PD-L1 Intellectual Property or DFCI's relevant DFCI Patents, infringe a Valid Claim of Additional PD-L1 Intellectual Property or DFCI's relevant DFCI Patents in the country in which the activity occurs (or, in the case of a Valid Claim that has not yet issued, would infringe such Valid Claim if it were to issue).

- 1.11 "Derivative" means a DFCI Antibody that has (a) been modified via isotype switching; (b) undergone a modification of effector function; (c) been adapted to enable the antibody to carry payloads; (d) been altered to change the expression characteristics, stability or biological half-life of the antibody; or (e) been mutated using an affinity maturation strategy designed to modify the affinity of either the variable regions and/or the constant regions of the antibody for any ligands, antigens or receptors. Derivatives may be full length antibodies, monoclonal and polyclonal antibodies, multispecific antibodies (e.g., bi-specific antibodies) and antibody fragments (including Fab, Fab', F(ab')2, Fy fragments, diabodies, linear antibodies and single-chain antibodies), in each case, of any origin, whether human, humanized, chimeric or otherwise.
- 1.12 "Development" or "Develop" means, with respect to a Licensed Product, the performance of all preclinical and clinical development (including, without limitation, toxicology, pharmacology, test method development and stability testing, process development, formulation development, quality control development, statistical analysis), clinical trials, and manufacturing and regulatory activities that are required to obtain Regulatory Approval of such Licensed Product.
 - 1.13 DFCI Antibodies" means the Antibodies supplied by or on behalf of DFCI to CTI under this Agreement as identified in Schedule 4.
- 1.14 "DFCI Know-How" means any and all Know-How that (a) is Controlled by DFCI or any of its Affiliates as of the Effective Date and (b) was developed in the laboratory of Dr. Wayne Marasco in the performance of research directly pertaining to the DFCI Patents and (c) is necessary for CTI to research, Develop, manufacture, use, or Commercialize Licensed Products. The DFCI Know-How is described in Schedule 2 hereto.

1.15 [Reserved]

- 1.16 "DFCI Patents" means (a) those patents and patent applications set forth on Schedule 1 hereto; (b) any additions, divisionals, continuations, conversion, supplemental examinations, extensions, term restorations, registrations, reinstatements, amendments, reissuances, corrections, substitutions, reexaminations, registrations, revalidations, supplementary protection certificates, renewals, and foreign counterparts of the patents and patent applications mentioned in clause (a) above; (c) all patents issuing from any of the patents and patent applications mentioned in clause (a) or (b) above and any foreign counterparts of any such patents and patent applications, and which shall include, in any case, patents surviving post grant review and *inter partes* review.
 - 1.17 "DFCI Technology" means the DFCI Patents, DFCI Know-How, DFCI Antibodies, Derivatives, and DFCI Materials.
 - **1.18** "EMA" means the European Medicines Agency or any successor agency.
- **1.19 "European Commission"** means the authority within the European Union that has the legal authority to grant Regulatory Approvals in the European Union based on input received from the EMA or other competent Regulatory Authorities.
 - **1.20** "FDA" means the United States Food and Drug Administration, or a successor federal agency thereto.
- 1.21 "Field" means all prophylactic, palliative, therapeutic or diagnostic uses in humans or animals for the prevention, diagnosis and treatment of hematological malignancies, including, without limitation, all Leukemia's, Lymphoma's, Multiple Myeloma and Waldentroms Macroglobulemia, but specifically excluding use in chimeric antigen receptor technology. Additionally, upon exercise of the Autoimmune Option, the Field shall include the prevention, diagnosis and treatment of Autoimmune Diseases.
- 1.22 "First Commercial Sale" means, with respect to a Licensed Product in any country, the first commercial transfer or disposition for value of such Licensed Product in the Field in such country to a Third Party, by TGTX, by an Affiliate of TGTX or by a Sublicensee after Regulatory Approval therefor has been obtained in such country, for cash or non-cash consideration to which a fair market value can be assigned for purposes of determining Net Sales.
 - **1.23** "GAAP" means United States generally accepted accounting principles.
- 1.24 "Governmental Body" means any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi-governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, officer, official, representative, organization, unit, body or entity and any court or other tribunal); (d) multi-national or supranational organization or body; or (e) individual, entity, or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police, military or taxing authority or power of any nature.

- 1.25 "Know-How" means any scientific or technical information, results and data of any type whatsoever, in any intangible form whatsoever, that is not in the public domain or otherwise publicly known and is not claimed or disclosed in a patent or pending patent application, including practices, protocols, regulatory filings, scientific techniques, works of authorship, plans, data (including, but not limited to, pharmacological, biological, chemical, toxicological, clinical and analytical information, quality control, trial and stability data), data analyses, reports, studies and procedures, designs for experiments and tests and results of experimentation and testing (including results of research or development), summaries and information contained in submissions to and information from ethical committees, the FDA or other Regulatory Authorities, and manufacturing process and development information. The fact that an item is known to the public shall not be taken to exclude the possibility that a compilation including the item, and/or a development relating to the item, is (and remains) not known to the public. "Know-How" excludes Additional PD-L1 Intellectual Property, DFCI Patents, DFCI Antibodies, and DFCI Materials.
- 1.26 "Law" or "Laws" means all applicable laws, statutes, rules, regulations, ordinances and other pronouncements having the binding effect of law of any Governmental Body.
- 1.27 "Licensed Product" means any pharmaceutical product, in any dosage form, preparation, composition, formulation, presentation or package configuration, (a) that is Covered in whole or in part by a Valid Claim in the DFCI Patents or Additional PD-L1 Intellectual Property, (b) that incorporates, constitutes, or contains DFCI Antibodies or Derivatives as an active ingredient, or (c) that shares at least [*]% of the amino acid sequence identity (combined or in the aggregate) to all the complementarity determining regions (CDRs) of any DFCI Antibodies or Derivatives and made using DFCI Technology or Additional PD-L1 Intellectual Property.
- **1.28** "Licensed Process" means processes which, (a) in the course of being practiced, is Covered in whole or in part by a Valid Claim in the Additional PD-L1 Intellectual Property or DFCI Patents, or (b) which incorporates or uses DFCI Antibodies or Derivatives in whole or in part.
- 1.29 "NDA" means a New Drug Application submitted pursuant to the requirements of the FDA, as more fully defined in 21 U.S. CFR § 314.3 et seq., a Biologics License Application submitted pursuant to the requirements of the FDA, as more fully defined in 21 U.S. CFR § 601, and any equivalent application submitted in any country, including a European Marketing Authorization Application, together, in each case, with all additions, deletions or supplements thereto.
- 1.30 "NDA Approval" means the receipt of notice from the relevant US Regulatory Authority that an NDA for a Licensed Product has met all the criteria for marketing approval.
- 1.31 "Net Sales" means the gross income derived by TGTX or its Affiliates or Sublicensees to unrelated Third Parties for a Licensed Product in the Field in bona-fide arms-length transactions, less the following deductions, which may not exceed reasonable and customary amounts in the country in which the transaction occurs:
 - (a) Normal and customary trade, quantity, cash and discounts and credits allowed and taken;
 - (b) Discounts, refunds, rebates, chargebacks, retroactive price adjustments, and any other allowances given and taken which effectively reduce the net selling price, including, without limitation, Medicaid rebates, institutional rebates or volume discounts;
 - (c) Product returns and allowances;
 - (d) Administrative fees paid to group purchasing organizations (e.g., Medicare) and government-mandated rebates;
 - (e) Shipping, handling, freight, postage, insurance and transportation charges, but all only to the extent included as a separate line item in the gross amount invoiced;
 - (f) Any tax, tariff or duties imposed on the sale, delivery or use of the Licensed Product, including, without limitation, sales, use, excise or value added taxes and customs and duties, but all only to the extent included as a separate line item (e.g., "taxes") in the gross amount invoiced.
 - (g) Bad debt actually written off during the accounting period (provided, that any bad debt write-off so taken which is later reversed shall be added back to Net Sales in the accounting period in which the reversal occurs).

No deduction shall be made for any item of cost incurred by TGTX, its Affiliates or Sublicensees in Developing or Commercializing Licensed Products except as permitted pursuant to clauses (a) through (g) of the foregoing.

Net Sales includes the fair market value of any non-cash consideration from sale of Licensed Products received by TGTX, its Affiliates or Sublicenses. Licensed Products are considered "sold" when billed, invoiced, or payment is received, whichever occurs first.

Notwithstanding the foregoing, amounts invoiced by TGTX and its Affiliates and Sublicensees for sales of Licensed Products among TGTX and its Sublicensees and their respective Affiliates for resale shall not be included in the computation of Net Sales except where such purchasing party is an end user or consumer of Licensed Products.

Net Sales of any Combination Product (as defined below) for the purpose of calculating royalties due under this Agreement shall be determined on a country-by-country basis as follows: the Net Sales of the Combination Product (prior to application of the following adjustment) shall be multiplied by the fraction A/(A+B), where A is the net selling price in such country of a Licensed Product without the additional active ingredient in the Combination Product, if sold separately for the same dosage as contained in the Combination Product, and B is the net selling price in such country of any other active ingredients (or delivery device) in the combination if sold separately for the same dosage (or form) as contained in the Combination Product. All net selling prices of the elements of such end-user product or service shall be calculated as the average net selling price of the said elements during the applicable accounting period for which the Net Sales are being calculated. In the event that, in any country, no separate sale of either such above-designated Licensed Product (containing only such Licensed Product and no other active ingredients) or any one or more of the active ingredients included in such Combination Product are made during the accounting period in which the sale was made or if the net selling price for an active ingredient cannot be determined for an accounting period, Net Sales for purposes of determining payments under this Agreement shall be calculated by multiplying the sales price of the Combination Product by the fraction C/(C+D) where C is the standard fully-absorbed manufacturing cost of the Licensed Product portion of the combination, and D is the standard fully-absorbed manufacturing cost of the other active ingredients or components included in the Combination Product, as determined by TGTX using its standard accounting procedures consistently applied. In the event that the standard fully-absorbed manufacturing cost of the Licensed Product and/or the other active ingredients or components included in such Combination Product cannot be determined. Net Sales allocable to the Licensed Product in each such country shall be determined by mutual agreement reached in good faith by the parties prior to the end of the accounting period in question based on an equitable method of determining same that takes into account, on a country-by-country basis, all relevant factors (including variations in potency, the relative contribution of each active ingredient in the combination, and relative value to the end user of each active ingredient).

- **1.32** "Person" means any natural person, corporation, firm, business trust, joint venture, association, organization, company, partnership or other business entity, or any government or agency or political subdivision thereof.
- 1.33 "Phase I Trial" means a clinical trial of a Licensed Product in human patients designated as a Phase I Trial and conducted primarily for the purpose of determining the safety of and/or the metabolism and pharmacologic actions of the Licensed Product in humans, as described under 21 CFR § 312.21(a) (as hereafter modified or amended) and any of its foreign equivalents. For purposes of this definition, Phase I Trial shall specifically exclude trials in healthy volunteers.
- 1.34 "Phase II Trial" means a clinical trial of a Licensed Product, designated as a Phase II Trial and the principal purpose of which is to make a preliminary determination that such Licensed Product is safe and active in a patient population for its intended use and is designed to obtain sufficient information about such Licensed Product's efficacy to permit the design of a Phase II Trial(s), and generally consistent with 21 CFR § 312.21(b). For purposes of this definition, Phase II trial shall specifically exclude expansion cohorts from Phase I Trial(s).
- 1.35 "Phase III Trial" means a clinical trial of a Licensed Product in human patients, which is designated as a Phase III Trial or a pivotal trial and is designed (a) to establish that the Licensed Product is safe and efficacious for its intended use; (b) to define warnings, precautions and adverse reactions that are associated with the Licensed Product in the dosage range to be prescribed; and (c) to be, either by itself or together with one or more other clinical trials having a comparable design and size, the final human clinical trial in support of Regulatory Approval of the Licensed Product, and (d) consistent with 21 CFR § 312.21(c) (as hereafter modified or amended) and any of its foreign equivalents.
- **1.36** "Regulatory Authority" means (a) the FDA, (b) the EMA or the European Commission, or (c) any regulatory body with similar regulatory authority over pharmaceutical or biotechnology products in any other jurisdiction anywhere in the world.
- 1.37 "Regulatory Approval" means any and all approvals, licenses, registrations, or authorizations of the relevant Regulatory Authority, necessary for the Development, manufacture, use, storage, import, transport and Commercialization of a given Licensed Product in a particular country or jurisdiction. For the avoidance of doubt, Regulatory Approval outside of the United States shall include any pricing or marketing approval needed prior to the sale of a Licensed Product in the Field.
- 1.38 "Royalty Term" means, on a Licensed Product-by-Licensed Product and country-by-country basis, the period from the First Commercial Sale of a given Licensed Product in such country until the later of (i) twelve (12) years after First Commercial Sale of the applicable Licensed Product in such country, or (ii) the expiry of the last-to-expire DFCI Patent or Additional PDL-1 Intellectual Property containing a Valid Claim to the Licensed Product in such country.
- **1.39** "Sublicensee" means a Person, other than an Affiliate of TGTX, to which TGTX (or its Affiliate) has, pursuant to Section 2.3, granted sublicense rights under any of the license rights granted under Section 2.1. "Sublicense" shall be construed accordingly.
- 1.40 "Sublicense Revenue" means any payments or other consideration that TGTX actually receives from a Sublicensee as consideration for the grant of a Sublicense, including, without limitation, milestone payments, license fees, license maintenance fees and equity. Sublicense Revenue excludes (i) purchases of equity or debt of TGTX, (ii) payments made for TGTX's performance of any research, Development, or Commercialization of any Licensed Product, (iii) (b) royalties on Net Sales (or, in the case of a profit sharing deal structure, shares of net profits) which are covered in Section 5.9, and (iv) any payment or reimbursement of any costs or expenses incurred by TGTX for filing, prosecution, maintenance, or defense of any DFCI Patents or CTI Patents. In the event such consideration received from a Sublicensee is not cash, Sublicense Revenue shall be calculated by TGTX based on the fair market value of such consideration, at the time of the transaction, assuming an arm's length transaction made in the ordinary course of business.
- 1.41 "Tax" or "Taxes" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, withholding, social security, unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not.

- 1.42 "Third Party" means any Person other than Adimab, DFCI, CTI, TGTX, or Affiliates of either of them, or any Sublicensees.
- 1.43 "Third Party Action" means any claim or action made by a Third Party against a Party that claims that a Licensed Product, or its use, Development, manufacture or sale infringes such Third Party's intellectual property rights.
 - 1.44 "United States" or "US" means the United States of America and its territories and possessions.
- 1.45 "Valid Claim" means (a) a claim of an issued and unexpired patent that has not been held permanently revoked, invalid or unenforceable by a patent office, court or other governmental agency of competent jurisdiction in a final and non-appealable judgment (or judgment from which no appeal was taken within the allowable time period) and that is not admitted to be invalid or unenforceable through reissue, disclaimer or otherwise (i.e. only to the extent the subject matter is disclaimed or is sought to be deleted or amended through reissue or (b) a claim of a pending patent application within DFCI Patents or Additional PD-L1 Intellectual Property that has not been abandoned, finally rejected or expired without the possibility of appeal or refiling, provided that (i) Valid Claim shall exclude any such pending claim in an application that has not been granted within the latter of five (5) years after the Effective Date or seven (7) years following the earliest priority filing date for such application (unless and until such claim is granted) and (ii) Valid Claim will exclude any such pending claim that does not have a reasonable bona fide basis for patentability, in either case of (i) or (ii), unless and until such claim is granted. Notwithstanding the foregoing, in the event that a claim in a pending patent application is involved in an interference action declared by the US Patent and Trademark Office or any analogous patentability determination by any other national patent office, and, at the time such proceeding is filed or initiated such claim is a Valid Claim, the time period set forth in subsection (i) above will be stayed for the pendency of such proceeding.
- 1.46 "CTI Patents" means (a) that patent applications set forth as CTI Patents in Schedule 1 hereto; (b) any additions divisionals, continuations, conversion, supplemental examinations, extensions, term restorations, registrations, reinstatements, amendments, reissuances, corrections, substitutions, re-examinations, registrations, revalidations, supplementary protection certificates, renewals, and foreign counterparts of the patent application mentioned in clause (a) above; (c) all patents issuing from the patent applications mentioned in clause (a) or (b) above and any foreign counterparts of any such patents and patent applications, and which shall include, in any case, patents surviving post grant review and *inter partes* review.
- 1.47 "Marketing Approval" each means, within any given country, approval to market a Licensed Product legally as a drug or biologic, including approval of a Biologic License Application (as defined in the U.S. Federal Food, Drug and Cosmetics Act and the regulations promulgated thereunder (21 C.F.R. §§ 600-680) in the United States, or approval of a comparable filing in the United States or any other jurisdiction. Pricing approval need not be obtained in order for Marketing Approval to be achieved.
- **1.48** "Additional PD-L1 Intellectual Property" means CTI Patents and any patents licensed to CTI under the Collaboration Agreement with Adimab (namely, Adimab Program Antibody Patents and Adimab Platform Patents).
 - 1.49 "Licensed Patents" means DFCI Patents, CTI Patents, Adimab Program Antibody Patents and Adimab Platform Patents.
- 1.50 "Adimab Program Antibody Patents" means (a) that patent applications set forth in Schedule 1 hereto; (b) any additions divisionals, continuations, conversion, supplemental examinations, extensions, term restorations, registrations, reinstatements, amendments, reissuances, corrections, substitutions, re-examinations, registrations, revalidations, supplementary protection certificates, renewals, and foreign counterparts of the patent application mentioned in clause (a) above; (c) all patents issuing from the patent applications mentioned in clause (a) or (b) above and any foreign counterparts of any such patents and patent applications, and which shall include, in any case, patents surviving post grant review and *inter partes* review.
- 1.51 "Adimab Platform Patents" means (a) that patent applications set forth in Schedule 1 hereto; (b) any additions divisionals, continuations, conversion, supplemental examinations, extensions, term restorations, registrations, reinstatements, amendments, reissuances, corrections, substitutions, re-examinations, registrations, revalidations, supplementary protection certificates, renewals, and foreign counterparts of the patent application mentioned in clause (a) above; (c) all patents issuing from the patent applications mentioned in clause (a) or (b) above and any foreign counterparts of any such patents and patent applications, and which shall include, in any case, patents surviving post grant review and *inter partes* review.
 - 1.52 CTI Antibodies" means the Antibodies supplied by or on behalf of CTI to TGTX under this Agreement as identified in Schedule 3.

ARTICLE II LICENSES AND OTHER RIGHTS

2.1 Grant of License to TGTX.

- (a) Subject to the terms and conditions of this Agreement, the Collaboration Agreement with Adimab, the excluded rights under Section 3.2(b)(ii) of the Collaboration Agreement with Adimab, the License Agreement, and the reserved rights described in Section 2.4 and Section 2.5 of the License Agreement, effective immediately at the time TGTX is no longer deemed to be an Affiliate of CTI (following receipt by one Party of a written notice from the other Party), CTI hereby grants to TGTX, and TGTX hereby accepts, an exclusive, worldwide, royalty-bearing right and license (with the right to sublicense, subject to the provisions of Section 2.3) under the Additional PD-L1 Intellectual Property and DFCI Patents to (i) research, Develop, manufacture, have manufactured, use, import and Commercialize and have Commercialized the Licensed Products, in and for the Field and (ii) to practice and have practiced any Licensed Processes, in and for the Field. CTI and its Affiliates grant no licenses or rights by implication, estoppel or otherwise under any other patent applications or patents owned in whole or in part by CTI other than as expressly set forth herein.
- (b) Pursuant to an obligation imposed on a sublicensee under Section 3.2(b)(iii) of the Collaboration Agreement with Adimab, TGTX, by executing this Agreement, explicitly agrees to comply with all applicable terms of the Collaboration Agreement with Adimab, including Section 9.3 (Commitments Regarding Program-Benefited Antibodies) thereof.
- 2.2 Affiliates. Effective immediately at the time TGTX is no longer deemed to be an Affiliate of CTI (following receipt by one Party of a written notice from the other Party), TGTX is entitled to extend its licenses under this Article II to its Affiliates, consistent with all of the terms and conditions of this Agreement. If TGTX does extend its license and an Affiliate assumes obligations under the Agreement, TGTX shall be responsible and liable for the acts or omissions of the Affiliate in the exercise of rights under this Agreement. If CTI has a claim arising under this Agreement against an Affiliate, CTI may seek a remedy directly against TGTX and may, but is not required to, seek a remedy against the Affiliate. Any termination of the Agreement under Article X as to TGTX also constitutes termination as to any Affiliates.
- 2.3 Grant of Sublicenses by TGTX. Effective immediately at the time TGTX is no longer deemed to be an Affiliate of CTI (following receipt by one Party of a written notice from the other Party), TGTX shall have the right, in its sole discretion, to grant Sublicenses, in whole or in part, under the license granted in Section 2.1; provided, however, that the granting by TGTX of a Sublicense shall not relieve TGTX of any of its obligations hereunder; and provided, further, that TGTX's right to grant a Person a Sublicense shall be subject to TGTX including within such Sublicense express provisions binding the Sublicensee to terms and condition consistent with those contained herein. TGTX shall be and remain fully responsible and primarily liable for the compliance by Sublicensees with the terms and conditions of this Agreement (as applicable to them) as if such Sublicensees were TGTX hereunder. TGTX shall promptly provide a copy of each executed sublicense agreement and any modifications of the sublicense agreement (provided that such copy may be redacted to remove commercially sensitive terms that are not necessary to confirm compliance with the terms and conditions of this Agreement) following execution of such agreement.
- **2.4 Delivery of DFCI Know-How, DFCI Antibodies, and CTI Antibodies.** Effective immediately at the time TGTX is no longer deemed to be an Affiliate of CTI, CTI shall deliver to TGTX DFCI Know-How, DFCI Antibodies, and CTI Antibodies within sixty (60) days of the Effective Date of this Agreement.
- 2.5 Extension of Rights. During such time as TGTX is deemed an Affiliate of CTI, CTI extends to TGTX all of its rights under the Collaboration Agreement with Adimab (except Adimab Materials, Optioned Program Antibody Know-How and Program Antibody Know-How), License Agreement (except CAIX) subject to the terms and conditions of this Agreement, the Collaboration Agreement with Adimab and the License Agreement, provided that such rights shall be limited to the Field and shall exclude the right to make and have made Licensed Products. TGTX hereby assumes the obligations of CTI under the Collaboration Agreement with Adimab and License Agreement with respect to its exercise of rights thereunder. Such extension of rights shall automatically terminate at the time TGTX is no longer deemed to be an Affiliate of CTI. It is the intention of TGTX and CTI for this Agreement to be consistent with the Collaboration Agreement with Adimab and the License Agreement. During the term of this Agreement, if CTI shall default on any obligations owed Adimab or DFCI then TGTX shall have the right to cure such defaults and set any amounts incurred by TGTX in curing such defaults against any future payments TGTX may owe to CTI.

ARTICLE III RIGHTS, DUTIES AND DILIGENCE

- 3.1 Diligence by TGTX. TGTX shall use Commercially Reasonable Efforts to Develop and to Commercialize Licensed Products targeting PD-L1 and GITR in the Field. The Parties acknowledge that TGTX may Develop and Commercialize Licensed Products that are a Combination Product containing one or more DFCI Antibodies, CTI Antibodies, or Derivatives thereof. Except as otherwise provided herein or agreed upon in writing, CTI agrees that it will not make, use or sell Licensed Products in the Field ("Exclusivity Covenant"). In addition, TGTX shall have the option (the "Autoimmune Option") to include Autoimmune Diseases in the Field by providing notice to CTI and making a \$1,000,000 payment. Such Autoimmune Option can be exercised up to 7 years from the date of the Original Agreement.
- **3.2 Projected Milestone Dates.** TGTX shall use its commercially reasonable efforts to meet the following milestones ("Milestones") by the dates specified in this paragraph, subject to annual adjustment as described below.

For purposes of this Section 3.2, CTI will consider efforts of an Affiliate or Sublicensee as efforts of TGTX.

(a) Milestone Dates for a Licensed Product Targeting PD-L1

Milestone	Achievement Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] years from the Effective Date
[*] for first PD-L1 Licensed Product	[*] Years from the Effective Date

(b) Milestone Dates for a Licensed Product Targeting GITR

Milestone	Achievement Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] years from the Effective Date
[*] for first GITR Licensed Product	[*] Years from the Effective Date

- **3.3** Adjustments. The parties acknowledge that since the program is in early pre-clinical development that the dates included in the Milestone table above are rough estimates to provide Adimab, DFCI and CTI a preliminary projection of what can be achieved by what dates, the accuracy of which the parties agree is impossible to predict and will be based on many factors completely outside the control of TGTX and its Diligence Efforts. On an annual basis, with its report contained below, TGTX will, in good faith, update the dates in the Milestones table above to provide CTI an updated assessment of the timing of the upcoming milestones. Upon providing such update, the table above shall be deemed amended notwithstanding Section 11.5 hereof.
- **3.4** Development and Commercialization Reports. Within 50 days of the Effective Date and at least 10 days before each anniversary of the Effective Date, TGTX shall provide to CTI a written report describing the efforts by TGTX, or any Affiliates or Sublicensees, to bring one or more Licensed Products to the marketplace. The report must be in sufficient detail to permit CTI to monitor TGTX' compliance with the due diligence provisions of this Agreement.

TGTX shall include at least the following in these reports: (a) a summary of TGTX' progress toward meeting the goals and objectives that had been established for the previous year; (b) a summary of TGTX' goals and objectives for the ensuing year for developing and commercializing Licensed Products, including an identification of Licensed Products that TGTX intends to develop, if any; and (c) to the extent not covered by the foregoing, a summary of TGTX' progress in meeting the Milestone timelines above.

If multiple technologies are covered by this Agreement, the progress report must provide the information set forth above for each Licensed Product.

- 3.5 Failure to Perform. TGTX's failure to use commercially reasonable efforts to perform any due diligence requirement provided in Section 3.1 through 3.4 is grounds for CTI to terminate this Agreement according to Section 10.2(d); provided that CTI shall only have the right to terminate this Agreement with respect to the specific Licensed Product for which such failure is claimed and the Agreement shall remain in full force and effect for the remaining Licensed Products. In the alternative, CTI may terminate the Exclusivity Covenant (if such failure occurs while TGTX is an Affiliate of CTI) or convert the exclusive licenses granted under this Agreement to a non-exclusive license (if such failure occurs after the time TGTX ceases to be an Affiliate of CTI), as further provided in Section 3.6, as to the specific Licensed Product for which such failure is claimed.
- 3.6 Conversion to Non-exclusive License. If (i) the Exclusivity Covenant is terminated as provided in Section 3.5 or (ii) the exclusive license granted under this Agreement is converted to a non-exclusive license for any Licensed Product as provided in Section 3.5, this Agreement is automatically amended as follows as it relates to such Licensed Product; (a) the exclusive license of Section 2.1 becomes a non-exclusive license, (b) TGTX loses the right to grant sublicenses under Section 2.3; provided that any sublicense granted prior to such conversion shall continue and not be affected by such conversion, (c) the obligations of Sections 3.1 through 3.4 continue to apply, (d) the obligation under Section 3.10 no longer applies, (e) TGTX has no further rights or obligations under Article VI; provided that CTI shall keep TGTX apprised of any new filings of patent applications and issuance of patents that fall within the DFCI Patents, and (f) CTI has the sole right to pursue apparent infringements and the terms of Article VI no longer apply.
- 3.7 Costs and Expenses. As between CTI and TGTX, (a) TGTX shall be solely responsible for all costs and expenses related to Development, and Commercialization of the Licensed Products, including without limitation costs and expenses associated with all preclinical activities and clinical trials, and all regulatory filings and administrative proceedings relating to Licensed Products in the Field and (b) CTI shall be the sole and exclusive manufacturer of Licensed Products for TGTX and its Affiliates and Sublicensees, such that TGTX and its Affiliates and Sublicensees shall purchase all of its requirements of Licensed Products from CTI and will not make or have made Licensed Products directly or through its Affiliates or Sublicensees unless CTI is unable to provide sufficient supplies at competitive prices, the terms of which shall be negotiated in a manufacturing and supply agreement. In such case, TGTX may manufacture Licensed Products provided TGTX obtains all necessary licensing rights for such manufacture. TGTX shall be solely responsible for the costs of such additional licenses to manufacture Licensed Products. With the exception of the above, CTI shall be solely responsible for all costs and expenses related to CMC including without limitation, CMC development and scale-up, CMC validation, analytical method development and validation, stability testing, manufacturing, finishing and release. TGTX shall reimburse CTI for CTI's out-of-pocket cost for Licensed Product used by TGTX for its Development activities and shall pay CTI a manufacturing transfer price for Commercial supplies equal to CTI's out-of-pocket cost of Licensed Product plus the lesser of: (a) 30% of such cost and (b) 3% of Net Sales generated by the materials supplied The Parties agree to execute a manufacturing and supply agreement within a reasonable time after the execution of the Agreement on these terms and including such other customary and reasonable terms.
- **3.8 Patent Marking**. TGTX agrees that with respect to each unit or package of Licensed Products sold in a given country, TGTX shall comply with the customary patent marking laws and practices of such country as to the applicable Additional PD-L1 Intellectual Property and DFCI Patents.
- **3.9 Trademarks**. As between TGTX and CTI, TGTX shall have the sole authority to select trademarks for Licensed Products and shall own all such trademarks. CTI does not grant TGTX the right to use any trademarks of CTI, Adimab, DFCI or its respective Affiliates.
- 3.10 U.S. Manufacture. To the extent TGTX manufactures Licensed Products (e.g. if TGTX and CTI agree that CTI will no longer be the sole manufacturer of Licensed Products), TGTX shall manufacture Licensed Products leased, used or sold in the United States substantially in the United States as required by 35 U.S.C. 204 and 37 C.F.R. 401 et. seq., as amended. TGTX shall also require any Affiliate(s) or Sublicensee(s) to comply with this U.S. manufacture requirement. Notwithstanding the foregoing, if TGTX or its Affiliate(s) or Sublicensee(s) determines that it is not commercially feasible or reasonable to manufacture such Licensed Products in the United States or determines that it is necessary to have additional manufacturers outside the United States for back-up supply or to supply Licensed Products outside the United States, then CTI agrees to make reasonable efforts to assist TGTX, or its Affiliate(s) or Sublicensee(s), as applicable, at TGTX' expense, in obtaining any necessary permission from the appropriate government authorities to manufacture such Licensed Products outside the United States.
- **3.11** Other Government Laws. TGTX shall comply with, and ensure that its Affiliates and Sublicensees comply with, all government statutes and regulations that relate to Licensed Products. These include but are not limited to FDA statutes and regulations, the Export Administration Act of 1979, as amended, codified in 50 App. U.S.C. 2041 et seq. and the regulations promulgated thereunder or other applicable export statutes or regulations.
- 3.12 Publicity. TGTX, its Affiliate and Sublicensees are not permitted to use the names of CTI, DFCI, its related entities or its employees, or any adaptations thereof, in any advertising, promotional or sales literature, or in any securities report required by the Securities and Exchange Commission (except as required by law), without the prior written consent of CTI, Adimab and/or DFCI in each case. However, TGTX may (a) refer to publications in the scientific literature by employees of Adimab, DFCI, or CTI or (b) state that a license from Adimab, DFCI, or CTI has been granted as provided in this Agreement.
- 3.13 Other Agreements. In the event that TGTX determines to conduct a clinical trial of a Licensed Product in the Field in the United States, TGTX shall consider in good faith and discuss with DFCI the potential of engaging DFCI to serve as a clinical site for such clinical trial; provided that (a) DFCI has the appropriate expertise and patient population to conduct the clinical trial, and (b) DFCI is economically competitive with other sites having substantially similar expertise and patient populations to conduct such clinical trial.

ARTICLE IV REGULATORY MATTERS

4.1 Regulatory Filings. As between CTI and TGTX, TGTX (or its applicable Affiliate) shall own and maintain all regulatory filings made after the Effective Date for Licensed Products and all Regulatory Approvals for Licensed Products. Once per year, representatives from CTI may visit TGTX and review all such regulatory filings, provided such representatives do not have a conflict of interest or involvement with any competitive companies or technologies and agree to TGTX's confidentiality agreement.

ARTICLE V FINANCIAL PROVISIONS

- **5.1 Upfront Fee**. Upon the signing of the Original Agreement, TGTX paid CTI an up-front, non-creditable, non-refundable fee in the amount of Five Hundred Thousand Dollars (\$500,000). Upon the signing of this Agreement, TGTX shall pay CTI a non-creditable, non-refundable fee in the amount of One Million Dollars (\$1,000,000).
- **5.2 Maintenance Fee.** Within thirty (30) days following the second anniversary of the Effective Date and each anniversary thereafter, TGTX shall pay CTI an annual license maintenance fee in the amount of [*] Dollars (\$[*]). Such fees are creditable against milestone payments due pursuant to Section 5.6, royalties due pursuant to Section 5.7 or Sublicense Revenue Share Payments (as defined in Section 5.9).
 - 5.3 Reserved.

5.4 Milestone Payments.

(a) **PD-L1-based Milestones.** As further partial consideration for CTI's grant of the rights to TGTX hereunder, TGTX shall pay to CTI the following one-time, PD-L1 targeting product-based milestone payments with regard to each Licensed Product targeting PD-L1 (as specifically set forth below). TGTX will pay the relevant milestone payment within thirty (30) days of such achievement.

PD-L1 Targeting Product-based Milestone Events	Milestone Payment
[*]*	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]

* [*].

If a later-stage PD-L1 targeting product-based clinical milestone event is achieved for any Licensed Product targeting PD-L1 without one or more earlier-stage clinical milestone events having been achieved for that Licensed Product, then TGTX shall pay the PD-L1 Milestone Payment(s) for such previous clinical milestone event(s) along with the payment for the most recently achieved milestone event. If a milestone event related to [*] is achieved without one or more of the clinical milestone events being achieved, then TGTX shall pay the PD-L1 Milestone Payment(s) for such previous clinical milestone event(s) along with the payment for the first milestone event related to [*].

(b) **GITR-based Milestones.** As further partial consideration for CTI's grant of the rights to TGTX hereunder, TGTX shall pay to CTI the following one-time, GITR targeting product-based milestone payments with regard to each Licensed Product targeting GITR (as specifically set forth below). TGTX will pay the relevant milestone payment within thirty (30) days of such achievement.

GITR Targeting Product-based Milestone Events	Milestone Payment
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]
[*]	\$[*]

- (c) Combination Approval Milestones. If any of the above milestones in (a) and (b) are triggered as a result of a combination approval of two or more Licensed Products or combination clinical trial of two or more Licensed Products, only one milestone payment (the higher payment) shall be due to CTI as if the combination was a single Licensed Product.
- (d) **Aggregate Net Sales Achievement Milestones:** As further consideration for CTI's grant of the rights to TGTX hereunder, TGTX shall pay to CTI the following one-time milestone payments upon first achievement of worldwide Net Sales (as specifically set forth below) by TGTX and its Affiliates and Sublicensees. TGTX will pay the relevant milestone payment within 90 days of such achievement.

Aggregate Net Sales Achievement Milestones	
The first time aggregate worldwide Net Sales for all Licensed Products exceeds \$[*] in any Calendar Year	\$[*]
The first time aggregate worldwide Net Sales for all Licensed Products exceeds \$[*] in any Calendar Year	\$[*]
The first time aggregate worldwide Net Sales for all Licensed Products exceeds \$[*] in any Calendar Year	\$[*]
The first time aggregate worldwide Net Sales for all Licensed Products exceeds \$[*] in any Calendar Year	\$[*]

5.5 Royalty, Etc. Payments for Licensed Products.

(a) With respect to Net Sales of all Licensed Products: As further consideration for CTI's grant of the rights to TGTX hereunder, TGTX shall pay to CTI a royalty of on aggregate annual worldwide Net Sales of all such Licensed Products by TGTX and its Affiliates and Sublicensees (but excluding Net Sales of a given Licensed Product after its applicable Royalty Term) at the percentage rates set forth below:

Annual Worldwide Net Sales of All Licensed Products per Calendar Year (US Dollars)	Incremental Royalty Rate
For Net Sales of such Licensed Products from \$0 up to and including \$[*]	[*]%
For that portion of Net Sales of such Licensed Products that is greater than \$[*]	[*]%

- (b) In no event shall the manufacture of a Licensed Product give rise to a royalty/payment in the nature of royalties obligation until the particular unit of Licensed Product is sold; but if Net Sales of a particular unit of Licensed Product might or might not be subject to a royalty/payment in the nature of royalties payment (e.g., manufactured in Country A where the Royalty Term has expired but sold in Country B where the Royalty Term has not expired), the sale shall be deemed to be subject to a royalty/payment in the nature of royalties payment. For clarity, TGTX's obligation to pay royalties to CTI under Section 5.7(a) is imposed only once with respect to the same unit of Licensed Product regardless of the number of DFCI Patents pertaining thereto or the number of times such Licensed Product has been sold or transferred to a Person.
- (c) On a Licensed Product by Licensed Product and country-by-country basis, upon expiration of the Royalty Term for a Licensed Product in a country, the rights, licenses and sublicenses granted to TGTX hereunder with respect to such Licensed Product in such country shall continue in effect but become fully paid-up, royalty-free, and perpetual.
- (d) Reserved.
- (e) Reserved.
- (f) In the event that a Licensed Product in a country is not Covered by a Valid Claim of a Licensed Patent, royalties with respect to such Licensed Product in such country shall be reduced by [*] percent ([*]%) of the applicable royalty rate as set forth in Section 5.5(a) and shall be due for the period commencing with the First Commercial Sale of such Licensed Product in such country and ending ten (10) years from date of such First Commercial Sale.
- (g) Notwithstanding the above, in no event shall the royalty rates set forth in Section 5.7(a) be reduced under 5.7(d), (e), and (f) above by more than [*]% collectively.
- 5.6 Timing of Royalty Payment. Royalties/payments in the nature of royalties payable under Section 5.5 shall be payable on actual Net Sales and shall accrue at the time provided therefor by US GAAP. Royalty/payment in the nature of royalties obligations that have accrued during a particular Calendar Quarter shall be paid, on a Calendar Quarter basis, within 80 days after the end of each Calendar Quarter during which the royalty/payment in the nature of royalties obligation accrued; provided that within 40 days after the conclusion of each Calendar Year TGTX shall provide notice to CTI of any adjustments necessary to account for any royalties/payment in the nature of royalties which were overpaid or underpaid for such prior Calendar Year's Calendar Quarters, and the Parties shall promptly true-up based on such adjustments, provided however, the lapse of such 50-day period shall not impact the right of TGTX to credit any over-payments discovered during an audit against future royalties due under Section 5.5 hereof.

- 5.7 Sublicense Revenue. TGTX shall pay to CTI [*] percent ([*]%) of all Sublicense Revenue received by TGTX ("Sublicense Revenue Share Payments"). Sublicense Revenue Share Payments shall be paid, on a Calendar Quarter basis, within 80 days after the end of each Calendar Quarter during which the respective Sublicense Revenue is received.
- 5.8 Royalty Reports and Records Retention. Within 50 days after the end of each Calendar Quarter during which Licensed Products have been sold, TGTX shall deliver to CTI, together with the applicable royalty/payment in the nature of royalties payment due, a written report, on a Licensed Product-by-Licensed Product (and specifying non-Covered status, as applicable) and country-by-country basis, of (a) (a)Number of Licensed Products manufactured and sold by TGTX, and any Affiliates or Sublicensees, in each country; (b) gross invoiced (or otherwise charged) amounts of sales, by TGTX and its Affiliates and Sublicensees, of Licensed Products subject to royalty payments for such Calendar Quarter (and, if non-Covered, subject to royalty/payment in the nature of royalties payments for such Calendar Quarter), (c) amounts deducted by category (following the definition of Net Sales) from such gross invoiced amounts to calculate Net Sales, (d) Net Sales subject to royalty or royalty/payment in the nature of royalties payments for such Calendar Quarter and Calendar Year to date, and (e) the corresponding royalty or royalty/payment in the nature of royalties, and (f) the nature and amount of Sublicense Revenue received by TGTX. Such report shall be deemed "Confidential Information" of TGTX subject to the obligations of Article VII of this Agreement. For three years after each sale of a Licensed Product (whether Covered or not), TGTX shall keep (and shall ensure that its Affiliates and Sublicensees shall keep) complete and accurate records of such sale in sufficient detail to confirm the accuracy of the royalty or royalty/payment in the nature of royalties calculations hereunder.
- **5.9** CTI shall be solely responsible for paying directly to DFCI all payments due to DFCI under Section 5 of the License Agreement that arise out of the exercise of rights by TGTX under this Agreement, including, without limitation, royalties on TGTX's Net Sales. Likewise, CTI shall be solely responsible for paying directly to Adimab all payments due to Adimab under Section 4 of the Collaboration Agreement with Adimab that arise out of the exercise of rights by TGTX under this Agreement, including, without limitation, royalties on TGTX's Net Sales.

5.10 Reserved.

5.11 Books and Audits.

TGTX shall keep, and shall require its Affiliates and Sublicensees to keep, true books of account containing an accurate record (together with supporting documentation) of all data necessary for determining the amounts payable to CTI. TGTX shall keep it records at its principal place of business or the principal place of business of the appropriate division of TGTX to which this Agreement relates and shall require its Affiliates and Sublicenses to keep their books and records in the same manner.

- (a) Commencing on the earlier of (i) the First Commercial Sale (of the first Licensed Product to have a First Commercial Sale) or (ii) receipt of Sublicense Revenue, and continuing until one Calendar Year after the conclusion of the final Royalty Term, upon the written request of CTI, and not more than once in each Calendar Year, TGTX shall permit, shall cause its Affiliates to permit, an independent certified public accounting firm of nationally recognized standing selected by CTI (who has not been engaged by CTI to provide services in any other capacity at any time during the three-year period before such selection), and reasonably acceptable to TGTX or such Affiliate, to have access to and to review, during normal business hours upon reasonable prior written notice, the applicable records of TGTX and its Affiliates to verify the accuracy of the royalty payments and Sublicense Revenue Share Payments. Such review may cover: (i) the records for the Calendar Year ending not more than three years before the date of such request, and (ii) only those periods that have not been subject to a prior audit.
- (b) If such accounting firm concludes that additional amounts were owed during such period, TGTX shall pay the additional royalties and/or royalties/payment in the nature of royalties within 15 days after the date such public accounting firm delivers to TGTX such accounting firm's written report. If such accounting firm concludes that an overpayment was made, such overpayment shall be fully creditable against amounts payable in subsequent payment periods. If TGTX disagrees with such calculation, TGTX may contest such calculation in writing at which point the parties will work in good faith to submit the matter to a mediator for resolution. If the parties are unable to reach an agreement via mediation, then TGTX may initiate a court action to seek to recover the additional payment or to increase the amount of credit or reimbursement. CTI shall pay for the cost of any audit by CTI, unless TGTX has underpaid CTI by 5% or more for a specific royalty period, in which case TGTX shall pay for the reasonable costs of audit, as well as any additional sum that would have been payable to CTI had the TGTX reported correctly, plus interest as set forth in Section 4.14.
- (c) Each Party shall treat all information that it receives under this Section 5.10 in accordance with the confidentiality provisions of Article VII of this Agreement, and shall cause its accounting firm to enter into an acceptable confidentiality agreement with the audited Party obligating such firm to retain all such financial information in confidence pursuant to such confidentiality agreement, except to the extent necessary for a Party to enforce its rights under the Agreement.

5.12 Mode of Payment and Currency. All payments to CTI under this Agreement, whether or not in respect of Net Sales or milestone events, shall be made by deposit of US Dollars in the requisite amount to the following, which CTI may from time to time amend by advance written notice to TGTX.

by check:

Checkpoint Therapeutics, Inc. 2 Gansevoort Street New York, NY 10014

by wire transfer:

[To be provided]

Conversion of sales or expenses recorded in local currencies to Dollars will be performed in a manner consistent with TGTX's normal practices used to prepare its audited financial statements for external reporting purposes, provided that such practices use a widely accepted source of published exchange rates. Based on the resulting Net Sales in US Dollars, the then applicable royalties/payment in the nature of royalties shall be calculated.

- 5.13 Late Payments. If a Party does not receive payment of any sum due to it on or before the due date therefor, simple interest shall thereafter accrue on the sum due to such Party from the due date until the date of payment at a rate equal to the lesser of (a) US dollar one-month LIBOR plus 300 basis points, or (b) the maximum rate permissible under applicable Law. Accrual and payment of interest shall not be deemed to excuse or cure breaches of contract arising from late payment or nonpayment. Waiver or deferral by CTI of any payment owed under any paragraph under this Article V may not be construed as a waiver or deferral of any subsequent payment owed by TGTX to CTI.
- 5.14 Taxes. All amounts due hereunder exclude all applicable sales, use, and other taxes and duties, and TGTX shall be responsible for payment of all such taxes (other than taxes based on CTI's income) and duties and any related penalties and interest, arising from the payment of amounts due under this Agreement. The Parties agree to cooperate with one another and use Commercially Reasonable Efforts to avoid or reduce tax withholding or similar obligations in respect of royalties, payments in the nature of royalties, milestone payments, and other payments made by TGTX to CTI under this Agreement. To the extent TGTX is required to withhold taxes on any payment to CTI, TGTX shall pay the amounts of such taxes to the proper governmental authority in a timely manner and promptly transmit to CTI official receipts issued by the appropriate taxing authority and/or an official tax certificate, or such other evidence as CTI may reasonably request, to establish that such taxes have been paid. CTI shall provide TGTX any tax forms that may be reasonably necessary in order for TGTX to not withhold tax or to withhold tax at a reduced rate under an applicable bilateral income tax treaty. CTI shall use Commercially Reasonable Efforts to provide any such tax forms to TGTX at least 45 days before the due date for any payment for which CTI desires that TGTX apply a reduced withholding rate. Each Party shall provide the others with reasonable assistance to enable the recovery, as permitted by applicable law, of withholding taxes, value added taxes, or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding tax or value added tax. CTI shall indemnify and hold TGTX harmless from and against any penalties, interest or other tax liability arising from any failure by TGTX (at the express request of CTI) to withholding by reduction (at the express request of CTI) in its withholding.
- 5.15 Currency Conversion. If any currency conversion is required in connection with any payment owed to CTI, the conversion will be made at the buying rate for the transfer of such other currency as quoted by the Wall Street Journal on the last business day of the applicable accounting period in the case of any payment payable with respect to a specified accounting period or, in the case of any other payment, the last business day before the date the payment is due.

ARTICLE VI

6.1 Patent Prosecution and Maintenance.

- (a) **DFCI Patents and Additional PD-L1 Intellectual Property**. TGTX shall reimburse CTI for [*]% of the patent expenses incurred under the License Agreement and incurred for the filing, maintenance and prosecution of patents included in Additional PD-L1 Intellectual Property for which CTI is responsible.
- (b) New or Revised Applications. CTI will, upon learning from DFCI of an intention to file or revise one or more patent applications which are DFCI Patents subject to the License grant in Article II, promptly inform TGTX of such intention, and will provide TGTX with the opportunity to comment on the content of such DFCI patent application before CTI sends comments to DFCI on such filing. CTI shall include any such reasonable TGTX comments in the comments to be sent to DFCI. CTI will inform TGTX of CTI's intention to file or revise one or more patent applications which are included in Additional PD-L1 Intellectual Property for which CTI is responsible subject to the License grant in Article II and will provide TGTX with the opportunity to comment on the content of such one or more patent applications. CTI shall take into account any such reasonable TGTX comments.
- (c) Liaising. CTI shall keep TGTX promptly and regularly informed of the course of the filing and prosecution of DFCI Patents and patents included in Additional PD-L1 Intellectual Property for which CTI is responsible or related proceedings (e.g., interferences, oppositions, reexaminations, reissues, revocations or nullifications) in a timely manner, and to reasonably take into consideration the advice and recommendations of TGTX.
- (d) Election Not to File/Prosecute/Maintain DFCI Patents and Patents Included in Additional PD-L1 Intellectual Property for which CTI is Responsible. TGTX acknowledges and agrees that CTI and DFCI shall not be required to file, prosecute or maintain patents included in Additional PD-L1 Intellectual Property for which CTI is responsible and DFCI Patents, respectively, provided, however, if DFCI decides to not pursue or maintain any such DFCI Patents then CTI shall promptly notify TGTX so the Parties can determine if they would like to assume responsibility for such activities in DFCI's name but at the Parties expense. The same notice applies to any decision made by CTI to drop a case included in Additional PD-L1 Intellectual Property for which CTI is responsible. In either event, TGTX will no longer owe any royalty obligation on account of such (country-level) DFCI Patents or patents included in Additional PD-L1 Intellectual Property for which CTI is responsible assumed by the Parties or TGTX, as the case might be. Similarly, to the extent CTI does not want to continue funding the patent costs of any portion of DFCI Patents, CTI will notify TGTX and give TGTX an opportunity to assume responsibility for such Patents at TGTX's expense, in which case TGTX shall owe DFCI directly the royalties due under the License Agreement and shall no longer owe royalty obligation to CTI on account of such (country-level) DFCI Patents assumed by TGTX. The Parties acknowledge that if neither CTI nor TGTX continues funding patent costs then such portion of DFCI Patents will no longer be included as DFCI Patents. The same course of action will be followed by the Parties in connection with patents included in Additional PD-L1 Intellectual Property for which CTI is responsible.
- 6.2 Certification under Drug Price Competition and Patent Restoration Act. Each of TGTX and CTI shall provide within a reasonable time written notice to the other of any certification of which they become aware filed pursuant to 21 U.S.C. Section 355(b)(2)(A) (or any amendment or successor statute thereto) claiming that any DFCI Patents covering a Licensed Product, or the manufacture or use of each of the foregoing, are invalid or unenforceable, or that infringement will not arise from the manufacture, use or sale in the US of a Licensed Product by a Third Party.
- 6.3 Listing of Patents. To the extent a DFCI Patent and/or a patent included in Additional PD-L1 Intellectual Property is applicable in support of a label associated with an approval to market a Licensed Product, TGTX shall have the sole right to determine which of such DFCI Patents and/or a patents included in Additional PD-L1 Intellectual Property, if any, shall be listed for inclusion in the Approved Drug Products with Therapeutic Equivalence Evaluations publication pursuant to 21 U.S.C. Section 355, any equivalent publication for biologics, or any successor Law in the United States, together with any comparable Laws in any other country. CTI will co-operate with TGTX to list any of said DFCI Patents and patents included in Additional PD-L1 Intellectual Property.

6.4 Enforcement of Patents.

(a) **Notice.** If either TGTX or CTI believes that a Licensed Patent is being infringed in the Field by a Third Party or if a Third Party claims that any Licensed Patent is invalid or unenforceable, the Party possessing such knowledge or belief shall notify the other and provide it with details of such infringement, misappropriation or claim that are known by such Party.

(b) Action by DFCI.

- (i) **Procedure.** TGTX acknowledges that DFCI is responsible for enforcing its DFCI Patents and prosecuting apparent infringers when, in DFCI's judgment, such action may be reasonably necessary and justified. TGTX may request that CTI request DFCI to take steps to protect the DFCI Patents from an apparent infringement. However, TGTX recognizes that before DFCI must respond to the request, TGTX shall supply CTI to provide to DFCI (i) an opinion of qualified legal counsel demonstrating to DFCI's reasonable satisfaction that an infringement of the DFCI Patents exists in a particular country and (ii) with written evidence demonstrating to DFCI's reasonable satisfaction that a Substantial Infringement of the DFCI Patents exists in a particular country ("Substantial Infringer").
- (ii) DFCI has three months from the date of receiving satisfactory written evidence from CTI of a Substantial Infringement to decide whether it will seek to terminate the Substantial Infringement. DFCI shall give CTI notice of its decision by the end of this three-month period, which CTI shall promptly forward to TGTX. If DFCI notifies CTI that it intends to prosecute the alleged infringer, then DFCI has six (6) months from the date of its notice to CTI to either (a) cause the Substantial Infringement to terminate or (b) initiate legal proceedings against the infringer. If any such suit is brought by DFCI in its own name, or jointly with CTI if required by law, it will be at DFCI's expense and on its own behalf, but DFCI shall not be obligated to bring more than one such suit at a time.
- (iii) CTI's Right to Join. If CTI shall exercise its rights to join any legal proceeding brought by DFCI under Section 6.4 of the License Agreement, then TGTX shall have the right to join CTI under the same terms and conditions of paragraph 6.4(b)(iii) of the License Agreement.

(c) Action by CTI and TGTX.

- (i) **Procedure.** If CTI has the right to prosecute a Substantial Infringement under Section 6.4(c) of the License Agreement, then CTI shall promptly notify TGTX, and it may initiate a legal proceeding against the alleged infringer. If CTI decides that it will not commence any legal proceeding with respect to the Substantial Infringement, then such right to prosecute a Substantial Infringement under Section 6.4(c) of the License Agreement passes to TGTX hereunder.
- (ii) **TGTX's Right To Join.** TGTX independently has the right to join any legal proceeding brought by CTI under this Section 6.4 and fund up to fifty percent of the cost of the legal proceeding from the date of joining. If TGTX elects to join as a party plaintiff pursuant to this Section 6.4, TGTX may jointly participate in the action with CTI, but CTI's counsel will be lead counsel.
- (iii) Reduction of Royalties. If CTI initiates legal proceedings under Section 6.4 of the License Agreement and TGTX joins pursuant to this Section 6.4, then TGTX shall have the same rights as CTI has under Section 6.4(c)(iii) of the License Agreement. Additionally, if TGTX prosecutes pursuant Section 6.4(i) of this Agreement after CTI decides not to prosecute and neither DFCI nor CTI independently (except as a necessary party) join the proceeding, then TGTX may deduct up to [*] percent ([*]%) of TGTX's documented costs and expenses of the proceeding (including reasonable attorney fees) from running and minimum royalties payable to CTI under Section 5.7(a) of this Agreement from sales of Licensed Products covered by the patent(s)-in suit. However, TGTX may not reduce CTI's royalty payments by more than fifty percent of the amount otherwise due under Article V. If [*] percent ([*]%) of TGTX's costs and expenses exceed the amount of royalties deducted by TGTX for any calendar year, TGTX may, to that extent, reduce the royalties due to CTI in succeeding calendar quarters for so long as TGTX is actively engaged in legal proceedings to terminate the Substantial Infringement. However, TGTX may not reduce total royalties due to CTI in a given calendar quarter by more than [*] percent ([*]%). TGTX's right to reduce royalty payments to CTI under this paragraph 6.4(c)(iii) applies only for so long as the Substantial Infringement continues.
- (iv) **Settlement.** Regardless of whether CTI or DFCI is joined or joins any legal proceeding initiated by TGTX, TGTX acknowledges and agrees that no settlement, consent judgment or other voluntary final disposition of the legal proceeding may be entered into without the consent of DFCI and CTI.

- **6.5 Cooperation.** If one party initiates legal proceedings to enforce the DFCI Patents pursuant to this Article VI, the other party shall cooperate with and supply all assistance reasonably requested by the party initiating the proceedings, at the initiating party's request and expense.
- 6.6 Distribution of Amounts Paid by Third Parties. Any amounts recovered by the Party initiating an Action pursuant to this Section 6.6, whether by settlement or judgment, shall be allocated in the following order: to reimburse the Parties for all out-of-pocket costs and expenses incurred in connection therewith, including attorneys' fees. If such recovery is insufficient to cover all such costs and expenses of both Parties, it will be shared pro-rata in proportion to the relative amount of such costs and expenses incurred by each Party. If after such reimbursement any funds remain from such damages, the remaining amount of such recovery shall be allocated as follows: the portion thereof attributable to "lost sales" in the Field shall be retained by TGTX and shall be deemed to be Net Sales for the Calendar Quarter in which the amount is actually received by TGTX and TGTX shall pay to CTI a royalty on such portion based on the royalty rates set forth in Section 5.7(a), and the portion thereof not attributable to "lost sales" and is not allocated to DFCI under Section 6.6 of the License Agreement shall be allocated 50% to TGTX and 50% to CTI.
- 6.7 Declaratory Judgment Actions. In the event that any third party initiates a declaratory judgment action alleging the invalidity or unenforceability of the Licensed Patents with respect to claims relating to the Field, or if any third party brings an infringement action against TGTX or its Affiliates or Sublicensees because of the exercise of the rights granted TGTX under this Agreement, then TGTX shall have the right to defend such action under its own control and at its own expense; provided, however, that TGTX acknowledges that DFCI has the right to assume control of such defense, at its own expense, if such action is directed to a DFCI Patent and DFCI in good-faith believes that assuming control of such defense is beneficial to CTI and DFCI. TGTX shall NOT enter into any settlement, consent judgment or other voluntary final disposition of any action under this Section 6.7 without the consent of the other party (the patent owner), which consent shall not be unreasonably withheld unless the settlement includes any express or implied admission of liability or wrongdoing on the other party's part, in which case the other party's right to grant or deny consent is absolute and at its sole discretion. Any recovery shall be first applied to reimburse each party pro rata for any out-of-pocket expenses it may have incurred with respect to defense of such action, and the remainder shall be retained entirely by the party controlling the action; provided, however, that any recovery for infringement will be distributed as described in Section 6.6.

ARTICLE VII CONFIDENTIALITY

- 7.1 Definitions. CTI and TGTX each recognizes that during the Term, it may be necessary for a Party (the "Disclosing Party") to provide Confidential Information (as defined herein) to another Party (the "Receiving Party") that is highly valuable, the disclosure of which would be highly prejudicial to such Party. The disclosure and use of Confidential Information shall be governed by the provisions of this Article VII. Neither Party shall use the other's Confidential Information except as expressly permitted in this Agreement. For purposes of this Agreement, "Confidential Information" means all information (including information relating to the business, operations and products of a Party or any of its Affiliates) disclosed by the Disclosing Party to the Receiving Party and which reasonably ought to have been understood to be confidential and/or non-public information at the time disclosed to the Receiving Party, or which is designated in writing by the Disclosing Party as "Confidential" (or equivalent), or which when disclosed orally to the Receiving Party is declared to be confidential by the Disclosing Party and is so confirmed in a writing delivered to the Receiving Party within 30 days after such oral disclosure, including but not limited to any technical information, Know-How, trade secrets, or inventions (whether patentable or not), that such Party discloses to another Party under this Agreement, or otherwise becomes known to another Party by virtue of or that relates to this Agreement.
- 7.2 Obligation. The Parties agree that they will disclose the other Party's Confidential Information to its own (or its respective Affiliate's, or with respect to TGTX, its Sublicensees') officers, employees, consultants and agents only if and to the extent necessary to carry out their respective responsibilities under this Agreement or in accordance with the exercise of their rights under this Agreement, and such disclosure shall be limited to the maximum extent possible consistent with such responsibilities and rights. Except as set forth in the foregoing sentence, no Party shall disclose Confidential Information of the other to any Third Party without the other's prior written consent. In all events, however, any and all disclosure to a Third Party (or to any such Affiliate or Sublicensee) shall be pursuant to the terms of a non-disclosure/nonuse agreement no less restrictive than this Article VII. The Party which disclosed Confidential Information of the other to any Third Party (or to any such Affiliate or Sublicensee) shall be responsible and liable for any disclosure or use by such Third Party, Affiliate or Sublicensee (or its disclosees) which would have violated this Agreement if committed by the Party itself. No Party shall use Confidential Information of the other except as expressly allowed by and for the purposes of this Agreement. Each Party shall take such action to preserve the confidentiality of each other's Confidential Information as it would customarily take to preserve the confidentiality of its own Confidential Information (but in no event less than a reasonable standard of care). Upon expiration or termination of this Agreement, each Party, upon the other's request, shall return or destroy (at Disclosing Party's discretion) all the Confidential Information disclosed to the other Party pursuant to this Agreement, including all copies and extracts of documents, within 60 days after the request, except for one archival copy (and such electronic copies that exist as part of the Party's comput

- **7.3 Exceptions**. The non-use and non-disclosure obligations set forth in this Article VII shall not apply to any Confidential Information, or portion thereof, that the Receiving Party can demonstrate by competent evidence:
 - (a) at the time of disclosure is in the public domain;
 - (b) after disclosure, becomes part of the public domain, by publication or otherwise, through no fault of the Receiving Party or its disclosees;
 - (c) is made available to the Receiving Party by an independent Third Party without obligation of confidentiality; provided, however, that to the Receiving Party's knowledge, such information was not obtained by said Third Party, directly or indirectly, from the Disclosing Party hereunder; or
 - (d) is independently developed by an employee of the Receiving Party not accessing or utilizing the Disclosing Party's information.

In addition, the Receiving Party may disclose information that is required to be disclosed by law, by a valid order of a court or by order or regulation of a governmental agency including but not limited to, regulations of the SEC or in the course of arbitration or litigation; provided, however, that in all cases the Receiving Party shall give the other party prompt notice of the pending disclosure and make a reasonable effort to obtain, or to assist the Disclosing Party in obtaining, a protective order or confidential-treatment order preventing or limiting (to the greatest possible extent and for the longest possible period) the disclosure and/or requiring that the Confidential Information so disclosed be used only for the purposes for which the law or regulation required, or for which the order was issued.

- 7.4 Third Party Information. The Parties acknowledge that the defined term "Confidential Information" shall include not only a Disclosing Party's own Confidential Information but also Confidential Information of a Third Party which is in the possession of a Disclosing Party. The Parties agree not to disclose to the other any Confidential Information of a Third Party which is in the possession of such Party, unless the other has given an express prior written consent (which specifies the owner of such Confidential Information) to receive such particular Confidential Information.
- 7.5 Press Release Announcing the Execution of the License Agreement and Related Disclosures. Either Party may make an initial press release announcing the execution of this Agreement, including any matter covered by this Agreement, and the Development or Commercialization of Licensed Products, but such Party shall provide the text of such planned disclosure to the other Party sufficiently in advance of the scheduled disclosure to afford such other Party a reasonable opportunity to review and comment upon the proposed text and the timing of such disclosure, and shall consider all reasonable comments of the other Party regarding such disclosure. (Provided, that no Party shall use the trademark or logo of the other Party, its Affiliates or their respective employee(s) in any publicity, promotion, news release or public disclosure relating to this Agreement or its subject matter, except as may be required by Law or required by the rules of an applicable US national securities exchange or except with the prior express written permission of such other Party, such permission not to be unreasonably withheld.)

ARTICLE VIII REPRESENTATIONS, WARRANTIES AND COVENANTS

- **8.1** Representations and Warranties. (a) TGTX represents and warrants to CTI, and (b) CTI represents to TGTX, in each case as of the Effective Date:
 - (a) Such Party is a corporation duly organized and validly existing under the Laws of the jurisdiction of its incorporation;
 - (b) Such Party has all right, power and authority to enter into this Agreement, and to perform its obligations under this Agreement;
 - (c) Such Party has taken all action necessary to authorize the execution and delivery of this Agreement and the performance of its obligations under this Agreement;
 - (d) This Agreement is a legal and valid obligation of such Party, binding upon such Party and enforceable against such Party in accordance with the terms of this Agreement, except as enforcement may be limited by applicable bankruptcy, fraudulent conveyance, insolvency, reorganization, moratorium and other Laws relating to or affecting creditors' rights generally and by general equitable principles;
 - (e) To the best of such party's knowledge, the execution, delivery and performance of this Agreement by such Party does not conflict with, breach or create in any Third Party the right to accelerate, terminate or modify any agreement or instrument to which such Party is a party or by which such Party is bound;
 - (f) To the best of such party's knowledge, all consents, approvals and authorizations from all governmental authorities or other Third Parties required to be obtained by such Party in connection with the execution and delivery of this Agreement have been obtained; and the execution, delivery and performance of this Agreement by such Party does not violate any Law of any Governmental Body having authority over such Party;
 - (g) No person or entity has or will have, as a result of the execution and delivery of or as a result of the transactions contemplated by this Agreement, any right, interest or valid claim against or upon such Party for any commission, fee or other compensation as a finder or broker because of any act by such Party or its Affiliates, agents or Sublicensees; and
 - (h) To the best of such party's knowledge, no agreement between it and any Third Party is in conflict with the rights granted to any other party pursuant to this Agreement.

8.2 Reserved.

- **8.3 Disclaimer**. Notwithstanding the representations and warranties set forth in this Article VIII, TGTX acknowledges and accepts the risks inherent in attempting to Develop and Commercialize any pharmaceutical product. There is no implied representation that the Licensed Products can be successfully Developed or Commercialized.
- 8.4 CTI MAKES NO WARRANTY, EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY OR OF FITNESS FOR A PARTICULAR PURPOSE WITH RESPECT TO ANY PATENT, TRADEMARK, SOFTWARE, NON-PUBLIC OR OTHER INFORMATION, DFCI MATERIALS, DFCI ANTIBODIES, KNOW-HOW, OR TANGIBLE RESEARCH PROPERTY, LICENSED OR OTHERWISE PROVIDED TO TGTX HEREUNDER AND HEREBY DISCLAIMS THE SAME.
- 8.5 CTI DOES NOT WARRANT THE VALIDITY OF THE LICENSED PATENTS LICENSED HEREUNDER AND MAKES NO REPRESENTATION WHATSOEVER WITH REGARD TO THE SCOPE OF THE LICENSED PATENTS OR THAT SUCH LICENSED PATENTS MAY BE EXPLOITED BY TGTX, AFFILIATE OR SUBLICENSEE WITHOUT INFRINGING OTHER PATENTS. CTI MAKES NO REPRESENTATION THAT DFCI ANTIBODIES, CTI ANTIBODIES OR THE METHODS USED IN MAKING OR USING SUCH DFCI ANTIBODIES OR CTI ANTIBODIES ARE FREE FROM LIABILITY FOR PATENT INFRINGEMENT.

ARTICLE IX INDEMNIFICATION; LIMITATION OF LIABILITY; INSURANCE

Indemnification and Defense.

- 9.1 TGTX shall indemnify, defend and hold harmless (i) DFCI and its trustees officers, medical and professional staff, employees, and agents and their respective successors, heirs and assigns and (ii) CTI and its directors, officers, employees, agents and contractors (the "CTI Indemnitees"), against any liability, damage, loss or expense (including reasonable attorneys' fees and expenses of litigation) incurred by or imposed upon the CTI Indemnitees, or any one of them, in connection with any claims, suits, actions, demands or judgments_arising out any theory of product liability (including but not limited to action in the form of tort, warranty, strict liability) concerning any product, process or service relating to, or developed by TGTX, its Affiliates or Sublicensees pursuant to (a) any right or license granted under this Agreement or (b) arising out of any other activities to be carried out by TGTX pursuant to this agreement. TGTX's indemnification under Section 9.1 does not apply to any liability, damage, loss or expense to the extent that it is attributable to (x) the grossly negligent activities of the CTI Indemnitees, or (y) the intentional wrongdoing or intentional misconduct of the CTI Indemnitees TGTX shall, at its own expense, provide attorneys reasonably acceptable to CTI to defend against any actions brought or filed against any party indemnified hereunder with respect to the subject of indemnity contained herein, whether or not such actions are rightfully brought.
- 9.2 CTI shall indemnify, defend and hold harmless TGTX and its directors, officers, employees, agents and contractors (the "TGTX Indemnitees"), against any liability, damage, loss or expense (including reasonable attorneys' fees and expenses of litigation) incurred by or imposed upon the TGTX Indemnitees, or any one of them, in connection with any claims, suits, actions, demands or judgments_arising out any theory of product liability (including but not limited to action in the form of tort, warranty, strict liability) concerning (a) any product, process or service relating to, or developed by CTI, its Affiliates or Sublicensees pursuant to the License Agreement or (b) any other activities to be carried out by CTI pursuant to this agreement. CTI's indemnification under Section 9.1 does not apply to any liability, damage, loss or expense to the extent that it is attributable to (x) the grossly negligent activities of the TGTX Indemnitees, or (y) the intentional wrongdoing or intentional misconduct of the TGTX Indemnitees. CTI shall, at its own expense, provide attorneys reasonably acceptable to DFCI and TGTX to defend against any actions brought or filed against any party indemnified hereunder with respect to the subject of indemnity contained herein, whether or not such actions are rightfully brought.
- 9.3 If any such action is commenced or claim made or threatened against a DFCI Indemnitee or CTI Indemnitee (collectively, "Indemnitees") as to which the other Party (the "Indemnifying Party") is obligated to indemnify it (them) or hold it (them) harmless, the Indemnitee shall promptly notify Indemnifying Party of such event. Indemnifying Party shall assume the defense of, and may settle, that part of any such claim or action commenced or made against an Indemnitee which relates to the Indemnifying Party's indemnification and CTI may take such other steps as may be necessary to protect it. Indemnifying Party will not be liable to Indemnitees on account of any settlement of any such claim or litigation affected without Indemnifying Party's consent. The right of Indemnifying Party to assume the defense of any action is limited to that part of the action commenced against Indemnitees that relates to Indemnifying Party's obligation of indemnification and holding harmless.
- 9.4 TGTX shall require any Affiliates or Sublicensee(s) to indemnify, hold harmless and defend DFCI and CTI under the same terms set forth in Sections 9.1 9.4.
- 9.5 Insurance. At such time as any product, process or service relating to, or developed pursuant to, this Agreement is being commercially distributed or sold (other than for the purpose of obtaining regulatory approvals) by TGTX or by a Sublicensee, Affiliate or agent of TGTX, TGTX shall, at its sole cost and expense, procure and maintain policies of commercial general liability insurance in amounts not less than \$2,000,000 per incident and \$2,000,000 annual aggregate and naming the Indemnitees as additional insureds. Such commercial general liability insurance must provide (a) product liability coverage and (b) contractual liability coverage for TGTX's indemnification under Sections 9.1 through 9.5 of this Agreement. If TGTX elects to self-insure all or part of the limits described above (including deductibles or retentions which are in excess of \$250,000 annual aggregate), such self-insurance program must be acceptable to the CTI, DFCI and the DFCI's associated Risk Management Foundation. The minimum amounts of insurance coverage required under these provisions may not be construed to create a limit of TGTX's liability with respect to its indemnification obligation under Sections 9.1 through 9.5 of this Agreement.
- 9.6 TGTX shall provide CTI with written evidence of such insurance upon request of CTI. TGTX shall provide CTI with written notice at least fifteen (15) days prior to the cancellation, non-renewal or material change in such insurance; if TGTX does not obtain replacement insurance providing comparable coverage within such fifteen (15) day period, CTI has the right to terminate this Agreement effective at the end of such fifteen (15) day period without any notice or additional waiting periods.
- 9.7 TGTX shall maintain such comprehensive general liability insurance beyond the expiration or termination of this Agreement during (a) the period that any product, process, or service, relating to, or developed pursuant to, this Agreement is being commercially distributed or sold (other than for the purpose of obtaining regulatory approvals) by TGTX or by a Sublicensee, Affiliate or agent of TGTX and (b) a reasonable period after the period referred to in 9.8 (a) above which in no event shall be less than fifteen (15) years.
- 9.8 TGTX shall require any of its Affiliates or Sublicensee(s) to, maintain insurance in favor of CTI, DFCI and the Indemnitees under the same terms set forth in Sections 9.5 9.7 of this Agreement.

ARTICLE X TERM AND TERMINATION

- 10.1 Term. The term of this Agreement shall commence on the Effective Date and, unless earlier terminated as provided in this Article X, shall continue in full force and effect, on a country-by-country and Licensed Product-by-Licensed Product basis until the Royalty Term in such country with respect to such Licensed Product expires, at which time this Agreement shall expire in its entirety with respect to such Licensed Product in such country. (The "Term" shall mean the period from the Effective Date until the earlier of termination of this Agreement as provided in this Article X or expiration of this Agreement upon the expiration of the last-to-expire Royalty Term.) The Parties confirm that subject to the foregoing sentence, this Agreement shall not be terminated or invalidated by any future determination that any or all of the Licensed Patents have expired or been invalidated.
- 10.2 Termination by CTI. CTI has the right to immediately terminate this Agreement, the extension of rights (if such termination occurs while TGTX is an Affiliate of CTI), and all licenses granted hereunder (if such failure occurs after the time TGTX ceases to be an Affiliate of CTI), or at CTI's option to convert the exclusive license granted in Article 2.1 to a non-exclusive license (if such failure occurs after the time TGTX ceases to be an Affiliate of CTI) in accordance with Section 3.6, by providing TGTX with written notice of such, upon the occurrence of any of the following events.
 - (a) TGTX's Board of Director's has agreed that TGTX will cease to carry on its business with respect to Licensed Products.
 - (b) TGTX fails to pay when due any undisputed royalty or other undisputed payment that has become due and is payable under Article V of this Agreement and has not cured the default by making the required payment, together with interest due, within ninety days of receiving a written notice of default from CTI requesting such payment.
 - (c) An officer of TGTX is convicted of a felony relating to the manufacture, use, sale or importation of Licensed Products.
 - (d) TGTX materially breaches any other provision of this Agreement (including but not limited to due diligence obligations under Article III and insurance obligations under Section 9.7 Section 9.10), unless TGTX has cured the breach within ninety days of receiving written notice from CTI specifying the nature of the breach; provided, however, that the due diligence obligations shall be determined on a Licensed Product by Licensed Product basis
- 10.3 Termination for insolvency. TGTX or CTI may terminate this Agreement immediately upon written notice, with no further notice obligation or opportunity to cure, if TGTX or CTI shall become insolvent, shall make an assignment for the benefit of creditors, or shall have a petition in bankruptcy filed for or against it (which is not dismissed within 60 days of such filing).
- 10.4 Notwithstanding Sections 10.2 and 10.3, in the event of a good-faith dispute as to whether any alleged breach, default, failure or any other act or omission gives rise to a right of termination under this Agreement, is in fact a breach, default, failure or other act or omission that gives rise to a right of termination under this Agreement, termination of this Agreement in respect of such alleged breach, default, failure or other act or omission shall not take effect unless and until (y) such dispute is resolved in accordance with Section 10.7 below in favor of the Party alleging such breach, default, failure or other act or omission or (z) the non-terminating Party's denial that the alleged breach, default, failure or other act or omission giving rise to a right of termination hereunder ceases to be in good faith.
- 10.5 Termination by TGTX. TGTX has the right to terminate this Agreement without cause by giving CTI one hundred and eighty days prior written notice in whole or on a Licensed Product by Licensed Product basis. Any milestones achieved by TGTX during this one hundred and eighty-day period will be due and payable to CTI.

10.6 Effect of Termination

- (a) No release. Upon termination of this Agreement for any reason, nothing in this Agreement may be construed to release either party from any obligation that matured prior to the effective date of the termination.
- (b) **Survival.** The provisions of Section 6.1(a) (patent expenses) Article V (Financial Provisions), Section 3.1.2 (Publicity –paragraph 10.6(c) (Inventory), Article IX (Indemnification), Sections 9.7 9.10 (Insurance), Article VIII (Representations and Warranties) and Section 10.7 (Dispute Resolution) survive termination or expiration of this Agreement.
- (c) **Inventory.** TGTX, any Affiliate(s) and any Sublicensees whose sublicenses are not converted as provided in paragraph 10.6(d) below, may, after the effective date of termination, sell all Licensed Products that are in inventory as of the date of written notice of termination, and complete and sell Licensed Products which the licensed entity(ies) can reasonably demonstrate were in the process of manufacture as of the date of written notice of termination, provided that TGTX shall pay to CTI the royalties thereon as required by Article V and shall submit the reports required by Section 5.10 on the sales of Licensed Products.

- (d) **Sublicenses.** Any Sublicenses will terminate contemporaneously with this Agreement; provided, however, that any Sublicenses that are not in default under the sublicense agreement shall, upon DFCI's and CTI's written approval, survive and remain in full force and effect so long as the Sublicensee agrees to be bound by all of the provisions of this Agreement, if not otherwise already provided for in the sublicense agreement. Such approval by DFCI and CTI shall not be unreasonably withheld and shall not require the payment of additional consideration.
- (e) If (i) this Agreement is in effect at the time of the termination of the License Agreement and (ii) TGTX is not an Affiliate of CTI at such time then, upon the written approval by DFCI, this Agreement survive and remain in full force and TGTX hereby agrees to be bound by the terms of the License Agreement pursuant to Section 10.6(d) of the License Agreement. If DFCI does not approve such survival, then this Agreement shall terminate upon termination of the License Agreement. Such approval by DFCI shall not be unreasonably withheld and shall not require the payment of additional consideration.
- (f) Pursuant to the License Agreement, TGTX is deemed an Affiliate of CTI, and thus at the time the License Agreement is terminated, this Agreement shall automatically terminate at such time; provided, that pursuant to Section 2.5, TGTX shall have the right to cure any breach and that CTI will not voluntarily terminate the License Agreement with TGTX's prior written consent.

10.7 Dispute Resolution.

- (a) **Negotiation between the Parties.** The parties shall first attempt to resolve any controversy that arises from this Agreement, or claim for breach of the Agreement, by good faith negotiations, first between their respective business development representatives and then, if necessary, between senior representatives for the Parties.
- (b) **Non-Binding Mediation.** If the controversy or claim cannot be settled through good faith negotiation between the parties, the parties agree first to try in good faith to settle their dispute by non-binding mediation under the Mediation Rules of the American Arbitration Association, before resorting to arbitration, litigation or other dispute resolution procedure.

ARTICLE XI MISCELLANEOUS PROVISIONS

11.1 Relationship of the Parties. Nothing in this Agreement is intended or shall be deemed to constitute a partnership, agency, joint venture or employer-employee relationship between the Parties. No Party shall have any right or authority to commit or legally bind any other Party in any way whatsoever including, without limitation, the making of any agreement, representation or warranty and each Party agrees to not purport to do so.

11.2 Assignment.

- (a) Any assignment not in accordance with this Section 11.2 shall be void.
- (b) No assignment shall relieve the assigning Party of any of its responsibilities or obligations hereunder.
- (c) TGTX may not transfer or assign its rights or licenses or delegate its obligations under this Agreement, in whole or in part, by operation of law or otherwise, to any Third Party without the prior written consent of CTI, which consent shall not be unreasonably withheld, conditioned or delayed; *provided that,* notwithstanding the foregoing, TGTX may, without such consent, assign its rights or licenses and/or delegate its obligations under this Agreement to (i) an Affiliate or (ii) a Third Party in connection with a Sale Event (and for the avoidance of doubt, at such time the extension of rights set forth in Section 2.5 shall terminate and the licenses granted to TGTX in Section 2 shall become effective). As a condition to any permitted assignment hereunder, the assignee must expressly assume, in a writing delivered to CTI and signed by a duly authorized officer of the assignee (and in a form reasonably acceptable to CTI) all of TGTX's obligations under this Agreement, whether arising before, at or after the assignment.
- 11.3 Further Actions. Each Party agrees to execute, acknowledge and deliver such further instruments and to do all such other acts as may be necessary or appropriate in order to carry out the purposes and intent of this Agreement.

- 11.4 Force Majeure. No Party shall be liable to any other Party or be deemed to have breached or defaulted under this Agreement for failure or delay in the performance of any of its obligations under this Agreement (other than obligations for the payment of money) for the time and to the extent such failure or delay is caused by or results from acts of God, earthquake, riot, civil commotion, terrorism, war, strikes or other labor disputes, fire, flood, failure or delay of transportation, omissions or delays in acting by a governmental authority, acts of a government or an agency thereof or judicial orders or decrees or restrictions or any other like reason which is beyond the control of the respective Party. The Party affected by force majeure shall provide the other Party with full particulars thereof as soon as it becomes aware of the same (including its best estimate of the likely extent and duration of the interference with its activities), and shall use Commercially Reasonable Efforts to overcome the difficulties created thereby and to resume performance of its obligations hereunder as soon as practicable, and the time for performance shall be extended for a number of days equal to the duration of the force majeure.
- 11.5 Entire Agreement of the Parties; Amendments. This Agreement and the Schedules hereto constitute and contain the entire understanding and agreement of the Parties respecting the subject matter hereof and cancel and supersede any and all prior or contemporaneous negotiations, correspondence, understandings and agreements between the Parties, whether oral or written, regarding such subject matter (provided, that any and all previous nondisclosure/nonuse obligations are not superseded and remain in full force and effect in addition to the nondisclosure/nonuse provisions hereof). Each Party acknowledges that it has not relied, in deciding whether to enter into this Agreement on this Agreement's expressly stated terms and conditions, on any representations, warranties, agreements, commitments or promises which are not expressly set forth within this Agreement. No modification or amendment of any provision of this Agreement shall be valid or effective unless made in a writing referencing this Agreement and signed by a duly authorized officer of each Party.
- 11.6 Governing Law. This Agreement shall be governed by and interpreted in accordance with the laws of the State of New York, excluding application of any conflict of laws principles.
- 11.7 Notices and Deliveries. Any notice, request, approval or consent required or permitted to be given under this Agreement shall be in writing and shall be deemed to have been sufficiently given if and only if delivered in person, by email or by express courier service to the Party to which it is directed at its physical or email address shown below or such other physical or email address as such Party shall have last given by such written notice to the other Party.

If to CTI, addressed to:

Checkpoint Therapeutics, Inc. 2 Gansevoort Street, 9th Floor New York, NY 10014 Attention: James F. Oliviero, CEO Email: jfo@checkpointtx.com

If to TGTX, addressed to:

TG Therapeutics, Inc. 2 Gansevoort Street, 9th Floor New York, NY 10014 Attention: Sean Power, CFO Email: sp@tgtxinc.com

- 11.8 Waiver. No waiver of any provision of this Agreement shall be valid or effective unless made in a writing referencing this Agreement and signed by a duly authorized officer of the waiving Party. A waiver by a Party of any of the terms and conditions of this Agreement in any instance shall not be deemed or construed to be a waiver of such term or condition for the future, or of any other term or condition hereof.
- 11.9 Rights and Remedies are Cumulative. Except to the extent expressly set forth herein, all rights, remedies, undertakings, obligations and agreements contained in or available upon violation of this Agreement shall be cumulative and none of them shall be in limitation of any other remedy or right authorized in law or in equity, or any undertaking, obligation or agreement of the applicable Party.

- 11.10 Severability. This Agreement is severable. When possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Agreement is held to be to any extent prohibited by or invalid under applicable Law, such provision will be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of this Agreement (or of such provision). The Parties shall make a good faith effort to replace the invalid or unenforceable provision with a valid one which in its economic effect is most consistent with the invalid or unenforceable provision.
- 11.11 Third Party Beneficiaries. Except for the rights of Indemnified Parties pursuant to Article IX hereof and the rights of Sublicensees set forth in Sections 2.3 and 10.6(d), the terms and provisions of this Agreement are intended solely for the benefit of each Party hereto and their respective successors or permitted assigns and it is not the intention of the Parties to confer third-party beneficiary rights upon any other person, including without limitation Sublicensees. The enforcement of any obligation of CTI under this Agreement shall only be pursued by TGTX or such Indemnified Party, and not Sublicensees (except as set forth in Sections 2.3 and 10.6(d)).
- 11.12 No Implied License. No right or license is granted to TGTX hereunder by implication, estoppel, or otherwise to any know-how, patent or other intellectual property right owned or controlled by CTI or its Affiliates, except by an express license granted hereunder. No right or license is granted to CTI hereunder by implication, estoppel, or otherwise to any know-how, patent or other intellectual property right owned or controlled by TGTX or its Affiliates, except by an express license granted hereunder.
- 11.13 No Right of Set-Off. Except as expressly provided in Article 5 of this Agreement, TGTX shall not have a right to set-off any royalties, milestones or other amount due to CTI under this Agreement against any damages incurred by TGTX for a breach by CTI of this Agreement.
- 11.14 Equitable Relief. Each Party recognizes that the covenants and agreements herein and their continued performance as set forth in this Agreement are necessary and critical to protect the legitimate interests of the other Party, that the other Party would not have entered into this Agreement in the absence of such covenants and agreements and the assurance of continued performance as set forth in this Agreement, and that a Party's breach or threatened breach of such covenants and agreements may cause the opposed Party irreparable harm and significant injury, the amount of which will be extremely difficult to estimate and ascertain, thus potentially making any remedy at law or in damages inadequate. Therefore, each Party agrees that an opposed Party shall be entitled to seek specific performance, an order restraining any breach or threatened breach of Article VII and all other provisions of this Agreement, and any other equitable relief (including but not limited to temporary, preliminary and/or permanent injunctive relief). This right shall be in addition to and not exclusive of any other remedy available to such other Party at law or in equity.
- 11.15 **Interpretation.** The language used in this Agreement is the language chosen by the Parties to express their mutual intent, and no provision of this Agreement shall be interpreted for or against a Party because that Party or its attorney drafted the provision.
- 11.16 Construction. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." All references herein to Articles, Sections and Schedules shall be deemed references to Articles and Sections of, and Schedules to, this Agreement unless the context shall otherwise require.
- 11.17 Counterparts. This Agreement may be executed in counterparts, each of which will be deemed an original, and all of which together will be deemed to be one and the same instrument. A facsimile or a portable document format (.pdf) copy of this Agreement, including the signature pages, will be deemed an original.

[the remainder of this page has been left blank intentionally]

CHECKPOINT THERAPEUTICS, I	INC.		
By: /s/ James Oliviero			
Name: James Oliviero			
Title: CEO			
TG THERAPEUTICS, INC.			
By: /s/ Michael S. Weiss	_		
Name: Michael S. Weiss	_		
Title: CEO			

IN WITNESS WHEREOF, the Parties have caused this Amended and Restated Collaboration Agreement to be executed and delivered by their respective duly authorized officers as of the day and year first above written.

Exhibit 10.2

MASTER SERVICES AGREEMENT

between

SAMSUNG BIOLOGICS CO., LTD.

and

TG THERAPEUTICS, INC.

Table of Contents		
SECTION 1	DEFINITIONS	3
SECTION 2	RELATED AGREEMENTS AND EXHIBITS	8
SECTION 3	MANAGEMENT OF SERVICE	9
SECTION 4	SERVICES	10
SECTION 5	SERVICE DESCRIPTIONS	12
SECTION 6	CHANGES TO THE SPECIFICATIONS, ANALYTICAL METHODS, MANUFACTURING PROCESS, FACILITY OR EQUIPMENT	16
SECTION 7	REGULATORY APPROVALS AND INSPECTIONS	17
SECTION 8	QUALITY COMPLIANCE	18
SECTION 9	CONSIDERATION AND PAYMENT TERMS	19
SECTION 10	CONFIDENTIALITY	20
SECTION 11	OWNERSHIP OF MATERIALS AND INTELLECTUAL PROPERTY	21
SECTION 12	WARRANTIES	22
SECTION 13	INDEMNIFICATION	23
SECTION 14	DISCLAIMER OF CONSEQUENTIAL DAMAGES; LIMITATION OF LIABILITY	23
SECTION 15	TERM AND TERMINATION OF AGREEMENT	24
SECTION 16	ARBITRATION	26
SECTION 17	MISCELLANEOUS	27

MASTER SERVICES AGREEMENT

THIS MASTER SERVICES AGREEMENT (this "MSA") is made and entered into as of the date of last signature below (the "Effective Date") by and between TG Therapeutics, Inc., a Delaware corporation having its principal place of business at 2 Gansevoort St., 9th Floor, New York, NY 10014 ("Client"), and Samsung BioLogics Co., Ltd., a company with offices at 300, Songdo bio-daero, Yeonsu-gu, Incheon, 21987, Republic of Korea ("SBL"). Client and SBL are sometimes referred to herein individually as a "Party" and collectively as the "Parties".

WHEREAS, Client and SBL wish to enter into a business relationship whereby SBL will provide Client with certain biologics manufacturing services;

NOW, THEREFORE, in consideration of the mutual promises, covenants and agreements hereinafter set forth and for other valuable consideration, the Parties agree as follows:

SECTION 1 DEFINITIONS

- 1.1 "Acceptance Procedure" shall mean the review of the Batch Related Documents and test(s) of the Product, if necessary, to verify that the Product delivered meets the Specifications and complies with Regulatory Authority requirements, conducted by Client after SBL's release of the Product, to determine whether to accept the same, in accordance with the applicable QAG as modified from time to time by written agreement between the Parties.
- 1.2 "Affected Party" is defined in Section 17.3.
- 1.3 "Affiliate" shall mean any corporation, company, partnership or other entity which directly or indirectly, controls, is controlled by or is under common control with either Party hereto. A corporation or other entity shall be regarded as controlling another corporation or other entity if it owns or directly or indirectly controls more than fifty percent (50%) of the voting stock or other ownership interest of the corporation or other entity, or if possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of the corporation or other entity or the power to elect or appoint more than fifty percent (50%) of the members of the governing body of the corporation or other entity.
- 1.4 "Applicable Laws" shall mean any and all applicable laws of any jurisdiction which are applicable to any of the Parties in carrying out activities described in this MSA or any PSAs that may be in effect from time to time, and shall include all statutes, enactments, acts of legislature, laws, ordinances, rules, regulations, notifications, guidelines, directions, directives and orders of any Regulatory Authority, statutory authority, stock exchange, securities regulatory agency, tribunal, board, or court or any central or state government or local authority or other governmental entity in such jurisdictions.
- 1.5 "Background IP" shall mean Intellectual Property which has been owned and/or controlled by a Party prior to the Effective Date or outside, or not relating to, the performance of the MSA and any pertinent PSA.
- 1.6 "Batch" shall mean Product Manufactured by SBL from a single run of the applicable Manufacturing Process.
- 1.7 "Batch Record" is defined in the applicable QAG.
- 1.8 "Batch Related Documents" means Manufacturing Documentation in support of the SBL's release of a Product.
- $\underline{1.9}$ "Binding Year" shall be defined in the applicable PSA.
- 1.10 "Business Day" shall mean a day on which commercial banks are open for business in both the Republic of Korea and New York, New York.
- 1.11 "Cell Line" shall mean the aliquot of cells supplied to SBL by Client to perform the Services and their progeny.
- $\underline{1.12}$ "Certificate of Analysis" is defined in the applicable QAG.
- 1.13 "Certificate of Compliance" is defined in the applicable QAG.
- 1.14 "Change" is defined in Section 6.1.
- 1.15 "Client" is defined in the preamble.

- 1.16 "Client Materials" shall mean Client reagents and other materials supplied by Client or its third party supplier to be used in the Service hereunder, as each is further defined in the PSA and/or applicable QAG. In the case of a Drug Product PSA, Client Materials shall also include Drug Substance and/or other active pharmaceutical ingredients, which may or may not be Manufactured by SBL.
- 1.17 "Client Technology" shall mean know-how, technology, research and other information of Client including and relating to the Manufacturing Process, analytical methods, quality control analysis, specifications, transportation and storage requirements provided by Client to SBL in connection with this MSA and applicable PSA.
- 1.18 "Clinical Product" shall mean a Drug Substance or Drug Product which is Manufactured by SBL pursuant to a PSA and which is to be used by Client in a research study or studies that investigate the safety of human use or prospectively assign human participants or groups of humans to one or more health-related interventions to evaluate the effects on health outcomes.
- 1.19 "Commercial Product" shall mean a Drug Substance or Drug Product which is Manufactured by SBL which is intended for commercial sale and use by humans and for importation or exportation into countries or regions designated in each PSA.
- 1.20 "Commercially Reasonable Efforts" shall mean with respect to an activity to be carried out by a Party, the carrying out of such activity in a diligent manner, which, in the case of SBL, requires using the efforts of sufficient numbers of personnel appropriately qualified by experience and training, in the application of such resources as would be reasonably applied by a qualified manufacturer of biologics for human use adhering to recognized standards in the manufacture and testing of such products, or, in the case of Client, using such efforts as would be typically be used in the biopharmaceutical industry by companies of comparable resources and expertise. "Commercially Reasonable Efforts" requires prompt assignment of responsibility for such task or activity to specific qualified employee(s) and allocation of resources designed to advance progress with respect to such task or activity but does not require the taking of actions which would require either Party to violate Applicable Laws or break any existing contractual commitments with third parties which were entered into prior to the Effective Date, the performance of which at such time was not reasonably foreseeable to be in conflict, or otherwise inconsistent, with such Party's obligations under the MSA or any PSA.
- 1.21 "Common Raw Materials" is defined in Section 5.3.1.
- 1.22 "Confidential Information" is defined in Section 10.1.
- 1.23 "Control" (including, with correlative meanings, "Controlled") means possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or other ownership interest, by contract or otherwise) of that person or entity and/or the ownership of more than 50% of the voting shares of that person or entity.
- 1.24 "Core Team" is defined in Section 3.3.
- 1.25 "current Good Manufacturing Practices" or "cGMP" shall mean current good manufacturing practices and regulations applicable to the Manufacture of Product that are promulgated by any Regulatory Authority, including as promulgated under and in accordance with (i) the U.S. Federal Food, Drug and Cosmetic Act, Title 21 of the U.S. Code of Federal Regulations, Parts 210, 211, 600, 601 and 610, (ii) relevant EU legislation, including European Directive 2003/94/EC or national implementations of that Directive, (iii) relevant guidelines, including the EU Guidelines for Good Manufacturing Practices for Medicinal Products (Eudralex Vol. 4 and Annexes thereto), (iv) International Conference on Harmonisation Good Manufacturing Practice Guide for Active Pharmaceuticals Ingredients and (v) and any analogous set of regulations, guidelines or standards as defined, from time to time, by any relevant Regulatory Authority having jurisdiction over the development, manufacture or commercialization of the Product, as applicable, in each case as in effect as of the date such manufacturing for the Product are or were conducted.
- 1.26 "Damages" means any damages, costs, expenses, fines, penalties (including reasonable attorneys' fees and costs), losses and liabilities.
- 1.27 "Disclosing Party" is defined in Section 10.1.

- 1.28 "Drug Product" means a finished or intermediate dosage form that contains a Drug Substance, generally, but not necessarily, in association with one or more other ingredients.
- 1.29 "Drug Substance" means an active ingredient that is intended to furnish pharmacological activity or other direct effect in the diagnosis, cure, mitigation, treatment, or prevention of disease or to affect the structure or any function of the human body, but does not include intermediates used in the synthesis of such ingredient.
- 1.30 "Effective Date" is defined in the preamble.
- 1.31 "EMA" shall mean the European Medicines Agency, or any successor agency.
- 1.32 "Engineering Batch" shall mean a Batch that is intended to demonstrate the transfer of the Manufacturing Process to the Facility. After Manufacture, Client shall have the right to make whatever further use of non-cGMP Engineering Batches as it shall determine, provided that Client pays for such Batches according to this MSA, such use is not for human use and does not violate any Applicable Laws. SBL makes no warranty that Engineering Batches will meet cGMP or the Specifications. If SBL determines that an Engineering Batch does meet cGMP and the Specifications, it will release such Engineering Batch as a cGMP Batch, and thereafter Client shall be entitled to use such Engineering Batch for human use.
- 1.33 "Facility" shall mean one or more of the manufacturing facilities of SBL where the Services shall be performed, located at 300, Songdo bio-daero, Yeonsugu, Incheon, 21987, Republic of Korea.
- 1.34 "FDA" shall mean the United States Food and Drug Administration or any successor agency thereto.
- 1.35 "Firm Period" shall be defined in the applicable PSA.
- 1.36 "Force Majeure Event" is defined in Section 17.3.
- 1.37 "Implementation Plan and Budget" is defined in Section 6.2(b).
- 1.38 "Indemnified Party" is defined in Section 13.3.
- 1.39 "Indemnifying Party" is defined in Section 13.3.
- 1.40 "Intellectual Property" is shall mean (a) patents, patent rights, provisional patent applications, patent applications, designs, registered designs applications, industrial designs, industrial design applications and industrial design registrations, including any and all divisions, continuations, continuations-in-part, extensions, restorations, substitutions, renewals, registrations, revalidations, reexaminations, reissues or additions, including supplementary certificates of protection, of or to any of the foregoing items; (b) copyrights, copyright registrations, copyright applications, original works of authorship fixed in any tangible medium of expression, including literary works (including all forms and types of computer software, including all source code, object code, firmware, development tools, files, records and data, and all documentation related to any of the foregoing), pictorial and graphic works; (c) trade secrets, technology, developments, discoveries and improvements, know- how, proprietary rights, formulae, confidential and proprietary information, technical information, techniques, inventions, designs, drawings, procedures, processes, models, formulations, manuals and systems, whether or not patentable or copyrightable, including all biological, chemical, biochemical, toxicological, pharmacological and metabolic material and information and data relating thereto and formulation, clinical, analytical and stability information and data which have actual or potential commercial value and are not available in the public domain; (d) trademarks, trademark registrations, trademark applications, service marks, service mark registrations, service mark applications, business marks, brand names, trade names, trade dress, names, logos and slogans, Internet domain names, and all goodwill associated therewith; and (e) all other intellectual property or proprietary rights, in each case whether or not subject to statutory registration or protection.

- 1.41 "Joint Steering Committee" or "JSC" is defined in Section 3.2.1.
- 1.42 "Manufacturing" or to "Manufacture" shall mean the manufacturing of the Product, and any services relating to such manufacturing, including, but not limited to, testing, quality control, documentations, archiving, handling, storage and packaging, release and delivery of the Product, to be performed by SBL at the Facility under the MSA and any applicable PSA.
- 1.43 "Manufacturing Documentation" shall mean data acquired and generated, documents and records describing or otherwise related to the Manufacturing Process including, without limitation: documents and records consisting of or containing process descriptions, requirements and specifications; Client Materials and Specifications; analytical methods, process trend and variability data; validations protocols and reports; process development reports; Batch Records; Batch Related Documents, and SOPs, including, without limitation, SOP's for the Raw Materials handling, the Manufacturing operations, equipment operation, inprocess, final Product and stability quality control testing, quality assurance, validation, storage and shipping.
- 1.44 "Manufacturing Process" shall mean the mutually agreed production process and analytical methods for the Manufacturing of the Product pursuant to the applicable PSA, as summarily described in the applicable QAG and as described in the Manufacturing Documentation, as such process may be changed from time to time in accordance with the MSA.
- 1.45 "Manufacturing Process Transfer" means the Commercially Reasonable Efforts of the parties undertaken pursuant to the Manufacturing Process Transfer Plan to transfer the Manufacturing Process (together with copies of relevant books and records) in SBL's possession or under its control, to Client as set forth in greater detail in the Manufacturing Process Transfer Plan. SBL shall only be obligated to use its Commercially Reasonable Efforts in the implementation of the Manufacturing Transfer Plan, and in no case shall SBL personnel be required to visit the site of Client or any third party manufacturer. For the avoidance of doubt, the foregoing prohibition shall not be construed as a basis for SBL refusing to assist in the transfer of analytical methods to an independent laboratory, including a visit by SBL personnel to such site to assist in method transfer, if, and only as, reasonably necessary, and at Client's cost and expense. For the avoidance of doubt, Manufacturing Process Transfer shall include, without limitation, the transfer of data, information, or samples of validated or SBL manufactured or partially manufactured Products or other indicia measured at various points during Manufacture, to the extent SBL possesses such data, information, or samples. SBL shall not be obligated to deliver the proprietary process information contained in any drug master file with respect to which SBL has granted Client a right of reference. SBL shall provide Manufacturing Process Transfer services with a capacity of one (1) full time employee equivalent ("FTE") month for each of upstream, downstream and analytics, for a total maximum of three (3) FTE months. SBL will have no obligation to provide Manufacturing Process Transfer services more than twelve (12) months after termination of the MSA or applicable PSA. For clarity, SBL's assistance shall be time- based, and SBL shall in no way guarantee or ensure that after Manufacturing Process Transfer services is complete, Client or its third-party manufacturer will be abl
- 1.47 "Non-Affected Party" is defined in Section 17.3.
- 1.48 "Non-Conforming Product" shall mean an entire Batch of Product, any portion of which fails to conform to the Specifications, cGMP (if applicable), and any other mutually agreed upon written express requirements for SBL to follow under the applicable PSA and the applicable QAG.
- 1.49 "Party" and "Parties" is defined in the preamble.
- 1.50 "Pilot Batch" means a Batch of Product designated as a pilot Batch which shall not comply with cGMP and is not required to meet the Specifications.
- 1.51 "Pre-Approval Inspection" means an on-site inspection of the Facility by the Regulatory Authority prior to granting the Regulatory Approval for a Commercial Product as required by various Regulatory Authorities to ensure that the Manufacturing Process and the Facility meet the appropriate requirements and comply with cGMP.
- 1.52 "Process Validation Batch" shall mean a Batch of Commercial Product produced from a process validation run conducted by SBL hereunder to (i) demonstrate and document the consistency and reproducibility of the Manufacturing Process at the Facility, and (ii) support the Regulatory Approval of both the Product Manufactured and the Manufacturing Process at the Facility each as defined in the Project Plan.

- 1.53 "Product" shall mean Clinical Product or Commercial Product to be Manufactured by SBL pursuant to this MSA and any applicable PSA.
- 1.54 "Product Purchase Commitment" is defined in Section 5.7.
- 1.55 "Product specific agreement" or "PSA" is defined in Section 2.1.
- 1.56 "Project Management Team Leader" is defined in Section 3.3.2.
- 1.57 "Project Plan" shall mean a formal, approved document used to guide both project execution and project control. The primary uses of the Project Plan are to document planning assumptions and decisions, facilitate communication among project stakeholders, and document approved scope, cost, and schedule baselines. The Project Plan will contain the description and overall objectives of the Services for Manufacturing a Product and shall include, among other things: (a) JSC and Core Team membership rosters, (b) change request procedures, (c) details, intentions, and deliverables for Technology Transfer, (d) project schedule, (e) detailed procurement plan, as needed, and (f) project budgets and invoicing plans.
- 1.58 "PSA Effective Date" shall mean the effective date of any PSA governed by this MSA.
- 1.59 "Purchase Order" is defined in Section 5.6.
- 1.60 "Quality Agreement" or "QAG" shall mean that certain quality agreement between the Parties that governs their respective responsibilities related to quality systems and quality requirements for the Product(s) Manufactured hereunder, including quality control, testing and release of such Product(s) at the Facility. Clinical Products and Commercial Products shall have separate forms of QAGs.
- 1.61 "Quarter" means each period of three (3) consecutive calendar months beginning on January 1, April 1, July 1, or October 1.
- 1.62 "Raw Materials" shall mean those materials that are used in the Manufacturing Process, including, but not limited to, chemicals, reagents, filters, excipients, disposable consumables, and secondary packaging materials. Raw Materials exclude the Client Materials.
- 1.63 "Receiving Party" is defined in Section 10.1.
- 1.64 "Reference Standards" shall mean materials for standards prepared by Client and/or SBL in accordance with the applicable OAG.
- 1.65 "Regulatory Approval" shall mean all approvals, licenses, registrations or authorizations thereof of any national, regional, state or local regulatory agency, department, bureau or other governmental entity in any jurisdiction where the Product is marketed or intended to be marketed, necessary for the manufacture and sale of the Product, which manufacturing includes the Manufacturing of the Products at the Facility.
- 1.66 "Regulatory Authority" shall mean any national (e.g., the FDA), supra-national (e.g., the EMA), regional, state or local regulatory agency, department, bureau, commission, council or other governmental entity, in any jurisdiction responsible for granting the Regulatory Approval.
- 1.67 "SBL Assignable Error" means: negligence, fraud, recklessness, willful misconduct, or material breach of cGMP (if applicable) on the part of employees, consultants, contractors, sub-contractors, agents or representatives of SBL.
- 1.68 "Service" or "Services" is defined in Section 2.1.

- 1.69 "Service Fee" is defined in Section 9.1.
- 1.70 "Specialized Raw Materials" is defined in Section 5.3.1.
- 1.71 "Specification(s)" shall mean the observable and measurable characteristics of the Products, Client Materials, or Raw Materials, as the case maybe, and the criteria for their storage, handling, packaging and shipping, which details are provided in documentation as reviewed and approved in writing by the Parties.
- 1.72 "Standard Operating Procedure(s)" or "SOP(s)" shall mean the standard operating procedures established by and mutually agreed upon by both Parties regarding the Manufacturing Process.
- 1.73 "Technology Transfer" shall mean the activities by the Parties necessary to Manufacture the Product for Client at the Facility or otherwise conduct the Services, as further described in the applicable PSA and/or Project Plan which may include: (i) transfer of the Client Technology and Client Material from Client to SBL; (ii) implementation of the Manufacturing Process at the Facility, including establishing a small scale Manufacturing Process model at SBL; (iii) all Manufacturing Process fit activities, including required small- and large-scale process development and validation work as allocated between the Parties to SBL and process engineering required to modify / equip, qualify and validate the Facility for the Manufacturing of the Commercial Product; (iv) stability testing, if applicable, for the Product required for licensure; (v) comparability testing to the appropriate reference product, and (vi) regulatory support for all Regulatory Approvals, each as further described in the Project Plan.
- 1.74 "Term" is defined in Section 15.1.
- 1.75 "Warehouse" shall mean SBL's warehouse for storage of the Product located at 300, Songdo bio-daero, Yeonsu-gu, Incheon, 21987, Republic of Korea.

SECTION 2 RELATED AGREEMENTS AND EXHIBITS

- 2.1 Product specific agreements. Pursuant to one or more product specific agreements entered into and mutually agreed from time to time by duly authorized representatives of the Parties ("Product specific agreements" or "PSAs"), SBL will perform manufacturing services for Client as specified in such PSAs and applicable Project Plan and in accordance with the terms and conditions of this MSA ("Services"). Each PSA shall refer to this MSA and contain as applicable (i) a high level scope of work of the Services to be performed under such PSA which describes key activities, (ii) the Product for which Samsung will perform such Services for Client, (iii) a description of the Cell Line; (iv) fees to be paid to SBL by Client for the Services with a general timing plan for invoicing and a more detailed plan to be in the Project Plan, (v) if the Services pertain to the manufacture of the Product, the number of batches of Product to be manufactured by SBL and delivered to Client and the Specifications, (vi) any other deliverables, (vii) the Samsung facility where the Services are to be performed, and (viii) the Regulatory Approvals to be obtained by the Parties. Services shall be governed by the terms and conditions of this MSA, the applicable PSA, and any applicable Quality Agreement. In the event of a conflict between a Quality Agreement will control. In the event of a conflict between any provision of this MSA and the PSA, this MSA shall control, except as otherwise explicitly specified in the PSA.
- 2.2 Project Plan. Concurrently with the execution of a PSA or within a reasonable time after the PSA effective date, the Parties shall agree upon a Project Plan which will specify in detail scope and schedule of the Services, including Technology Transfer and Manufacture. The Project Plan shall also set forth the JSC members (if applicable), Core Team members, and Project Management Team Leader for the Services as well as the frequency and duration of meetings. The Project Plan may be updated as needed by the mutual agreement of the Client and SBL and is governed by and incorporated into the applicable PSA by reference. If there is a conflict between the Project Plan and the applicable PSA, the PSA shall control. If any of the assumptions shared in writing by the Parties and on which the Parties have relied in defining the scope of the activities required to effect the Technology Transfer and/or other Services including but not limited to Manufacture, and the associated timeframes, fees, expenditures and costs proves to be invalid, or if for any reason other than a Party's negligence in planning, it becomes apparent that additional activities are required as part of or in connection with the Technology Transfer and/or other Services including but not limited to Manufacture, the Parties shall (acting reasonably and in good faith) discuss and seek to agree appropriate revisions to the Technology Transfer activities, and associated timelines and pricing.
- 2.3 Quality Agreement (QAG). The Parties shall agree upon and finalize a Quality Agreement in good faith within a reasonable period time after each PSA Effective Date which shall cover such PSA. No Manufacturing of Products for human use shall be conducted without an agreed upon, applicable QAG.Clinical Products and Commercial Products shall have separate QAGs. The Quality Agreement may be amended from time to time, subject to the JSC's approval followed by the Parties' written agreement pursuant to Section 17.9 (if applicable), which is incorporated herein by reference.

SECTION 3 MANAGEMENT OF SERVICE

3.1 General. Each Party will be responsible for its internal decision making process and for reasonably informing the other Party of decisions affecting the Service in a regular and timely manner. Without limiting the foregoing, the Parties shall establish the joint committees or teams set forth herein to advise the Parties on certain matters including, without limitation, Facility modification, Technology Transfer, and optimization of the Manufacturing operation relating to the Product.

3.2 Joint Steering Committee.

- 3.2.1 Formation and Composition. The applicable Project Plan will set forth a Joint Steering Committee for that Product (the "Joint Steering Committee" or "JSC") if the Parties mutually agree that such JSC is necessary. The JSC will be a cross-functional committee composed of an equal number of representatives appointed by each of Client and SBL with each of Client and SBL having at least three (3) representatives, and with one (1) representative from each of Client and SBL having oversight for quality activities, and with one (1) representative from each of Client and SBL having oversight for manufacturing and supply chain activities, including the transfer and implementation of the Manufacturing Process at the Facility. Either Party may replace any or all of its representatives at any time. Such Party shall notify, in writing, of such replacement to the other Party.
- 3.2.2 Responsibilities. The JSC shall (i) establish and oversee the governance structure for the Service including the formation of any subcommittee hereunder; (ii) monitor any Facility modification and the Technology Transfer and Manufacturing strategy of the Product at the Facility, including strategies for the Regulatory Approval of the Facility to Manufacture the Product; (iii) provide strategic guidance to the Core Team as required by the Project Plan; (iv) conduct high level project stage reviews with the Core Team as required by the Project Plan at appropriate milestones or completion of key deliverables or a sequence of event to review and approve key deliverables, evaluate the Core Team's progress and performance, all in order to ensure that the Manufacturing Process is being implemented appropriately; (v) advise on and/or resolve business, manufacturing, supply chain, quality, regulatory or other issues unresolved at the Core Team level; (vi) review and recommend for approval by the Parties any changes to the MSA or the applicable PSA; (vii) review and approve changes to the Specifications, analytical methods, the Manufacturing Process, the Facility or equipment as escalated to the JSC by the Core Team or by a Party pursuant to Section 3.6 below; (viii) review completion of the Service; (ix) settle disputes or disagreements unresolved by a subcommittee; and (x) perform such other functions as appropriate to further the purposes of the MSA as determined by the Parties. For the avoidance of doubt, the JSC shall be a reviewing and consultative body, without authority to make decisions not otherwise concurred in by the Parties.

3.3 Core Team.

- 3.3.1 Formation and Composition. The applicable Project Plan will set forth a Core Team for that Product (the "Core Team"). The Core Team shall be composed of an equal number of representatives from each of SBL and Client, with up to four (4) representatives appointed by each of Client and SBL. Such representatives will include the Project Management Team Leaders of Client and SBL as well as such of their representatives from manufacturing, technical operations, supply chain, quality assurance, quality control, regulatory affairs or other individuals with expertise and responsibilities for those functions required from time-to-time to execute the Facility modification, the Technology Transfer and Manufacturing. Either Party may replace any or all of its representatives at any time. Such Party shall notify, in writing, of such replacement to the other Party.
- 3.3.2 <u>Appointment of Project Management Team Leader.</u> Each Party shall appoint a Project Management Team Leader (each, a "Project Management Team Leader.") to act as the primary contact for such Party in connection with matters related to the Service. Each Project Management Team Leader, unless otherwise mutually agreed, shall serve as the leaders of the Core Team. A Party may replace its Project Management Team Leader at any time and from time to time for any reason. Such Party shall notify, in writing, of such replacement to the other Party.
- 3.3.3 Responsibilities. The Core Team shall (i) develop and maintain the Project Plan and monitor, review and manage the Service according to the MSA and applicable PSA; (ii) conduct project stage reviews with the JSC as required by the Project Plan at appropriate milestones or completion of key deliverables or a sequence of event to review key deliverables, review its progress and performance against plans; (iii) develop a change management process to identify, review and recommend any significant changes in the project scope, time, fee or risk to the JSC; (iv) investigate and resolve business, manufacturing, supply chain, quality, regulatory or other issues arising during the Service; (v) review and escalate to the JSC, as needed, changes to the Project Plan or applicable QAG; (vi) review and recommend to the JSC changes to the Specifications, analytical methods, the Manufacturing Process, the Facility or equipment; (vii) coordinate the activities of the Parties relating to the Manufacturing hereunder, including but not limited to: managing the technical operations and quality aspects of routine manufacturing, conducting Product testing and release, and managing supply chain activities including shipping and delivery logistics; (viii) report periodically on operation and quality progress and performance; and (ix) perform such other tasks and undertake such other responsibilities as may be specifically delegate to the Core Team by mutual agreement of the Parties. For the avoidance of doubt, the Core Team shall be without authority to make decisions not otherwise concurred in by the Parties.

3.4 Meetings

- 3.4.1 JSC. The JSC shall meet by audio or video teleconference or in-person as agreed by the JSC or as necessary to make determinations as required of it. Any member of the JSC may designate a substitute to attend and perform the functions of that member at any meeting of the JSC and each Party may, in its reasonable discretion, invite non-member representatives of such Party to attend such meeting with advance notice to the other Party; provided, however, such non-member representatives are subject to enforceable obligations of confidentiality to the inviting Party.
- 3.4.2 Core Team. The Core Team shall meet by audio or video teleconference or in-person as agreed by the Core Team. Any member of the Core Team may designate a substitute to attend and perform the functions of that member at any meeting of the Core Team and each Party may, in its reasonable discretion, invite non-member representatives of such Party to attend such meetings with advance notice to the other Party; provided, however, such non-member representatives are subject to enforceable obligations of confidentiality to the inviting Party.
- 3.4.3 <u>Travel Expenses.</u> Each Party shall be responsible for all of its own expenses of traveling to and participating in any joint committee or team meeting, including the JSC and Core Team.
- 3.5 Decisions. All decisions of JSC, the Core Team and any other joint committee or team formed under the MSA or any applicable PSA, except as expressly set forth herein, shall be made by the unanimous agreement of its members or their designated representatives, with a Party's members of designated representatives having, in the aggregate, one (1) vote (with the casting of fractional votes being impermissible), and shall be reflected in written meeting minutes which summarily address topics discussed, delegation of work, schedules and decision of such committee or team. SBL shall prepare written minutes of the JSC and Core Team within fifteen (15) Business Days of the meeting to which they relate, which shall be subject to approval by the authorized representatives of the Parties; provided, however, no joint committee or team herein may amend or waive any provision of the MSA or applicable PSA, including without limitation, the financial terms set forth in Section 9. The MSA or any PSA may be amended, and provision of the MSA or any PSA may be waived, pursuant to Section 17.9 only.

3.6 Disputes.

- 3.6.1 General. In the event that the Core Team and any other joint committee or team formed under the MSA or any applicable PSA, is unable, despite the good faith efforts of all members, to resolve a disputed issue that is within the purview of such joint committee or team within ten (10) Business Days of meeting request by either Party, the disputed issue shall be referred immediately by such joint committee or team to the JSC. If the disputes still cannot be resolved within an additional twenty (20) Business Days of meeting request by the JSC, the matter may be handled in accordance with Section 16.
- 3.6.2 Project Management Team Leaders. Subject to Section 3.6.1, the Project Management Team Leaders (or their respective designees) will in good faith attempt to mutually resolve in a timely fashion any disagreement with respect to the Service hereunder, which could reasonably affect the quality of the Manufacturing of the Product, including without limitation, the related management processes and operations, control of production planning and scheduling, prioritization decisions, allocation of resources, timing of in-process and release testing, oversight of auxiliary facilities (e.g., in-process tests that need to be conducted at laboratories other than those at the Facility), Facility modification, the Technology Transfer, registration and troubleshooting decisions, and any other matters relating to implementation of the Manufacturing Process and the Manufacturing of the Product hereunder.

SECTION 4 SERVICES

- 4.1 Services. During the Term, in accordance with and subject to the terms and conditions set forth in this MSA, applicable PSA, and the applicable QAG, SBL shall provide the Services to Client relating to the Product(s). SBL shall at all times make Commercially Reasonable Efforts to complete the Services in accordance with the timelines set forth in the applicable PSA. Except as otherwise expressly set forth in the MSA, applicable PSA, or the applicable QAG or as otherwise mutually agreed in writing by the Parties, SBL shall be responsible at its own cost and discretion for operating and maintaining the Facility.
- 4.2 Compliance with Applicable Law. Subject to the provisions of Section 6 below, SBL shall maintain the Facility in accordance with cGMP and in such condition as will allow SBL to Manufacture the Products in accordance with the terms of the MSA and the applicable QAG. SBL shall perform the Services under the MSA in conformance with cGMP, if applicable, any requirements of the Regulatory Authorities that shall be mutually agreed upon by the Parties, and all Applicable Laws.
- 4.3 Project Personnel. SBL shall adequately staff the Facility with personnel necessary (including consultants and contractors), who have sufficient technical expertise by virtue of training and experience to perform its obligations under the MSA and any applicable PSA. Notwithstanding anything to the contrary and in addition to the JSC and Core Team meetings described in Section 3 above, Client and SBL may arrange for core project personnel to have regular meetings, which shall be by audio or video teleconference. The Project Plan shall specify the frequency and duration of such meetings; provided that the associated costs for meetings requested solely by Client in excess of the number set forth in the Project Plan shall be passed through to Client by SBL, unless such meeting is for the purpose of addressing a deficiency in Manufacturing caused by SBL Assignable Error or assessing remediation thereof.

- 4.4 Subcontract. SBL may subcontract any portion of the Services without approval from Client; provided, however, that SBL will not subcontract or otherwise engage subcontractors or other third party agents to manufacture, conduct quality control testing, or perform Services directly related to manufacturing or quality control testing without the prior written approval of Client, which shall not be unreasonably withheld, conditioned, or delayed. Notwithstanding any subcontracting, SBL shall be primarily obligated to Client for any subcontracted services as if it were providing the Services itself, and shall be jointly and severally liable to Client for the performance of any such Service. The preceding sentence is not meant to nor shall it be construed as altering or eliminating any mutually agreed upon limitations of liability contained within this MSA or any applicable PSA. Every subcontractor or other third party agent of SBL performing pursuant to this Section 4.4 shall be required by SBL to agree in writing to comply with relevant portions of this MSA, together with pertinent PSAs and QAGs. For the avoidance of doubt, SBL shall not directly or indirectly use on its own behalf or through any subcontractor, other vendor or any third party in the performance of this MSA or any PSA, any invention, technology, data or information that would require Client to obtain a license from SBL not otherwise granted pursuant to this MSA or an applicable PSA, or obtain a license from any such subcontractor, vendor or third party a license in order to make, have, made, use, offer for sell, sell, import or otherwise exploit any Product, unless Client has agreed in advance in writing.
- 4.5 <u>Development and Manufacturing Site.</u> Except as is provided in Section 4.4 or otherwise agreed by Client, all Services shall be performed by SBL at the Facility.

4.6 Access to the Facility.

- 4.6.1 SBL shall accommodate visits by Client personnel in the Facility during operational hours during the Term, upon Client's request, to coordinate, expedite and guide the Service. Client will provide SBL with written notice at least one (1) month prior to any visit, and the Parties shall decide on a mutually agreeable date, duration, visitor list, and agenda prior to the visit. Additionally, Client may, at no cost to SBL, request up to two (2) of its personnel to be on-site at the Facility to observe and consult with SBL during the performance of Services under this MSA and such additional personnel in such numbers as deemed necessary by Client shall be accommodated upon mutual agreement. SBL shall make reasonable efforts to provide such Client personnel working space and access to guest wireless connections. All applicable out- of-pocket expenses associated with such on-site Client personnel shall be passed through to Client by SBL.
- 4.6.2 While at the Facility, all such Client personnel shall have reasonable access to all areas as are relevant to SBL's performance of the Service hereunder, provided that SBL may reasonably restrict Client personnel's access to those portions of the Facility as it deems reasonably necessary for safety, confidentiality and cGMP, and visitors admitted pursuant to this Section, upon request of SBL, shall comply with SBL policies and procedures as they apply to SBL personnel generally while at the Facility.
- 4.7 Manufacturing Documentation. SBL shall maintain in the English language, complete, true and accurate Manufacturing Documentation, and shall keep them in strict confidence and shall not use them for purposes other than providing or performing the Service or other obligations hereunder. SBL shall maintain all such Manufacturing Documentation for at least that period specified in the applicable QAG or such longer period as may be required by Applicable Law. Upon written request of Client and at mutually agreeable times, Client shall have the right to review Manufacturing Documentation, including the Batch Records, at the Facility as further defined in the applicable QAG; provided, however, SBL shall not delay any such request of Client for more than fifteen (15) Business Days. Client may also request scanned or printed copies of such Manufacturing Documentation, but shall be responsible for reasonable costs associated therewith. SBL shall record and maintain such records, data, documentation and other information in the language as so required in the applicable QAG or as so required by a Regulatory Authority and in compliance with Applicable Law, or as otherwise may be set forth in an applicable PSA. The form and style of Batch documents, including, but not limited to, Batch production records, lot packaging records, equipment set up control, operating parameters, and data printouts, raw material data, and laboratory notebooks are the exclusive property of SBL. Notwithstanding anything to the contrary, SBL SOPs not specific to the Client's Products may be provided to Client for on-site review if deemed necessary by both SBL and Client. Such SOPs cannot be removed from the SBL premises, copied, photographed or otherwise replicated, but to the extent required for filing in connection with any Regulatory Approval related to a Product including, but not limited to, an IND, BLA or MAA, SBL will file, maintain and update such SOPs with appropriate Regulatory Authorities by means of a Drug Master File, Common Technic

<u>SECTION 5</u> <u>SERVICE DESCRIPTIONS</u>

- 5.1 Technology Transfer Requirements. The Parties shall make their personnel available at the Facility to enable the transfer and implementation in accordance with the Project Plan. Client shall transfer to and grant SBL the limited license set forth below in Section 11.2 in respect of the Reference Standards, Client Technology, Client Materials, and Cell Line In the event that Client agrees to utilize SBL's SharePoint portal for Technology Transfer, Client agrees that (a) in the event of any relevant change that affects a Client user's authorization to use such portal, Client shall promptly notify SBL so that SBL may disable their usernames and remove / change passwords in order to secure the SBL Portal and (b) Client shall ensure that all of Client's users have up-to-date antivirus software installed on the computer devices used to access such portal.
- 5.2 Facility Modification and Equipment. Except as otherwise specifically provided herein to the contrary, and upon mutual agreement of the Parties, Client and SBL will agree on what equipment in the Facility is necessary to perform the Services, and if it is necessary or Client deems it necessary to procure additional equipment beyond that which is in the Facility as of the applicable PSA Effective Date, the Core Team shall determine equitable allocation of costs including, as applicable, procurement, validation, installation, maintenance, commissioning, and decommissioning/validation (which determination shall be escalated to the JSC if in dispute). Thereafter, if any additional equipment is necessary at the request of Client, such costs shall be dealt with by the Change provisions of this MSA. SBL shall modify the Facility and engineer, procure, install, commission, test, qualify, troubleshoot and validate necessary equipment and instruments in order to accommodate Manufacture of the Product at the Facility, as further described in the Project Plan and applicable QAG. Except as provided in this Section 5.2 or any applicable PSA, the Facility, Warehouse and all the equipment shall be maintained, tested, validated, calibrated and qualified for their intended uses by SBL at SBL's expense. For the avoidance of doubt, it shall be the responsibility of SBL to effect such changes at the Facility including, but not limited to changes in equipment, as are required by changes in laws, rules or regulations related to the manufacture, handling, storage, packaging and disposal of biologics generally, compliance therewith being necessary in order for SBL to hold itself out as a qualified manufacturer of biologics for human use. It shall be the responsibility of SBL to obtain, maintain and pay all costs associated with establishment licenses required for the Facility, the Warehouse and any other facilities used by SBL in the performance of its obligations under this MSA and any applicable PSA.

5.3 Raw Materials.

- 5.3.1 Management. SBL shall procure and maintain a reasonable quantity of the Raw Materials, required for the Services in accordance with the MSA and any applicable PSA, as further described in the applicable QAG. On a per-Product basis, the Core Team shall finalize the categorization of the Raw Materials into Raw Materials which shall be used for that specific Product only ("Specialized Raw Materials"), Raw Materials which can be used across multiple products and/or customers ("Common Raw Materials"), and Raw Materials which will not be charged on a cost-plus basis to the Client, and shall attach such list to the applicable PSA. Such list of Common Raw Materials and Specialized Raw Materials may be amended from time to time, subject to the Parties' approval. Prior to the commencement of Manufacture, the Core Team shall agree on estimates for quantities of Raw Materials anticipated to be consumed in the Manufacture of each Batch. Although SBL will make Commercially Reasonable Efforts to use no more than those amounts, SBL will not be responsible for Raw Materials used in excess of the agreed-upon estimate; provided, however, that SBL shall be responsible for any such excessive use, loss, spoilage, or waste of such Raw Materials caused by an SBL Assignable Error. In order to become effective, SBL's reasonable strategies regarding Raw Material safety stock and sourcing from qualified vendors shall be agreed upon by Client. In the event SBL is not able to utilize any capacity reserved to Manufacture Product according to an agreed-upon forecast or manufacturing plan due to Client's failure to agree to such reasonable strategies, then Client shall be responsible for the costs of such reserved capacity regardless of whether it is utilized or not.
- 5.3.2 **Data Transfer.** Client and SBL shall agree on the Specifications for the Raw Materials, including without limitation analytical methods, supplier information including supplier site information, and other information concerning the stability, storage, and safety thereof that are available and required for the Manufacturing hereunder, as further described in the applicable QAG.
- 5.3.3 Testing and Evaluation. SBL or vendors qualified by SBL shall perform all testing and evaluation of the Raw Materials as required by the Specifications for the Raw Materials and the cGMPs, as further described in the applicable QAG, if applicable. SBL shall not release any Raw Materials from quarantine that do not meet their Specifications or are not otherwise suitable for cGMP use.
- 5.3.4 Storage. SBL shall secure sufficient and suitable cGMP storage facilities and/or equipment at the Facility that meet the Specifications for storage of the Raw Materials. SBL shall preserve and protect the Raw Materials from loss and damage while in SBL's possession, consistent with reasonable technical and business judgment, the Specifications and any relevant SOPs or other instructions provided by Client; provided however, SBL shall be responsible for loss of Raw Materials after receipt by SBL and prior to Manufacture only in cases of SBL Assignable Error or breach of this Section 5.3.4. In all other cases, Client shall be responsible for the risk of loss of the Raw Materials. Upon obsolescence, or upon expiration or earlier termination of a PSA, Client shall be responsible for the loss of Raw Material purchased in reliance on a Purchase Order, Firm Period, or Binding Year which Client fails to honor and SBL cannot reasonably otherwise utilize such Raw Material.
- 5.3.5 Service Fee Related to Raw Material. Common Raw Materials and Specialized Raw Materials will be charged on a cost-plus basis to Client in accordance with Sections 9.1(ii) and 9.2.2, subject to any changes in the scope of work.

5.4 Client Materials.

- 5.4.1 Management. Client shall provide to SBL free of charge, either by itself or through its third party supplier, Client Materials in amounts reasonably necessary to carry out the Services as agreed by the Parties. The applicable PSA shall set forth the exact timing of such provision of Client Materials to SBL. SBL shall make Commercially Reasonable Efforts to import the Client Materials to the Republic of Korea in a timely manner, provided that Client provides reasonable assistance. The title to such Client Materials shall remain at all times with the Client. Prior to the commencement of Manufacture, the Core Team shall agree on estimates of quantities for Client Material anticipated to be consumed in the Manufacture of each Batch. Although SBL will make Commercially Reasonable Efforts to use no more than those amounts, SBL will not be responsible for Client Materials used in excess of the agreed-upon estimate; provided, however, that SBL shall be responsible for any such excessive use, loss, spoilage, or waste of such Client Materials caused by an SBL Assignable Error, and further provided that SBL shall not be liable for the monetary value of Cell Line loss if it complied with risk mitigation measures to protect the Cell Line, as reasonably agreed to by the Parties and set forth in the Quality Agreement. In order to become effective, SBL's reasonable strategies regarding Client Material safety stock and sourcing from qualified vendors shall be agreed to by Client. In the event SBL is not able to utilize any capacity reserved to Manufacture Product according to an agreed-upon forecast or manufacturing plan due to Client's failure to agree to such reasonable strategies, then Client shall be responsible for the costs of such reserved capacity regardless of whether it is utilized or not.
- 5.4.2 **Data Transfer.** Client shall provide SBL with the Specifications of the Client Materials, including without limitation analytical methods, supplier information, and other information concerning the stability, storage, and safety thereof that are available and required for the Manufacturing hereunder, as further described in the applicable QAG.
- 5.4.3 Testing and Evaluation. SBL shall perform testing of the Client Materials in accordance with the applicable QAG and/or Client's instruction prior to the performance of the Manufacturing hereunder, in order to determine whether such Client Materials meet the Specification described in the applicable QAG (if applicable). SBL shall inform Client of (a) any damage to the Client Materials received that is visually obvious (e.g., damaged or punctured containers and temperature monitoring results outside of predetermined Specifications) within five (5) Business Days after SBL's receipt of the Client Materials and (b) any non-conformance of the Client Materials to Specification either: (i) within sixty (60) calendar days after SBL's receipt of the Client Materials or (ii) if release testing of Client Materials is not to be performed until it is needed for Manufacture, within sixty (60) calendar days after such release testing is performed; or (iii) as is otherwise agreed between the Parties. If, prior to performing any Service on the Client Materials, SBL determines that such Client Materials are defective or damaged, SBL shall not perform the Service on such Client Materials and shall follow Client's written instructions regarding disposal or return of such Client materials to Client, such disposal or return to be at Client's discretion and cost.
- 5.4.4 Storage. SBL shall secure sufficient and suitable cGMP storage facilities and/or equipment at the Facility that meet the Specifications for storage of reasonable quantities of Client Materials in light of quantities of Product anticipated to be Manufactured. SBL shall preserve and protect the Client Materials from loss and damage while in SBL's possession, consistent with reasonable technical and business judgment, the Specifications and any relevant SOPs or other instructions provided by Client; provided that SBL shall only be liable for loss and damage after receipt/acceptance by SBL and prior to Manufacture in cases of SBL Assignable Error or breach of this Section 5.4.4. In all other cases, Client shall be responsible for the risk of loss of the Client Materials.
- 5.4.5 Service Fee Related to Client Material. Handling fees relating to the Client Material will be charged to Client in accordance with Sections 9.1(iii) and 9.2.3.
- 5.5 Forecasts. For each Commercial Product, the Parties shall determine a mutually agreeable mechanism for forecasting of each Product, which shall be detailed in writing and attached to each relevant PSA. For Clinical Product, the Parties shall agree upon the number and schedule of Batches to be Manufactured by SBL in the applicable PSA.
- 5.6 <u>Purchase Orders.</u> For each Clinical Product or Commercial Product, Client shall notify SBL in a binding form and procedure to be agreed upon in the applicable PSA requesting a specific amount of Product to be Manufactured (a "**Purchase Order**").
- 5.7 <u>Product Purchase Commitment.</u> As further set forth in a PSA, during the Term the Parties may agree that Client will purchase a minimum quantity of batches of a certain Product in a given year (a "**Product Purchase Commitment**").

5.8 Batch Failure during Manufacture

- 5.8.1 If, during Manufacture of a Batch and prior to SBL's Batch release, the Core Team determines that a Batch is Non-Conforming Product (a "Batch Failure"), SBL shall take Commercially Reasonable Efforts to promptly Manufacture (except to the extent prohibited by cGMP or applicable QAG) and deliver to Client a replacement Batch on a date to be mutually agreed by the Parties, which Batch shall be invoiced and paid for as if it were the failed Batch (i.e., Client shall only be invoiced for the Batch conforming to the Specifications and Manufactured in compliance with cGMP, and actually delivered). Client shall be responsible for the costs and fees of the Raw Materials and Client Materials for the replacement Batch as if it were the failed Batch (i.e., Client shall only be invoiced the applicable cost of the Raw Materials and Client Materials actually used to Manufacture the conforming, cGMP compliant replacement Batch). Client shall use Commercially Reasonable Efforts to ensure that SBL has adequate Client Materials to Manufacture such Batches and Client's failure to provide such Client Materials shall relieve SBL of its obligations to provide a remedy pursuant to this Section 5.8. The remedies contained in Section 5.8 of this MSA shall be the sole and exclusive remedies of Client regarding a Batch Failure and a Batch Failure shall not constitute a material breach of this MSA or a PSA unless SBL fails to provide the remedies contained in this Section 5.8.
- 5.8.2 The Parties shall conduct a root cause analysis (including an analysis of whether the Batch Failure was the fault of SBL, Client, neither or both) of the Batch Failure, which shall be done through SBL's deviation process and which result will be reviewed and confirmed by the JSC. If either the Core Team does not agree on the Batch Failure root cause, and the JSC does not agree on the results of the Core Team's Batch Failure root cause analysis, the Parties shall refer review of such root cause analysis to an independent mutually agreed-on laboratory or firm with international repute, acting as a neutral arbiter, to conduct a root cause analysis of the Batch Failure. The costs of the independent laboratory will be borne by the Party responsible for the Batch Failure, as determined by the independent laboratory, whose written decision shall be binding on the Parties. If the independent laboratory is paid in advance of a final written determination, the Parties shall share equally such expense, and the Party at fault shall reimburse the other Party for its share of any such expenses paid after there has been a final written determination made by the independent laboratory.
- 5.8.3 The PSA applicable to such Product Batch Failure shall set forth responsibility among the Parties for the following costs in the event of a Batch Failure: (1) the SBL Service Fee to Manufacture the failed Batch; (2) SBL's costs to procure the Raw Materials used in the failed Batch plus applicable SBL handling fees associated with such Raw Materials; (3) SBL handling fees associated with the applicable Client Materials used in the failed Batch; and (4) Client's cost to procure the Client Materials (which shall not include Cell Line) used in the failed Batch which amount is to be calculated based on the actual replacement value (as opposed to the market value) of such materials as supported by reasonable documentary evidence by Client.
- 5.8.4 In the event that any of the foregoing procedures results in a Batch being delivered in a different year than the year in which the original Batch was ordered for delivery by Client, the Service Fee for such re-Manufactured Batch shall be the Service Fee in effect in the Year in which such re-Manufactured Batch is actually delivered by SBL unless the root cause analysis determined the Batch Failure resulted from SBL Assignable Error, in which case, the Service Fee in effect at the time of the Purchase Order for the failed Batch shall apply.

5.9 Storage, Packaging and Delivery.

5.9.1 Service Deliverables other than Products. Storage, packaging and delivery of the Service deliverables other than Products Manufactured hereunder shall be made in accordance with the terms of this MSA, applicable PSA, Project Plan, applicable QAG and the Applicable Laws.

5.9.2 Products.

- (a) Release by SBL and Acceptance by Client.
 - (i) SBL shall perform all testing in accordance with the Specifications of the Product and release the Product in accordance with the terms of the applicable QAG. Upon such release SBL shall deliver to Client a copy of the Manufacturing Documentation in support of the SBL's release of the Product for each Batch ("Batch Related Documents"), including a Certificate of Analysis and Certificate of Compliance, in accordance with the applicable QAG;

- (ii) Acceptance of Product. Client will complete the Acceptance Procedure and determine the acceptability of such Product in accordance with the applicable QAG and notify SBL of the result within the latter of thirty (30) Business Days of SBL's release of Product or Client's receipt of the Batch Related Documents. Upon Client's acceptance, SBL will have no liability for such Product once entrusted to the carrier (the Parties shall use goodfaith efforts to minimize the time-period between Client's acceptance and SBL's delivery to the carrier), except as set forth in Section 5.9.2(a)(iv) regarding Latent Defects. If Client does not reject such Product within the thirty (30) Business Day period, the Product will be deemed to have been accepted by Client and SBL will have no liability for such Product once entrusted to the carrier (the Parties shall use good-faith efforts to minimize the time-period between Client's acceptance and SBL's delivery to the carrier), except (A) as set forth in Section 5.9.2(a)(iv) regarding Latent Defects or (B) due to a breach of SBL's obligations with regard to (I) handling or storage or (II) the Specifications insofar as they relate to packaging and shipping.
- (iii) **Non-Conforming.** If, during the Acceptance Procedure, any Product is determined by Client or SBL as Non-Conforming Product, at the option of Client, SBL shall take Commercially Reasonable Efforts to promptly Manufacture replacement Product (except to the extent prohibited by cGMP or applicable QAG) and deliver to Client the quantity of the Product equivalent to the quantity of Non-Conforming Product on a date to be mutually agreed by the Parties and such replacement Batch shall be invoiced and paid for as if it were the original Non-Conforming Product. If SBL does not confirm the non-conformity, the Parties shall refer to an independent and mutually agreed-on laboratory or firm with international repute to test the disputed Product. The costs of the independent laboratory will be borne by the Party responsible for the Batch Failure, as determined by the independent laboratory, whose written decision shall be binding on the Parties. If the independent laboratory is paid in advance of a final written determination, the Parties shall share equally such expense, and the Party at fault shall reimburse the other Party for its share of any such expenses paid after there has been a final written determination made by the independent laboratory. Section 5.9.2(a) (i) and (ii) shall apply to such replaced Product <u>mutatis mutandis</u>. Responsibility for the costs of such Non-Conforming Product shall be as if such Non-Conforming Product is a Batch Failure and Section 5.8.2 5.8.4 shall apply to such Non-Conforming Product <u>mutatis mutandis</u>. The remedies contained in this Section 5.9.2 shall be the sole and exclusive remedy of client for Non-Conforming Product.
- (iv) Latent Defect. At any time after completion of review of the Batch Related Documents, if Client finds any hidden defects of the Product which could not have been reasonably discovered through the review of the Batch Related Documents ("Latent Defect"), Client shall promptly give notice of such claim in writing to SBL. In such case, if Client proves the Latent Defect is solely due to SBL Assignable Error, the above Section 5.9.2(a)(iii) and Section 13.1 shall apply. If no written claim for Latent Defect of the Product is received by SBL within one (1) month from Client's discovery of the Latent Defect, the Product shall be deemed as irrevocably accepted. Notwithstanding anything to the contrary; such claim for Latent Defect must be made within a time period to be set forth in the applicable PSA, but in no event after the expiration date of the Product. For the avoidance of doubt, a Latent Defect is a defect in existence at the time of completion of the Acceptance Procedure.
- (b) **Delivery.** Shipping conditions, title transfer to Client and risk of loss for the Product Manufactured hereunder shall be FCA SBL Facility (INCOTERMS 2010). The Parties further agree as follows:
 - (i) At the time of SBL's release of the Product and prior to each pick-up by Client or Client's designated carrier, SBL shall propose to Client a delivery schedule of the Product, in order for the Parties to agree on it in advance for each pick-up. SBL shall schedule Delivery with the carrier selected and paid for by Client;
 - (ii) SBL shall not deliver the Product until it has been instructed to by Client in accordance with the applicable QAG. Client shall confirm specific delivery instructions with SBL prior to SBL release. Upon SBL's release of Product, SBL shall store the Manufactured Product as described in Section 5.9.2(c) and Client shall compensate SBL for storage costs for the Manufactured Product stored with SBL for more than sixty (60) calendar days following release, as set forth in the applicable PSA;
 - (iii) SBL shall provide Client with invoice, packing lists, supporting export documents as specified by Client by separate delivery and shipment documentation instructions, together with each shipment of the Product (or such other deliverables); and
 - (iv) In cooperation with Client and subject to the delivery schedule agreed by the Parties, SBL shall adhere to the first-expire-first-out (FEFO) principle in shipping all released Product.

(c) Storage, Packaging and Shipping Container.

- (i) Pursuant to the terms of this MSA and any applicable PSA, SBL shall store the Products Manufactured hereunder.
- (ii) SBL shall store, package, label and prepare shipment according to the Specifications for the Product Manufactured hereunder, the applicable QAG and the SOPs, and using storage and/or shipping containers determined in the applicable PSA.
- (iii) If Client does not direct SBL to prepare Manufactured Product to be picked up by Client or Client's designated carrier with a pick-up date within sixty calendar (60) days of Client's receipt of the Batch Related Documents, SBL shall store the Product at the Warehouse and Client shall pay storage fees to SBL as set forth in Section 9.1 for the period of storage at the Warehouse in excess of sixty (60) calendar days from release until the actual delivery date, in accordance with the applicable PSA.
- (iv) SBL shall store and handle all Manufactured Product in accordance with applicable storage and handling Specifications, including all times when SBL and Client agree that SBL shall store Manufactured Product per Section 5.9.2(c)(iii), but shall have no obligation regarding risk of loss and damage of Manufactured Product in its possession unless such loss or damage resulted from SBL's negligence or willful misconduct.
- 5.10 **Supply Interruptions**. On a PSA by PSA basis, the Parties will discuss appropriate steps to alleviate any expected or actual shortfall in Manufactured Product. The Parties may agree on a PSA by PSA basis, in what events, if any, such a supply failure could constitute a material breach of this MSA or a PSA.

SECTION 6 CHANGES TO THE SPECIFICATIONS, ANALYTICAL, METHODS, MAUNFACTURING PROCESS, FACILITY OR EQUIPMENT

- 6.1 <u>Approval for Change.</u> SBL shall not make any change to the Manufacturing Process, the Services, or the Specifications (a "Change"), without the prior written consent of Client in accordance with the applicable QAG.
- 6.2 Changes Required by cGMP, Regulatory Authorities or Requested by Client. Except as otherwise expressly set forth to the contrary in the applicable QAG, in the event that cGMP, a Regulatory Authority, Applicable Law, or any other regulatory or legal authority requires, or Client requests, a Change, SBL shall accommodate such requirements or requests, subject to the following:
 - (a) Client shall promptly notify SBL in writing of the required and/or requested Change(s), and provide information necessary for SBL to evaluate the effect of such Change(s); provided, however, in the event SBL first learns of a required Change it shall notify Client thereof in writing, and SBL shall promptly advise Client as to any (i) additional equipment required, modifications to the Facility or equipment, and/or additional equipment and the Facility qualification and validation requirements; (ii) Manufacturing Process development, transfer, scale-up, testing, qualification, or validation requirements; (iii) regulatory requirements pursuant to such Changes; (iv) changes to the Manufacturing scheduling and/or Product delivery schedule; and (v) other impacts on the Facility or SBL's ability to manufacture products (including the Products) in the Facility, if any, which may result from such Change(s). The notification and formal approval procedure of such Changes shall be in accordance with the applicable QAG (i.e., change control procedures) (if applicable). The Parties shall meet in a timely manner to identify and discuss such Changes as appropriate;
 - (b) Prior to implementation of any such Change(s), SBL, subject to Section 5.2, shall provide Client with an estimated plan and budget of the reasonable and necessary costs that would be incurred by SBL as a result of the implementation of any such Change(s), including, but not limited to for (i) process and analytical development; (ii) equipment and/or the Facility modifications, qualification, validation, maintenance, and decommissioning/disposal; (iii) process and analytical validation; (iv) document revisions or changes, the Facility, equipment, and system modifications or changes; (v) additional stability testing; and (vi) preparing submissions to Regulatory Authorities (collectively, the "Implementation Plan and Budget"). Following review and approval by Client of such Implementation Plan and Budget, subject to the Core Team's approval and agreement followed by the Parties' written agreement pursuant to Section 17.9 (if applicable), SBL shall commence implementation of such Change(s);
 - (c) During any such implementation, SBL shall provide Client with regular updates on the progress of implementation. Subject to any timeframe imposed by Applicable Law, SBL shall exercise Commercially Reasonable Efforts to implement the Change according to the Implementation Plan and Budget's target completion date. SBL shall provide written notice to Client if SBL becomes aware of any cause which may create delay with the implementation of Changes. Following any such notice, both Parties shall discuss an amendment of Implementation Plan and Budget; and

- (d) Upon the approval of the Implementation Plan and Budget for Change(s), both Parties shall negotiate in good faith to determine the allocation of the costs incurred by SBL for the implementation of any such Change(s) between the Parties, in accordance with the following principles:
- (i) the costs for the general Facility Changes required by cGMP, any Regulatory Authority, or any Applicable Laws related to the maintaining the Manufacturing Facility by SBL as set forth in Section 7.2, shall be borne by SBL, provided that where the Change relates exclusively or partially to the Manufacture of Product in which case the costs shall be borne by Client fully or proportionally, respectively; (ii) the costs for the Changes other than (i) above, and requested by Client and required uniquely to the Manufacture of the Product and beneficial solely to Client shall be borne by Client; and (iii) the costs for the Changes other than (i) and (ii) above shall be discussed in good faith by the Parties to achieve equitable allocation of costs.

SECTION 7 REGULATORY APPROVALS AND INSPECTIONS

- 7.1 Regulatory Approvals. SBL shall provide reasonable assistance and cooperation in order for Client to obtain and maintain the Regulatory Approvals. The direct costs and fees associated with such assistance and cooperation, to the extent not detailed in the MSA or PSA shall be borne by Client, or as otherwise mutually agreed between the Parties. As specified in the applicable PSA, the Parties shall agree on which Regulatory Approvals are to be obtained.
- 7.2 Regulatory Approvals for the Facility. SBL shall obtain and continuously maintain all approvals, licenses, registrations or authorizations of any federal, state or local regulatory agency, department, bureau or other governmental entity (other than the Regulatory Approvals, which will be obtained or maintained by Client) that are required to Manufacture and ship the Product at the Facility and perform the Services including, but not limited to, cGMP Manufacture.
- 7.3 Regulatory Inspections. SBL shall facilitate on-site inspections of the Facility conducted by Regulatory Authorities. SBL shall notify Client according to the applicable QAG provisions of any contacts or inquiries by the Regulatory Authorities, including inspections, Pre-Approval Inspections, sample requests, and written correspondence and its result, related to the Manufacture of Product by SBL at the Facility, as further defined in the applicable QAG. Any direct expenses or costs incurred by SBL for such inspections including Pre-Approval Inspections at the Facility shall be borne by Client. Unless prohibited by the pertinent Regulatory Authority, Client shall be entitled to witness any inspection or audit by a Regulatory Authority of the Facility or Warehouse related to the Manufacture of Product by SBL at the Facility. Further, Client shall be entitled to attend, as an observer, any wrap-up meeting between SBL and a Regulatory Authority related to such an inspection related to the Services or the Manufacture of Product by SBL at the Facility. SBL shall timely provide Client with a copy of any Form 483, inspection report or regulatory letter issued by a Regulatory Authority, and shall provide Client a meaningful opportunity to review and comment upon any response of SBL thereto. Client's comments in such regard shall be considered in good faith and be given due regard by SBL in formulating any proposed response to a Regulatory Authority.
- 7.4 Regulatory Support. During the Term, SBL will assist Client with all regulatory matters relating to Services and Manufacture and review the Common Technical Document pertaining to the Products, as it relates to Chemistry, Manufacturing and Controls, and make such corrections as are necessary to accurately reflect the Products, in each case at Client's request and reasonable expense; provided, however, SBL shall review and correct such documents as they relate to SBL activities at no charge to Client (updates will be made with respect to regulatory filings for which Client has engaged SBL and paid the associated flat rate fees for support). In addition, SBL will maintain at SBL's expense, the relevant Drug Master File, including any updates thereto, and shall provide a letter authorizing Client to reference SBL's Drug Master Files on file with the FDA and other Regulatory Authorities in connection with the pursuit of Regulatory Approval for the Products.

SECTION 8 QUALITY COMPLIANCE

- 8.1 Quality Agreement. Both Parties shall adhere to the provisions of the applicable QAG and the Parties agree that all elements of quality assurance, quality control and the like shall be governed by the terms and conditions of the applicable QAG. In the event of a conflict between the MSA and the applicable QAG, the MSA shall prevail over those of the applicable QAG with the exception of Product quality-related matters, cGMP and related regulatory requirements in which case, the terms of the applicable QAG shall prevail.
- 8.2 Responsibility for Recalled Product. For a Commercial Product, either Party shall notify the other Party as soon as practicably possible if any Commercial Product is the subject of a threatened or actual recall by a Regulatory Authority, (a "Recall") which may be attributable to any Service or Manufacture by or on behalf of SBL hereunder. Client shall be responsible for conducting all Recalls and shall make all decisions regarding, and in all events shall have sole authority for, conducting any recalls, market withdrawals or corrections with respect to the Product and SBL shall at all times exercise Commercially Reasonable Efforts to provide its assistance and cooperation to Client in conducting such Recalls to the extent the Recall arises out of SBL's Manufacture of Product. Details regarding the roles and responsibilities of the Parties in regard to Recalls are set forth in the applicable QAG. If such Recall results solely from SBL Assignable Error, SBL shall be responsible for direct and documented out-of-pocket costs associated with such Recall subject to the limitation of liability sections of the PSA applicable to such Recalled Product and, subject to written instructions of Client to the contrary, will use its Commercially Reasonable Efforts to replace the Recalled Products with new Products, contingent upon the receipt from Client free-of-charge of all Client Materials required for the Manufacture of the replacement Product. If SBL is unable to replace the Recalled or returned Products (except where this inability results from a failure to receive the Client Materials), then SBL shall credit to Client the price that Client paid to SBL for Manufacturing and Services for the affected Products. In all other circumstances, Recalls will be made at Client's cost and expense. The provisions of this Section 8.2 shall not apply to any Clinical Product and shall be Client's sole and exclusive remedy with respect to a Recall due to the delivery of Non-Conforming Product.

For the purpose of this Section, Recall expenses shall not include the value of the Product that is the subject of such Recall or any loss of reputation or good will, or lost profits or any other costs or expenses not associated with the Product that is the subject of the Recall. If SBL and Client cannot agree which Party is at fault or whether a Recall was reasonably beyond the control of the Parties, then an independent third party technical expert of international repute, acceptable to both Parties, shall be designated to make such determination. The designated technical expert shall not be an employee, consultant, officer, director or shareholder of, or otherwise associated with, or have been retained during the prior five (5) years by, or be retained during the ensuing five (5) years, by SBL, Client or their respective Affiliates. The technical expert's determination will be, in the absence of fraud or manifest error, binding and conclusive upon the Parties. The cost of designating such technical expert shall be borne by the Party determined at fault, and if neither Party is determined to be at fault, then the cost of such technical expert shall be borne by both Parties in equal proportion.

8.3 Records & Audit.

- 8.3.1 Audit by Client. Upon Client's request, but no more than once per year, SBL shall accept an audit of the Facility and, if necessary in the judgment of Client, the Warehouse, by Client and allow Client to inspect and audit the Facility and, if necessary in the judgment of Client, the Warehouse, and Manufacture of the Product solely to ascertain compliance by SBL with the terms of this MSA or any applicable PSA; provided, however that in the event Client uses a designee, SBL must provide prior written consent, which shall not be unreasonably withheld, conditioned or delayed. SBL shall be reimbursed for its reasonable costs for audits beyond the audit described in the first sentence of this Section 8.3.1, except for audits conducted to measure corrective action or remediation following a finding of deficiency either by Client in a previous audit or by a Regulatory Authority or as is contemplated by Section 8.3.2(ii). Client shall not be limited in the number of its audits conducted to measure corrective action or remediation, and such audits shall not count against the limit expressed in the first sentence of this Section 8.3.1. SBL will make Commercially Reasonable Efforts to require vendors or subcontractors to accept an audit or visit to their facilities by Client upon similar notice as described in Section 8.3.2 below.
- 8.3.2 Audit Notice. Client shall provide SBL with a written notice at least three (3) months prior to the initiation of the audit of the Facility and, if necessary in the judgment of Client, the Warehouse, set forth in Section 8.3.1, which shall be conducted on a mutually agreeable date and time, and with a mutually agreed duration, agenda, and number of attendees, which in the case of Client shall not be limited to less than three (3) without the consent of Client. Notwithstanding the foregoing, if the audit is required for cause (i) due to safety reasons or other reasons that necessitates immediate audit of or visit to the Facility or (ii) due to the SBL Assignable Error, the foregoing sentence shall not apply and Client may conduct such audit or visit by providing SBL with a prior notice by email. Access to SBL's facilities shall be coordinated with SBL so as to minimize disruption to SBL's ability to perform services for its other clients. Client representatives must comply with all of SBL's generally applicable cGMP, confidentiality and security procedures and protocols during such observations, consultations, and inspections. SBL shall at all times cooperate and provide all the necessary documents reasonably required by Client during such audit; provided that, to the extent necessary, SBL may redact or withhold documents to protect the confidential information of its other clients. Client shall be solely responsible for any costs and liability caused by Client's or its representatives' failure to comply with SBL's security, safety or confidentiality procedures.

SECTION 9 CONSIDERATION AND PAYMENT TERMS

9.1 Consideration. In consideration for SBL's performing the Service and other obligations undertaken by SBL pursuant to a PSA, Client shall pay SBL amounts as set forth in the applicable PSA (the "Service Fee"); (ii) a handling surcharge of a certain percentage or certain amount to be set forth in the applicable PSA of the costs of Raw Materials paid by SBL (including but not limited to taxes and customs duties/fees); (iii) a handling surcharge of a certain percentage or certain amount to be set forth in the applicable PSA related to the Client Materials (which shall be based on the actual costs of such materials as supported by reasonable documentary evidence as opposed to the market value thereof); and (iv) storage fees as set forth in the relevant PSA.

9.2 **Invoices.**

- 9.2.1 Service Fee of the Project Stages and Batches. According to the invoicing plan set forth in the applicable Project Plan or applicable PSA, or upon SBL's release of a Batch of Product, as applicable, SBL shall invoice Client for the Service Fee set forth in the applicable PSA.
- 9.2.2 Raw Materials. With respect to the Raw Materials, SBL shall submit invoices to Client for the applicable Raw Materials cost (including any agreed upon safety stock) as set forth according to Section 9.1 as follows. SBL shall submit an invoice to Client (i) for the cost of Specialized Raw Materials procured upon receipt of the invoice from vendors/suppliers; and (ii) for the cost of Common Raw Materials used by SBL's completion of such project stage or upon SBL's release of a Batch of Product as applicable. Notwithstanding the foregoing, the Parties shall collaborate in the selection of the vendors of the Raw Materials. All such vendors shall be approved by Client before supplying SBL with Raw Materials for Product.
- 9.2.3 <u>Client Materials.</u> With respect to the Client Materials, which shall be supplied by Client to SBL at no cost during SBL's performance the Service, SBL shall submit an invoice to Client in an amount as set forth in Section 9.1 upon SBL's completion of such project stage of the Service SBL's release of a Batch of Product, as applicable.

9.3 Payment.

- 9.3.1 Mode of Payment; Foreign Exchange. All payments to SBL due under the MSA or any applicable PSA shall be made within thirty (30) Business Days from the receipt of the SBL's invoice in USD \$ by means of telegraphic transfer to the account with the bank designated by SBL in the foregoing invoice. For the purpose of computing payment amounts incurred in a currency other than USD\$, such currency shall be converted into USD\$ using the basic exchange rate published by Bloomberg (or its successor institution) on its website "http://www.bloomberg.com/markets/currencies" (or any other website that may be used by Bloomberg or its successor institution for publication of currency exchange rates) at the opening of business on such invoice date.
- 9.3.2 Taxes. All prices and charges are exclusive of any applicable taxes, levies, imposts, duties and fees of whatever nature imposed by any law or regulations in any country in respect of the Services, importation or exportation of Raw Materials, Client Materials, Batches, and Product, which shall be paid by Client. For the avoidance of doubt, the foregoing shall not include any taxes imposed on the income or profit of SBL and any withholding tax lawfully levied on any payment to be made by Client to SBL, each of which shall be solely borne by SBL. Client shall pay or reimburse SBL for all customs duties and taxes in connection with the purchase, sale, importation or exportation of any Raw Materials, Client Materials, Batches, or Product or the provision of Services, except to the extent such duties and taxes are recoverable by or refundable to SBL. SBL agrees to assist Client in claiming exemption under double taxation or similar agreement or treaty from time to time in force to obtain a refund of any customs duties, value added taxes, and other taxes payable by SBL.
- 9.3.3 Price Adjustments. The Service Fees as set forth in the applicable PSA, shall be adjusted annually, on the last day of January, effective January 1 of each year during the Term, by the percentage change in the consumer price index as published by the Bank of Korea for the immediately preceding twelve (12) months. The relevant date for price adjustment under this Section shall be the issue date of SBL's invoice.
- 9.3.4 **Default Interest.** Any amount that is not paid by a Party to the other when due under the MSA or any PSA shall bear default interest at the rate of ten percent (10%) per annum, or such lesser maximum rate allowed by Applicable Law, from the day following the due date until paid in full. In the event there is an amount which is invoiced by SBL but not paid by Client for more than six (6) months after the due date, such event shall be considered a material breach of the relevant PSA.

SECTION 10 CONFIDENTIALITY

10.1 Confidential Information. "Confidential Information" shall mean any data, know-how and other information, whether technical or non-technical disclosed by one Party (hereinafter the "Disclosing Party") or otherwise became known to the other Party (hereinafter the "Receiving Party") hereunder relating to the subject matter of the MSA, regardless of form or manner of disclosure, i.e., whether disclosed in writing, in electronic file or format or in other tangible manner, or orally, visually or in other intangible manner. If a Party intends to disclose such information in writing, in electronic file or format or in other tangible manner, such Party will make reasonable efforts to indicate it is confidential; and if to disclose orally, visually or in other intangible manner, such Party will make reasonable efforts to reduce it in summary form in writing or in electronic file or format, identified as confidential and delivered to the other Party within thirty (30) days after such oral or visual disclosure; provided, however, in each case, a failure to do so shall not constitute a breach of this term nor shall deny, negate or destroy the confidential nature thereof, and no such failure shall serve as conclusive evidence that the disclosed information shall not be considered Confidential Information by and between the Parties. Furthermore, the existence and terms of the MSA shall be deemed to be the Confidential Information of both Parties and any information, data or results disclosed by SBL to Client relating to the Product, the Services or Manufacture under this MSA or any PSA, as well as the Manufacturing Process, the Project Plan and the Specifications shall be the Confidential Information of Client, unless otherwise specifically identified as SBL Confidential Information by the terms of this Agreement. The Parties acknowledge and agree that the Manufacturing Process and the Project Plan may contain a mixture of both Party's Background IP, ownership of which shall not be affected by this MSA.

Notwithstanding the foregoing, Confidential Information shall not include the information, which as evidenced by written records:

- (a) was at the time of disclosure by the Disclosing Party hereunder publicly known or available;
- (b) after disclosure by the Disclosing Party hereunder, became publicly known or available by publication or otherwise, other than by an authorized act or omission by the Receiving Party;
- (c) was in the possession of the Receiving Party without confidentiality restriction at the time of the disclosure by the Disclosing Party hereunder;
- (d) was lawfully received from any third party having the lawful right to make such disclosure, without obligation of confidentiality; or
- (e) was independently developed by the Receiving Party's directors, officers or employees without reference to the Confidential Information, as demonstrated by records created contemporaneously with such development.
- 10.2 Confidentiality. The Receiving Party recognizes the proprietary and confidential nature of the Disclosing Party's Confidential Information and agrees that no right, title, ownership, license, or interest of any character in the Disclosing Party's Confidential Information other than as specifically granted herein, is conveyed or transferred to the Receiving Party. Both Parties further agree to maintain the Disclosing Party's Confidential Information in confidence and not to disclose or divulge the Disclosing Party's Confidential Information, in whole or in part, to any third party, and not use the Disclosing Party's Confidential Information for any purpose other than pursuing the MSA. Each Party shall guard such Confidential Information using the same degree of care as it normally uses to guard its own confidential or proprietary information of like importance, but in any event no less than reasonable care. The Receiving Party shall limit disclosure of the Disclosing Party's Confidential Information to its and those of its Affiliates' directors, officers, employees, consultants and agents ("Representatives") who have a need to know the Disclosing Party's Confidential Information for performance of the Service and implementation of the MSA, provided that, the Receiving Party shall undertake procedures to ensure that each of its Representatives to whom the Disclosing Party's Confidential Information is disclosed understands (i) the confidential nature of the Disclosing Party's Confidential Information and (ii) that he or she is under an obligation similar to those contained herein to hold the Disclosing Party's Confidential Information disclosed strictly confidential.
- 10.3 Authorized Disclosures. Disclosure is permitted in the event that (a) the Disclosing Party's Confidential Information is reasonably required to obtain or maintain any Regulatory Approvals for the Products in any or all jurisdictions or (b) the Disclosing Party needs to disclose such Confidential Information to comply with Applicable Law; provided that such Receiving Party shall exercise its Commercially Reasonable Efforts to limit disclosure of the Disclosing Party's Confidential Information to that which is necessary for compliance and to otherwise maintain the confidentiality of the Confidential Information.
- 10.4 <u>Survival of confidential obligations</u>. The confidential obligations of the Receiving Party shall survive for a period of five (5) years from the expiration or termination of this MSA.
- 10.5 Return of the Confidential Information. All written, printed or other tangible Confidential Information of the Disclosing Party disclosed under the MSA, and all copies thereof shall be returned to the Disclosing Party (or destroyed at the Disclosing Party's request) by the Receiving Party within thirty (30) business days from the written request by the Disclosing Party. All Confidential Information disclosed electronically shall be completely deleted and destroyed by the Receiving Party within thirty (30) business days from the written request by the Disclosing Party. Notwithstanding the foregoing, (i) digital backup files automatically generated by the Receiving Party's customary electronic data processing system may be retained and properly stored as confidential files for the sole purpose of backup and will be deleted in accordance with the Receiving Party's retention policy, and (ii) a single copy of the Confidential Information may be retained in the secured files of the Receiving Party for the sole purpose of determining the scope of obligations incurred by it under the MSA provided that the Receiving Party shall keep such Confidential Information in confidence and will use the Confidential Information solely to comply with the terms of the MSA as well as the applicable law, rule and regulation.

SECTION 11 OWNERSHIP OF MATERIALS AND INTELLECTUAL PROPERTY

- 11.1 Reference Standard, Client Technology, Client Materials, Cell Line, and Product. SBL hereby understands and agrees that all rights to, titles of and interests in the Reference Standards, Client Technology, Client Materials, Cell Line, Product, the Manufacturing Process and any work in process or semi processed goods thereof belong to Client, unless otherwise provided herein.
- 11.2 Background Intellectual Property. It is acknowledged that each Party possesses Background IP. Any Intellectual Property relating to the Reference Standards, Client Technology, Client Materials and Cell Line owned and/or controlled by Client as of the date of provision of such Reference Standards, Client Technology, Client Materials and Cell Line by Client to SBL pursuant to Section 5.1, shall be deemed to be included in the Background IP of Client. Client hereby grants SBL a royalty-free, non-transferable, revocable and non-sublicensable and fully-paid-up right and license to use such Intellectual Property relating to such Reference Standards, Client Technology, Client Materials, Cell Line, during the Term for the sole purposes of Manufacturing of the Product or Services in accordance with the MSA.
- 11.3 Inventions. Any Intellectual Property arising out of or resulting from the Service under the MSA, including but not limited to those contained in the Manufacturing Documentation, shall be hereinafter collectively called an "Invention".
 - 11.3.1 Client Invention. Any Invention that is conceived and first reduced to practice solely by one or more employees or officers of SBL (or its third party consultant or subcontractor) and does not constitute an SBL Invention shall be a "Client Invention". SBL shall notify Client of such Client Invention(s) to Client immediately after SBL, the Project Management Team Leader, respective project personnel, SBL employees or officers or other applicable third parties working for SBL hereunder makes, conceives or reduces to practice such Client Invention, and shall take all necessary measures so that Client would have the sole and exclusive ownership of any and all Client Invention. Client may use any Client Invention for any purpose, including filing patent application and SBL shall provide reasonable cooperation to Client at the expense of Client (as to all reasonable out-of-pocket expenses incurred by SBL that are supported by adequate documentation). In the event that SBL wishes to use the Client Invention for purposes outside the scope of the MSA, Client hereby grants SBL a worldwide, irrevocable, royalty-free and fully-paid- up license to use such Client Invention. This Section 11.3 shall survive the expiration or earlier termination and continue in effect as long as the intellectual property right to such Client Invention is legally valid.
 - 11.3.2 SBL Invention. Any Invention that is conceived and first reduced to practice solely by one or more employees or officers of SBL (or its third party consultant or subcontractor) and which is not derived from, or arises out of Client Background IP or Client's Confidential Information or any other proprietary right of Client or its third party vendors, contractors, or other partners or clients under or in connection with the MSA, shall be the property of SBL ("SBL Invention"), and shall not be deemed to be Client Invention or Joint Invention for the purposes of the MSA. No Background IP of SBL and the SBL Invention shall be incorporated into the Product or Manufacturing Process without the joint review and approval of both Parties and the terms of a license to use such Background IP or SBL Invention shall be agreed upon prior to such incorporation.
 - 11.3.3 Client-SBL Joint Invention. Any Invention that is conceived and first reduced to practice jointly by one or more employees, or officers of SBL (or its third party consultant or subcontractor) or their respective Affiliates, on the one hand and one or more employees, officers, agents or contractors of Client (or its Affiliates) on the other hand and which is not derived from, or arises out of either Party's Background IP or Confidential Information or any other proprietary right of a Party or its third party vendors, contractors, or other partners or clients under or in connection with the MSA, shall be jointly owned by Client and SBL (a "Joint Invention"), and shall not be a Client Invention or SBL Invention for the purposes of the MSA. Subject to the terms and conditions of the MSA, any such Joint Invention may be exploited by SBL or Client without compensation and liability of other obligation (including accounting obligations) to the other Party, and each Party has a non-exclusive, royalty-free, worldwide license, with the right to sublicense, under its interest in such Joint Inventions and any Intellectual Property rights (including patents) covering the same, for any purpose; provided that in the event of sublicense, each Party shall make written notice to the other Party of the fact. This license shall continue for the life of the applicable right.

SECTION 12 WARRANTIES

- 12.1 The Parties General Warranties. Each Party warrants and represents that: (i) it has the corporate power and authority to enter into this MSA and has taken all necessary action on its part required to authorize the execution, delivery and performance of this Agreement; (ii) it is aware of no legal, contractual or other restriction, limitation or condition that might adversely affect its ability to enter into this MSA and perform its obligations hereunder; (iii) it is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated; (iv) this MSA (a) has been duly executed and delivered by a duly authorized representative of it, and (b) is the legal, valid and binding obligation of it, enforceable against it in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to or affecting creditors' rights generally; and (v) the execution, delivery and performance of this Agreement by it does not and will not (a) violate any Applicable Laws applicable to it, or (b) violate or conflict with any provision of its Articles of Incorporation or By-laws or other organizational documents.
- 12.2 Client's Warranties. Client represents, warrants and covenants to SBL that as of the Effective Date of the MSA and during the Term: (a) Client will comply with all Applicable Laws, and that it will keep SBL informed of any information known to Client which would affect SBL's provision of the Service hereunder and (c) to the best of its knowledge, SBL's use of the Client Materials, Manufacturing Process, and Client Technology for the purpose of the Service and to the extent as set forth in the MSA will not infringe any third party's Intellectual Property rights.
- 12.3 SBL's Warranties. SBL represents, warrants and covenants that:
 - 12.3.1 As of the Effective Date and during the Term, (i) SBL is the lawful owner, lessee, operator, or licensee of the Facility, equipment, machinery, as well as permissions required, to enable SBL to perform its obligations under this MSA, and (ii) to the best of SBL's knowledge none of the SBL Inventions or SBL Background IP infringes any third party Intellectual Property Right.
 - 12.3.2 All Product Batches, at the time of delivery to Client's designated carrier, shall (a) conform to the Specifications (except for Pilot Batches and Engineering Batches unless otherwise agreed); (b) be Manufactured, packaged, handled and stored in compliance with the requirements of cGMPs (except for Pilot Batches and Engineering Batches unless otherwise agreed) and all Applicable Laws; (c) comply with the Standard Operating Procedures; (d) be Manufactured in compliance with the Quality Agreement; and (e) be transferred free and clear of any liens, claims or encumbrances of any kind.
 - 12.3.3 (i) SBL is not nor has it ever been, and (ii) SBL has not used, and will not use, the services of any person excluded, debarred, suspended (or subject to exclusion, debarment or suspension) under 21 U.S.C. §335(a) or (b) or otherwise disqualified by Applicable Law, including, the FDA Debarment List (http://www.fda.gov/ora/compliance_ref/debar/default.htm), as amended or replaced from time to time, in connection with any of the Services or Manufacturing performed under this MSA or any PSA. SBL agrees to notify Client promptly in the event of any violation of SBL's obligations under this Section. This certification applies to SBL and its respective officers, agents, and employees as well as subcontractors performing on behalf of SBL under this Agreement.
- 12.4 No Other Warranties. THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS SECTION ARE EXPRESSLY IN LIEU OF AND EXCLUDE, AND THE PARTIES HEREBY EXPRESSLY DISCLAIM AND NEGATE, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAWS, ALL OTHER REPRESENTATIONS AND WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED (ARISING BY OPERATION OF LAW OR OTHERWISE), INCLUDING IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, EVEN IF THAT PURPOSE IS KNOWN.

SECTION 13 INDEMNIFICATION

- 13.1 Indemnification by SBL. SBL shall indemnify and hold harmless Client, its Affiliates, and their respective officers, directors, employees or agents from and against any Damages arising or resulting from any third party (which shall exclude Client Affiliates) claims to the extent such Damages are relating to, arising out of, in connection with, or resulting from claims, demands, or actions based upon (i) gross negligence or willful misconduct; (ii) breach of the MSA or any PSA; (iii) violation of Applicable Law, in the case of clauses (i) through (iii) of or by SBL or its officers, directors, employees or agents, including any subcontractor or vendor used in the Manufacture of Product, or (iv) any claim that SBL's use of SBL Background IP infringes any third party's Intellectual Property rights, except to the extent that such Damages are caused by the causes as set forth in Section 13.2 for which Client is obliged to indemnify.
- 13.2 Indemnification by Client. Client shall indemnify and hold harmless SBL, its Affiliates, and their respective officers, directors, employees or agents from and against any Damages arising or resulting from any third party (which shall exclude SBL Affiliates) claims to the extent such Damages are relating to, arising out of, in connection with, or resulting from claims, demands or actions based upon (i) gross negligence or willful misconduct, (ii) any claim that SBL's use of the Client Materials, Manufacturing Process, and Client Technology for the purpose of the Services and solely to the extent as set forth in the MSA infringes any third party's Intellectual Property rights, (iii) breach of the MSA or any PSA, or (iv) violation of Applicable Law, in each case except to the extent that such Damages are caused by the causes as set forth in Section 13.1 for which SBL is obliged to indemnify and in the case of clauses (i), (iii), and (iv) of or by Client or its officers, directors, employees, or agents.
- 13.3 Indemnification Procedure. The foregoing indemnification by SBL or Client shall be conditioned, if and to the extent Damages are based on or related to a third party claim, upon a Party who intends to claim indemnification under Sections 13.1 and 13.2 (the "Indemnified Party") (i) providing written notice to the other Party ("Indemnifying Party") within twenty (20) calendar days after the Indemnified Party have been given written notice of such third party claim, provided that absence or delay of such prior written notice will not relieve the Indemnifying Party of its obligation to indemnify except to the extent such absence or delay materially prejudices the Indemnifying Party's ability to defend the third party claim, (ii) permitting the Indemnifying Party, upon timely notice by the Indemnified Party, the opportunity to assume full responsibility (at the Indemnifying Party's cost and expense) for the investigation and defense of any such claim with counsel reasonably satisfactory to the Indemnified Party, provided, however, the Indemnifying Party shall keep the Indemnified Party informed as to the progress of the defense of any claim and that the Indemnified Party shall cooperate in such defense and shall make available all records, materials and witness reasonably requested by the Indemnifying Party in connection therewith; and (iii) not settling or compromising any such claim without the Indemnifying Party's prior written consent, with such consent not to be unreasonably denied, withheld or conditioned. Furthermore, the Indemnifying Party shall not settle or compromise any such claim without the Indemnified Party's prior written consent; provided, however, no such consent shall be required if such settlement or compromise involves only the payment of money, does not require a finding or admission of fault or guild on the party of the Indemnified Party, and provides for a general release of the Indemnified Party.

SECTION 14 DISCLAIMER OF CONSEQUENTIAL DAMAGES: LIMITATION OF LIABILITY

14.1 Disclaimer of Consequential Damages. EXCEPT FOR DAMAGES BASED ON OR RELATED TO A THIRD PARTY CLAIM ARISING UNDER SECTIONS 13.1 AND 13.2, OR BASED ON GROSS NEGLIGENCE OR WILLFUL MISCONDUCT, NEITHER PARTY WILL BE LIABLE UNDER THIS AGREEMENT FOR ANY SPECIAL, PUNITIVE, CONSEQUENTIAL, INCIDENTAL OR OTHER INDIRECT DAMAGES OF ANY TYPE OR NATURE, WHETHER BASED IN CONTRACT, TORT, STRICT LIABILITY, NEGLIGENCE OR OTHERWISE, INCLUDING LOSS OF PROFITS OR REVENUES.

14.2 Limitation of Liability. Specific caps on Damages shall be set forth in the applicable PSA.

SECTION 15 TERM AND TERMINATION OF AGREEMENT

- 15.1 Term. This MSA will become effective as of the Effective Date and will be in effect for as long as a PSA is in effect (the "Term"). Each PSA will have its own initial term as stated therein. Each PSA for clinical supply shall automatically renew for successive terms of one (1) year each or upon execution of a PSA for commercial supply, whichever is earlier, and each PSA for commercial supply, including supply of stocking inventories in advance of Regulatory Approval, shall automatically renew for successive terms of three (3) years each, unless a Party gives written notice to the other Party of its intention to not renew a PSA for clinical supply ninety (90) days prior to the end of the then current PSA term, or a Party gives written notice to the other Party of its intention to not renew a PSA for commercial supply thirty-six (36) months prior to the end of the then current PSA term.
- 15.2 **Termination**. This MSA or a PSA may be earlier terminated as set forth in this Section 15.2.
 - 15.2.1 Material Breach. A Party may terminate any PSA for a material breach by the other Party; provided, however, that the non-breaching Party shall give the breaching Party written notice of such breach and if the breaching Party fails to commence Commercially Reasonable Efforts to cure that breach within twenty (20) Business Days after receipt of such written notice, then the non-breaching Party may terminate this Agreement on twenty (20) Business Days' written notice after expiration of such twenty (20) Business Day period. This MSA shall terminate if all effective PSAs are terminated.
 - 15.2.2 Insolvency. This MSA may be terminated by either Party upon written notice at any time during the MSA if the other Party: (a) files in any court pursuant to any statute a petition in bankruptcy or insolvency or for reorganization or for an arrangement or for the appointment of a receiver or trustee of such Party, or of its assets; (b) proposes a written agreement of composition for extension of its debts; (c) is served with an involuntary petition against it, filed in any insolvency proceeding which is admitted in the court; or (d) makes an assignment for the benefit of its creditors. The Party affected shall immediately notify the other Party in writing of the occurrence of any of the foregoing events.
 - 15.2.3 Force Majeure. Either Party may terminate a PSA in accordance with Section 17.3 in the event a Party is unable to perform its obligations pursuant to a PSA due to a Force Majeure Event.
 - 15.2.4 Import Failure. Client shall be entitled to terminate a PSA upon twenty (20) Business Days' notice to SBL, if, despite the efforts of SBL pursuant to Section 5.4.1, Client Material destined for use in connection with the Services have been prohibited from entry into the Republic of South Korea by the Korean Custom Service. Such termination right shall only exist if such Client Materials have been denied entry and shall not be implicated by mere delays in the importation process. If such termination relates to efforts to import into the Republic of Korea Client Materials for the initial Batch of a Product under a PSA, the applicable PSA shall be deemed to be void *ab initio* without further cost or expense to either Party; provided, however, the Party's thereafter shall reasonably cooperate with regard to disposition of such Client Materials at the cost of Client.
 - 15.2.5 Other Specified Events. The Parties may additionally terminate a PSA as set forth in the applicable PSA.

15.3 Effect of Expiration or Termination.

- 15.3.1 Payment of Amounts Due. Expiration or termination of the MSA or PSA for any reason shall not exempt any Party from paying to any other Party any amounts owing to such Party at the time of such expiration or termination; provided, however, with respect to a termination deemed to be *ab initio*, and upfront fee paid to SBL shall not be deemed earned and shall be returned to Client.
- 15.3.2 **Decommissioning.** Upon expiration or termination of a PSA for any reason, SBL shall cease and refrain from the Services described in any applicable PSA (including the Manufacturing and supplying the Product) for Client unless otherwise provided in the following Sections 15.3.2(a) to 15.3.2(d), and both Parties shall pursue decommissioning activities as set forth hereunder.

(a) Fully Manufactured Product.

- (i) If Client terminates a PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3, 15.2.4 or 15.2.5, upon Client's election, SBL shall (i) deliver already fully Manufactured Product to Client in accordance with the terms and conditions of the MSA and applicable PSA or (ii) destroy such Product. If Client elected (i) above, Client shall pay the Service Fees and any related costs or fees for the Service relating to such Product in accordance with the terms and conditions of the MSA and applicable PSA, and if Client elected (ii) above, SBL shall bear the costs and expenses for such destruction, except in the case of Section 15.2.4 or Section 15.2.5 for which destruction Client shall pay the cost.
- (ii) If SBL terminates a PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3, or 15.2.5, upon SBL's election, SBL may (i) deliver the fully Manufactured Product to Client in accordance with the terms and conditions of the MSA and applicable PSA (including the current Firm Period period or Binding Year in the PSA) or (ii) destroy such Product. If SBL elected (i) above, Client shall pay the Service Fee and any related costs or fees for the Service relating to such Product in accordance with the terms and conditions of the MSA and applicable PSA, and if SBL elected (ii) above, Client shall bear the costs and expenses for such destruction and the costs incurred by SBL for the Service; provided, however, such costs and expenses to be borne by Client shall in no event exceed the Service Fee for the Service relating to such Product.
- (iii) If a PSA is naturally expired pursuant to Section 15.1, the provisions of (ii) above shall apply

(b) Client Materials being used for the Service (Product in Process).

- (i) If Client terminates a PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3, 15.2.4, or 15.2.5, upon Client's election, SBL shall (i) continue to use the Client Materials being used for the Manufacturing hereunder (the Product in process) and deliver the fully Manufactured Product to Client in accordance with the terms and conditions of the MSA and applicable PSA, or (ii) deliver to Client or destroy such Product in process. If Client elected (i) above, Client shall pay the Service Fee and any related costs or fees for the Service relating to the fully Manufactured Product in accordance with the terms and conditions of the MSA and applicable PSA, and if Client elected (ii) above, SBL shall bear the costs and expenses for such activities except in the case of Section 15.2.4 or Section 15.2.5 for which destruction Client shall pay the cost.
- (ii) If SBL terminates a PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3, or 15.2.5, upon SBL's election, SBL may (i) continue to use the Client Materials being used for the Manufacturing hereunder (the Product in process) and deliver the fully Manufactured Product to Client in accordance with the terms and conditions of the MSA and applicable PSA, or (ii) destroy such Product in process. If SBL elected (i) above, Client shall pay the Service Fee and any related costs or fees for the Service relating to the fully Manufactured Product in accordance with the terms and conditions of the MSA, and if SBL elected (ii) above, Client shall bear the costs and expenses for such activities.
- (iii) If a PSA is naturally expired pursuant to Section 15.1, the provisions of (ii) above shall apply.
- (c) Client Materials, Cell Line, and Reference Standards. Upon expiration or termination of a PSA, upon Client's election, SBL shall deliver to Client and/or destroy all remaining Client Materials (subject to Sections 15.3.2(a) and 15.3.2(b)), all remaining Cell Line vials, Reference Standards and other materials required for Manufacturing.

The costs and expenses for such activities shall be borne by the Parties as follows:

- (i) If Client terminates the PSA pursuant to Section 15.2.1, 15.2.2 or 15.2.3, SBL shall deliver or dispose at no additional cost to Client and SBL shall bear such costs and expenses for such activities;
- (ii) If SBL terminates the PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3, or 15.2.5, or Client terminates the PSA pursuant to Section 15.2.4 or 15.2.5, Client shall bear such costs and expenses for such activities;
- (iii) If a PSA is naturally expired pursuant to Section 15.1, the provisions of (ii) above shall reply.

(d) Raw Materials.

- (i) If Client terminates a PSA pursuant to Section 15.2.1, 15.2.2 or 15.2.3 and if Client so elects, SBL shall deliver the remaining Raw Materials to Client for Client's payment of SBL's cost to procure such Raw Materials, or dispose of them at Client's election. SBL shall bear the costs and expenses for the delivery of the Raw Materials.
- (ii) If SBL terminates a PSA pursuant to Section 15.2.1, 15.2.2, 15.2.3 or 15.2.5, or Client terminates a PSA pursuant to Section 15.2.4 or 15.4.5, SBL may deliver the remaining Raw Materials to Client or dispose of them at SBL's election. If so delivered, Client shall pay SBL's cost to procure such Raw Materials to SBL and bear the costs and expenses for the delivery of such Raw Materials by SBL.
- (iii) If a PSA is naturally expired pursuant to Section 15.1, the provisions of (ii) above shall apply.

(e) Outstanding Obligations Regarding Purchase of Product.

- (i) If Client terminates a PSA pursuant to Section 15.2.1 or 15.2.2 Client shall be released from any outstanding binding obligations to purchase Product as of the date of notice of termination including but not limited to pursuant to a Firm Period, binding forecast, purchase order, minimum purchase commitment, or otherwise; provided, however, that Client shall purchase Product Manufactured prior to such date of notice, pursuant to the terms of this MSA and PSA.
- (ii) If SBL terminates a PSA pursuant to Section 15.2.1 or 15.2.2, outstanding binding obligations to purchase Product as of the date of notice of termination shall survive termination of such PSA, including but not limited to pursuant to a Firm Period, binding forecast, purchase order, minimum purchase commitment, or otherwise.
- (iii) For all other cases of termination of a PSA, subsection (ii) shall apply.
- (f) <u>Survival</u>. Any termination or expiration of this MSA shall not affect any outstanding obligations due hereunder prior to such termination or expiration, nor shall it prejudice any other remedies that the parties may have under this MSA. For greater certainty, except as otherwise expressly provided, termination or expiration of this MSA, irrespective of the cause, shall not affect any rights or obligations which, from the context thereof, are intended to survive termination or expiration of this MSA, including but not limited to Sections 1, 8, 9, 10, 11, 12, 13, 14, 15, 16 and 17.2.
- 15.3.3 Outbound Transfer. Upon any termination of a PSA or in connection with its expiration, Client shall be entitled to give notice to SBL of a Manufacturing Process Transfer, which shall commence the preparation and effectuation of a Manufacturing Process Transfer Plan for the Product identified therein. In addition, Client shall be entitled to give notice to SBL of a Manufacturing Process Transfer and commence the preparation and effectuation of a Manufacturing Process Transfer Plan in the absence of a termination or expiration of a PSA in order to establish second source Manufacturing. Client shall pay SBL's reasonable costs of assistance related to effectuation of the Manufacturing Process Transfer Plan.

SECTION 16 ARBITRATION

- 16.1 Informal Discussions. Except as otherwise provided herein, in the event of any controversy or claim arising out of or relating to this MSA, or the rights or obligations of the Parties hereunder, the Parties shall first try to settle their differences amicably between themselves through the Core Team and then the JSC. Thereafter, either Party may initiate informal dispute resolution on the executive level by sending written notice of the dispute to the other Party, and within thirty (30) days after such notice appropriate executives of the Parties shall meet for attempted resolution by good faith negotiations. If such representatives are unable to resolve such disputed matter within such thirty (30) day period, either Party may refer the matter by written notice to the Chief Executive Officer of the other Party, or his/her designee, and the Chief Executive Officer of such Party, for discussion and resolution. If such individuals or their designees are unable to resolve such dispute within thirty (30) days of such written notice, and such dispute relates to a claimed breach of this MSA, a PSA or a QAG, either Party may initiate binding arbitration proceedings in accordance with the provisions of this Article 16. For the avoidance of doubt, no claim other than a claim of a breach of this Agreement shall be subject to arbitration.
- 16.2 Arbitration. If the Parties do not fully settle a claimed breach of this MSA, a PSA or a QAG pursuant to Section 16.1, and a Party wishes to pursue the matter, each such claim shall be finally resolved by binding arbitration in accordance with the Commercial Arbitration Rules of the International Chamber of Commerce ("ICC"), and judgment on the arbitration award may be entered in any court having jurisdiction thereof to enforce the arbitration award. The arbitration shall be conducted by a panel of three neutral persons experienced in the pharmaceutical business, and within thirty (30) days after initiation of arbitration, each Party shall select one person to act as arbitrator and the two Party-selected arbitrators shall select a third arbitrator within thirty (30) days of their appointment. If the arbitrators selected by the Parties are unable or fail to agree upon the third arbitrator, the third arbitrator shall be appointed by the ICC. The place of arbitration shall be New York, New York, United States and all proceedings and communications shall be in English. Either Party may apply to the arbitrators for interim injunctive relief until the arbitration award is rendered or the controversy is otherwise resolved. Either Party also may, without waiving any remedy under this Agreement, seek from any court having jurisdiction any injunctive or provisional relief necessary to protect the rights or property of that Party pending the arbitration award. The arbitrators shall have no authority to award punitive or any other type of damages not measured by a Party's direct compensatory damages, and in all cases, any decision or determination by the arbitrators shall comply with Article 14, as applicable. The Parties agree that, in the event of a good faith dispute over the nature or quality of performance under this Agreement, neither Party may terminate this Agreement until final resolution of the dispute through arbitration or other judicial determination. The Parties further agree that any payments made pursu

16.3 Costs and Fees. Each Party shall bear its own attorneys' fees, costs, and disbursements arising out of the arbitration, and shall pay an equal share of the fees and costs of the arbitrators. Absent the filing of an application to correct or vacate the arbitration award as permitted by Applicable Law, each Party shall fully perform and satisfy the arbitration award within fifteen (15) days after the service of the award on such Party.

SECTION 17 MISCELLANEOUS

17.1 Notices. Any notice required or permitted under the MSA shall be in writing with duly authorized signature and made to the following addresses or facsimile numbers:

If to Client:

TG Therapeutics, Inc.
2 Gansevoort St., 9th Floor
New York, NY 10014
Attention: Senior Vice President Operations
Facsimile:

If to SBL:

Samsung BioLogics Co., Ltd.
300, Songdo bio-daero, Yeonsu-gu
Incheon 21987, South Korea
Attention: Head of Corporate Business Operations

Facsimile: +82-32-455-3242

With copy to: SBL Legal & Compliance Department

Either Party may change its designated address and facsimile number by notice to the other Party in the manner provided in this Section 17.1.

Any notice shall be deemed to have been delivered on the date of delivery of delivered personally, or on the next day of sending if sent by facsimile, or on the fifth day of posting if sent by registered or certified mail with return receipt requested and postage prepaid.

- 17.2 Governing Law. This MSA shall be construed and interpreted in accordance with the laws of State of New York, United States and all rights and remedies shall be governed by such laws without regard to principles of conflicts of law. The United Nations Convention on Contracts for the International Sale of Goods shall not apply to the transactions contemplated by the MSA.
- <u>17.3</u> <u>Effect of Force Majeure Event.</u> Neither Party (the "Affected Party") shall be liable to the other Party (the "Non-Affected Party") for failure or delay to perform its obligation under the MSA or any applicable PSA when such failure or delay is due to riots, storms, fires, explosions, floods, earthquakes, war, embargoes, blockades, insurrections, terrorism, an act of God or any other cause similar thereto which is beyond the control of the Affected Party including those affected upstream suppliers ("Force Majeure Event").

Each Party agrees to give the other Party prompt written notice of the occurrence of any Force Majeure Event, the nature thereof, and the extent to which the affected Party will be unable fully to perform its obligations under the MSA. If a condition constituting Force Majeure Event as defined herein exists for more than one hundred eighty (180) consecutive days, or it is reasonably foreseeable that such Force Majeure Event will persist for more than one hundred eighty (180) days consecutive days, the Parties shall endeavor in good faith to negotiate a mutually satisfactory solution to the problem, if practicable, including use of a third party to fulfill the obligations hereunder of the party invoking Force Majeure Event, at the expense of the party invoking Force Majeure Event. If after good faith efforts to negotiate a mutually satisfactory solution for at least sixty (60) days the Parties have failed to reach agreement, the Non-Affected Party shall have the right to terminate this MSA upon thirty (30) days' written failure.

17.4 Assignment. Neither Party shall assign, in whole or in part, the MSA without the prior written consent of the other Party, such approval not to be unreasonably withheld, conditioned or delayed. Notwithstanding the above, Client may, without such consent, assign the MSA to (i) its Affiliate or (ii) any purchaser of Client's rights relating to the Product or all or substantially all of the assets of Client, or of all of its capital stock, or to any successor corporation or entity resulting from any merger or consolidation of such Party with or into such corporation or entity. Notwithstanding the above, SBL may, without such consent, assign the MSA to (i) its Affiliate or (ii) any purchaser of substantially all of the assets of SBL, or of all of its capital stock, or to any successor corporation or entity resulting from any merger or consolidation of such Party with or into such corporation or entity.

- 17.5 No Grant of License. Nothing in the MSA shall affect, or grant any right to, patents, know-how or other intellectual property owned by either Party prior to the commencement of the MSA unless otherwise expressly provided in the MSA.
- 17.6 No Right to Use Names. Except as expressly provided herein, no right, expressed or implied, is granted by the MSA to use in any manner the name of either of the Parties or any other trade name, symbol, logo or trademark of the other Party in connection with the performance of the MSA, without the prior written consent of the other Party.
- 17.7 Independent Contractors. The Parties hereto are independent contractors and nothing contained in the MSA shall be deemed or construed to create a partnership, joint venture, employment, franchise, agency or fiduciary relationship between the Parties.
- 17.8 Integration. This MSA constitutes the entire agreement between the Parties relating to the subject matter of the MSA and supersedes all previous oral and written communications between the Parties with respect to the subject matter of the MSA.
- 17.9 Amendment; Waiver. Except as otherwise expressly provided herein, no alteration of or modification to the MSA shall be effective unless made in writing and executed by an authorized representative of both Parties. No course of dealing or failing of either Party to strictly enforce any term, right or condition of the MSA in any instance shall be construed as a general waiver or relinquishment of such term, right or condition. The observance of any provision of the MSA may be waived (either generally or any given instance and either retroactively or prospectively) only with the written consent of the Party granting such waiver.
- 17.10 Severability. The Parties do not intend to violate any Applicable Law. However, if any sentence, paragraph, clause or combination of the MSA is in violation of any law or is found to be otherwise unenforceable, such sentence, paragraph, clause or combination of the same shall be deleted and the remainder of the MSA shall remain binding, provided that such deletion does not alter the basic purpose and structure of the MSA.
- 17.11 Construction. The Parties mutually acknowledge that they have participated in the negotiation and preparation of the MSA. Ambiguities, if any, in the MSA shall not be construed against any Party, irrespective of which Party may be deemed to have drafted the MSA or authorized the ambiguous provision.
- 17.12 Interpretation. The captions and headings to the MSA are for convenience only, and are to be of no force or effect in construing or interpreting any of the provisions of the MSA. Unless context otherwise clearly requires, whenever used in the MSA: (a) the words "include" or "including" shall be construed as incorporating, also, "but not limited to" or "without limitation"; (b) the words "hereof," "herein," "hereby" and derivative or similar words refer to the MSA; (c) the word "law" or "laws" shall mean any applicable, legally binding statute, ordinance, resolution, regulation, code, guideline, rule, order, decree, judgment, injunction, mandate or other legally binding requirement of a governmental authority (including a court, tribunal, agency, legislative body or other instrumentality of any (i) government or country or territory, (ii) any state, province, county, city or other political subdivision thereof, or (iii) any supranational body); and (d) all references to the word "will" are interchangeable with the word "shall" and shall be understood to be imperative or mandatory in nature. All references to days, months, quarters or years are references to calendar days, calendar months, calendar quarters, or calendar years. Whenever any matter hereunder requires consent or approval, such consent or approval shall not be unreasonably withheld or delayed.
- 17.13 Counterparts. This MSA may be executed in two or more counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

IN WITNESS WHEREOF, the Parties have executed the MSA as of the date first above written.

TG THERAPEUTICS, INC.

Signature: <u>/s/ Michael S. Weiss</u>
Name: Michael S. Weiss
Title: Chief Executive Officer

Date: February 19, 2018

SAMSUNG BIOLOGICS CO., LTD.

Signature: /s/ Dr. Tae Han Kim

Name: Dr. Tae Han Kim

Title: Representative Director and President

Date: February 21, 2018

CERTIFICATION OF PERIODIC REPORT PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Michael S. Weiss, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TG Therapeutics, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2019 /s/ Michael S. Weiss

Michael S. Weiss Executive Chairman, Chief Executive Officer and President Principal Executive Officer

CERTIFICATION OF PERIODIC REPORT PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Sean A. Power, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TG Therapeutics, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2019 /s/ Sean A. Power

Sean A. Power Chief Financial Officer Principal Financial and Accounting Officer

STATEMENT OF CHIEF EXECUTIVE OFFICER OF TG THERAPEUTICS, INC. PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of TG Therapeutics, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2019 as filed with the Securities and Exchange Commission (the "Report"), I, Michael S. Weiss, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, based on my knowledge:

- 1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2019 /s/ Michael S. Weiss

Michael S. Weiss Executive Chairman, Chief Executive Officer and President Principal Executive Officer

STATEMENT OF CHIEF FINANCIAL OFFICER OF TG THERAPEUTICS, INC. PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of TG Therapeutics, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2019 as filed with the Securities and Exchange Commission (the "Report"), I, Sean A. Power, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, based on my knowledge:

- 1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2019 /s/ Sean A. Power

Sean A. Power Chief Financial Officer Principal Financial and Accounting Officer